Oracle® Fusion Middleware

User's Guide for Oracle Business Process Management 11*g* Release 1 (11.1.1.9) **E15175-16**

February 2015

Documentation for process participants, process owners, administrators, and supervisors using either Oracle Business Process Management Workspace or Oracle Business Process Management Process Spaces to act on tasks, activity guides, and process instances.



Oracle Fusion Middleware User's Guide for Oracle Business Process Management 11g Release 1 (11.1.1.9)

E15175-16

Copyright © 2010, 2015, Oracle and/or its affiliates. All rights reserved.

Primary Author: Oracle Corporation

Contributor: Oracle Business Process Management Suite development, product management, and quality assurance teams

This software and related documentation are provided under a license agreement containing restrictions on use and disclosure and are protected by intellectual property laws. Except as expressly permitted in your license agreement or allowed by law, you may not use, copy, reproduce, translate, broadcast, modify, license, transmit, distribute, exhibit, perform, publish, or display any part, in any form, or by any means. Reverse engineering, disassembly, or decompilation of this software, unless required by law for interoperability, is prohibited.

The information contained herein is subject to change without notice and is not warranted to be error-free. If you find any errors, please report them to us in writing.

If this is software or related documentation that is delivered to the U.S. Government or anyone licensing it on behalf of the U.S. Government, then the following notice is applicable:

U.S. GOVERNMENT END USERS: Oracle programs, including any operating system, integrated software, any programs installed on the hardware, and/or documentation, delivered to U.S. Government end users are "commercial computer software" pursuant to the applicable Federal Acquisition Regulation and agency-specific supplemental regulations. As such, use, duplication, disclosure, modification, and adaptation of the programs, including any operating system, integrated software, any programs installed on the hardware, and/or documentation, shall be subject to license terms and license restrictions applicable to the programs. No other rights are granted to the U.S. Government.

This software or hardware is developed for general use in a variety of information management applications. It is not developed or intended for use in any inherently dangerous applications, including applications that may create a risk of personal injury. If you use this software or hardware in dangerous applications, then you shall be responsible to take all appropriate fail-safe, backup, redundancy, and other measures to ensure its safe use. Oracle Corporation and its affiliates disclaim any liability for any damages caused by use of this software or hardware in dangerous applications.

Oracle and Java are registered trademarks of Oracle and/or its affiliates. Other names may be trademarks of their respective owners.

Intel and Intel Xeon are trademarks or registered trademarks of Intel Corporation. All SPARC trademarks are used under license and are trademarks or registered trademarks of SPARC International, Inc. AMD, Opteron, the AMD logo, and the AMD Opteron logo are trademarks or registered trademarks of Advanced Micro Devices. UNIX is a registered trademark of The Open Group.

This software or hardware and documentation may provide access to or information about content, products, and services from third parties. Oracle Corporation and its affiliates are not responsible for and expressly disclaim all warranties of any kind with respect to third-party content, products, and services unless otherwise set forth in an applicable agreement between you and Oracle. Oracle Corporation and its affiliates will not be responsible for any loss, costs, or damages incurred due to your access to or use of third-party content, products, or services, except as set forth in an applicable agreement between you and Oracle.

The information contained in this document is for informational sharing purposes only and should be considered in your capacity as a customer advisory board member or pursuant to your beta trial agreement only. It is not a commitment to deliver any material, code, or functionality, and should not be relied upon in making purchasing decisions. The development, release, and timing of any features or functionality described in this document remains at the sole discretion of Oracle.

This document in any form, software or printed matter, contains proprietary information that is the exclusive property of Oracle. Your access to and use of this confidential material is subject to the terms and conditions of your Oracle Master Agreement, Oracle License and Services Agreement, Oracle PartnerNetwork Agreement, Oracle distribution agreement, or other license agreement which has been executed by you and Oracle and with which you agree to comply. This document and information contained herein may not be disclosed, copied, reproduced, or distributed to anyone outside Oracle without prior written consent of Oracle. This document is not part of your license agreement nor can it be incorporated into any contractual agreement with Oracle or its subsidiaries or affiliates.

Contents

Preface	xix
Audience	xix
Related Documents	xix
Conventions	xix

What's New in This Guide for Release 11.1.1.9

Part I Using Oracle Business Process Management Workspace

1 Getting Started with Process Workspace

1.1	Understanding Process Workspace	1-1
1.2	Starting and Logging In to Process Workspace	1-3
1.3	Navigating Process Workspace	1-4
1.3.1	About Task States	1-6
1.3.2	About the Process Workspace Tasks Page	1-6
1.3.2.1	Navigating the Process Workspace Tasks Page	1-7
1.3.2.2	Using the Process Workspace Task Details Page	1-10
1.3.2.3	Customizing the Task List in Process Workspace	1-14
1.3.2.4	Searching for Tasks in Process Workspace	1-25
1.3.3	About the Process Workspace Activity Guides Page	1-26
1.3.4	About the Process Workspace Process Tracking Page	1-26
1.3.4.1	Navigating the Process Workspace Process Tracking Page	1-27
1.3.5	About the Process Workspace Standard Dashboards Page	1-29
1.4	Setting Accessibility Preferences in Process Workspace	1-30
1.5	Providing a Digital Certificate in Process Workspace	1-30
1.6	Accessing Administration Capabilities	1-32
1.7	Setting Your Preferences in Process Workspace	1-33

2 Working on Tasks in Process Workspace

2.1	Claiming and Releasing Tasks in Process Workspace	2-1
2.2	Modifying History Approval Options in the Task Details Page	2-2
2.2.1	Combining Repeated Approvals of a Task in the History	2-2
2.2.2	Excluding System Approvals from a Task	2-3
2.2.3	Including New Participants in an Existing Task	2-3
2.3	Reassigning and Delegating Tasks in Process Workspace	2-4

2.4	Adding Comments and Attachments to Tasks in Process Workspace	. 2-5
2.5	Requesting Information in Process Workspace	. 2-8
2.6	Setting Rules, Vacation Periods, and Notification Preferences in Process Workspace .	2-10
2.6.1	How to Create User Rules in Process Workspace	2-10
2.6.2	How to Create Group Rules in Process Workspace	2-12
2.6.3	How to Enable a Vacation Period in Process Workspace	2-14
2.6.4	Specifying Notification Settings	2-15
2.6.4.1	Messaging Filter Rules	2-15
2.6.4.2	Rule Actions	2-16
2.6.4.3	Managing Messaging Channels	2-17
2.6.4.4	Managing Messaging Filters	2-18
2.7	Creating ToDo Tasks in Process Workspace	2-20
2.8	Creating Subtasks in Tasks in Process Workspace	2-21
2.8.1	What You May Need to Know About Creating Subtasks	2-21
2.8.2	How to Create a SubTask	2-22
2.8.2.1	Creating SubTasks from the Actions List	2-22
2.8.2.2	Creating SubTasks from an Existing Task	2-23
2.9	Acting on Tasks that Require Passwords in Process Workspace	2-25
2.10	Acting on a Task that Requires a Digital Signature in Process Workspace	2-25
2.11	Routing Tasks in Process Workspace	2-26
2.12	Evidence Store Service and Digital Signatures	2-28
2.13	Creating Worklist Reports	2-29
2.13.1	How To Create Reports	2-30
2.13.2	What Happens When You Create Reports	2-31
2.13.2.1	1 Unattended Tasks Report	2-32
2.13.2.2	2 Tasks Priority Report	2-33
2.13.2.3	3 Tasks Cycle Time Report	2-33
2.13.2.4	4 Tasks Productivity Report	2-34

3 Working on Process Instances in Process Workspace

3.1	Initiating Process Instances in Process Workspace	3-1
3.2	Searching for Process Instances in Process Workspace	3-2
3.2.1	How to Perform a Basic Search for Process Instances	3-2
3.2.2	How to Perform an Advanced Search for Process Instances	3-3
3.3	Viewing the Details of a Process Instance in Process Workspace	3-3
3.3.1	Viewing the Audit Trail in Process Workspace	3-4
3.3.2	Associating Comments and Attachments with a Process Instance in Process Workspace 3-6	
3.4	Performing a Bulk Operation in Process Workspace	3-6
3.5	Canceling a Process Instance in Process Workspace	3-6
3.6	Customizing the Presentation of Process Instances in Process Workspace	3-7

4 Working with Activity Guides in Process Workspace

4.1	Understanding Activity Guides	4-1
4.1.1	Why Use Activity Guides	4-2
4.2	Navigating Activity Guides	4-2
4.3	Showing or Hiding the Activity Guide Tab Page	4-4

5 Creating and Editing Custom Pages in Process Workspace

5.1	Creating Custom Pages in Process Workspace	5-1
5.1.1	How to Create a Custom Page in Process Workspace	5-2
5.1.2	How to Add an External URL to a Customized Page in Process Workspace	5-2
5.1.3	How to Add an Applications Panel to a Customized Page in Process Workspace	5-2
5.1.4	How to Add a Dashboards Panel to a Customized Page in Process Workspace	5-2
5.1.5	How to Add a Process Instance Panel to a Customized Page in Process Workspace . 5-4	
5.1.6	How to Grant and Manage Access to Customized Pages in Process Workspace	5-4
5.2	Establishing Communication Between Panels in Process Workspace	5-6
5.2.1	How to Establish Communication Between Panels in Process Workspace	5-6
5.3	Editing Custom Pages and Panels in Process Workspace	5-6

6 Using Dashboards in Process Workspace

6.1	Using Standard Dashboards	6-1
6.1.1	Participant Dashboards	6-1
6.1.1.1	Participant Workload Dashboard	6-2
6.1.1.2	Participant Performance Dashboard	6-2
6.1.2	Process Dashboards	6-3
6.1.2.1	Process Workload Dashboard	6-3
6.1.2.2	Process Performance Dashboard	6-4
6.2	Creating Custom Dashboards	6-5
6.2.1	Specifying Graph Content	6-5
6.3	Understanding Data Visibility Rules in Dashboards	6-8
6.3.1	Workload Sampling Data	6-8
6.3.2	Activity and Measurement Sampling Data	6-8
6.3.3	Process Level Sampling Data	6-9

7 Managing Workloads Among Users in Process Workspace

7.1	Managing the Availability of Users in Process Workspace	7-1
7.2	Distributing Work among Users in a Group	7-1
7.3	Setting Rules	7-2
7.3.1	How To Create User Rules	7-2
7.3.2	How To Create Group Rules	7-4
7.4	Setting Assignment Rules for Tasks with Multiple Assignees	7-5
7.5	Reassigning or Delegating Tasks in Process Workspace	7-6
7.6	Using Reports to Manage Workflows in Process Workspace	7-7
7.7	Using Dashboards to Manage Workflows in Process Workspace	7-8
7.8	Handling Escalations in Process Workspace	7-8

8 Working with Adaptive Case Management in Process Workspace

8.1	About Adaptive Case Management	8-1
8.2	Navigating Cases	8-1
8.2.1	About the Case Page	8-2
8.2.2	About the Navigation Bar	8-2

8.2.3	Using the Navigation Bar	8-3
8.2.3.1	Selecting a Case	8-3
8.2.3.2	Closing a Case	8-4
8.2.3.3	Viewing a Closed Case	8-4
8.2.3.4	Adding a Comment to a Case	8-4
8.2.3.5	Raising an Event on a Case	8-5
8.3	Viewing Case Information	8-5
8.3.1	Viewing Case Instances	8-5
8.3.2	Viewing the Case List	8-6
8.3.2.1	Viewing Hidden Panels	8-6
8.3.2.2	Case List Panel	8-6
8.3.2.3	Sorting the Case List	8-8
8.3.2.4	Searching for Cases	8-8
8.3.3	Working With Case Activities	8-9
8.3.3.1	Activities Panel	8-9
8.3.3.2	Filtering the Activities Displayed	8-10
8.3.3.3	Initiating an Activity	8-10
8.3.3.4	Viewing Activity Details	8-10
8.3.4	Viewing the Audit Trail	8-10
8.3.4.1	Audit Trail Panel	8-11
8.3.4.2	Viewing Events in the Audit Trail	8-11
8.3.4.3	Updating Event Permissions	8-11
8.3.5	Data Panel	8-12
8.3.5.1	Working with Data Forms in the Data Panel	8-12
8.3.6	Documents Panel	8-12
8.3.6.1	Documents Panel	8-12
8.3.6.2	Viewing Documents in the Documents Panel	8-13
8.3.6.3	Performing Actions on a Document	8-13
8.3.7	Viewing Milestones	8-13
8.3.7.1	Milestones Panel	8-13
8.3.7.2	Changing a Milestone's Status	8-14
8.3.8	Working with Stakeholders	8-14
8.3.8.1	Stakeholders Panel	8-14
8.3.8.2	Adding Stakeholders	8-15
8.3.8.3	Editing Stakeholders	8-15
8.4	Working on Cases	8-15

Part II Administering Process Workspace

9 Managing Process Workspace

9.1	What You May Need to Know When Managing Process Workspace	9-1
9.2	Setting the Display of the Work Items Panel in Process Workspace	9-1
9.2.1	How to Specify the Login Page Realm Label in Process Workspace	9-2
9.2.2	How to Specify the Resource Bundle in Process Workspace	9-3
9.2.3	How to Specify the Language Locale Information in Process Workspace	9-3
9.2.4	How to Specify the User Name Format	9-4
9.2.5	How to Specify a Branding Logo in Process Workspace	9-4

9.2.6	How to Specify the Branding Title in Process Workspace	9-5
9.2.7	How to Choose a Skin in Process Workspace	9-5
9.2.8	How to Enable Customized Applications and Links in Process Workspace	9-6
9.2.9	How to Specify an Image for a Task Action	9-6
9.2.10	How to Specify, Show, or Hide the Default Tab Pages in Process Workspace	9-7
9.3 Ad	ministering Approval Groups in Process Workspace	9-7
9.3.1	How to Use Task Forms	9-8
9.3.1.1	Header View	9-9
9.3.1.2	Task Payload View	9-10
9.3.1.3	Task History View	9-11
9.3.1.4	Comments and Attachments View	9-13
9.3.2	How to Create Mapped Attribute Labels	9-13
9.3.2.1	Importing and Exporting Attribute-Label Definitions	9-14
9.3.2.2	Internationalizing Attribute Labels	9-14
9.3.3	How to View Approval Groups	9-15
9.3.4	How to Search for an Approval Group	9-15
9.3.5	How to Add a Static Approval Group	9-16
9.3.6	How to Add a New Member to a Static Approval Group	9-17
9.3.7	How to Delete a Member from an Approval Group	9-18
9.3.8	How to Move an Approval Group's Members	9-19
9.3.9	How to Nest Approval Groups	9-19
9.3.10	How to Rename an Approval Group	9-19
9.3.11	Using Dynamic Approval Groups	9-20
9.3.11.1	How to Write a Custom Dynamic Approval Group Class	9-20
9.3.11.2	How to Register the Custom Dynamic Approval Group Class	9-21
9.3.11.3	How to Make the Custom Dynamic Approval Group Class Available	9-21
9.3.11.4	How to Add a Dynamic Approval Group	9-21
9.3.12	How to Delete an Approval Group	9-22
9.4 Ma	naging Mapped Attributes (Flex Fields) in Process Workspace	9-22
9.4.1	How to Specify Mapped Attributes	9-23
9.5 Usi	ing Task Configuration in Process Workspace	9-24
9.5.1	About Task Configuration in Process Workspace	9-25
9.5.2	How to Edit Workflow Task Settings	9-27
9.5.3	How to Specify Routing Settings	9-27
9.5.4	How to Specify an Expiration Policy	9-27
9.5.5	How to Extend an Expiration Policy Period	9-28
9.5.6	How to Specify an Escalation Policy	9-28
9.5.7	How to Specify Participant Notification Preferences	9-29
9.5.7.1	How to Make Email Messages Actionable	9-29
9.5.7.2	How to Secure Notifications to Exclude Details	9-29
9.5.7.3	How to Set Up Reminders	9-30
9.5.7.4	How to Notify Recipients of Changes to Task Status	9-30
9.5.7.5	How to Edit the Notification Message	9-32
9.5.7.6	How to Prevent Multiple Email Notifications for a Human Task Event	9-32
9.5.8	How to Enable Task Access	9-32
9.5.9	How to Specify Actions for Acting Upon Tasks	9-33
9.5.10	How to View Settings for Rules	9-34

9.5.11	How to Add a Rule to a Ruleset in Process Workspace	9-35
9.5.11.	1 How to Define Rule Conditions	9-36
9.5.11.2	2 How to Add Actions	9-36
9.5.12	How to Add and Customize Decision Tables	9-38
9.5.12.	1 How to Customize an Existing Decision Table in Process Workspace	9-38
9.5.12.2	2 How to Edit an Action (Multiple Parameters: All Rules at a Time)	9-39
9.5.12.3	3 How to Edit an Action (Single Cell Editing)	9-40
9.5.12.4	4 How to Add a New Decision Table to a Ruleset in Process Workspace	9-41
9.5.13	How to Add a Variable to a Rule	9-41
9.6	Using Evidence Search in Process Workspace	9-42
9.7	Managing Exceptions in Process Workspace	9-42
9.8	Using the User Metadata Migration Utility	9-43

10 Managing Your Organization in Process Workspace

tion Entities in Process Workspace	10-1
kspace	10-2
orkspace	10-3
s (as an Administrator) in Process Workspace	2
	10-6
	10-7
or a Role	10-7
	10-8
ecific Views and Layouts	10-9
Vorkspace	10-9
Members of an Organization Unit	10-9
Tasks	10-9
	10-10
	10-14
	10-15
Process Workspace	10-15
Workspace	10-18
	tion Entities in Process Workspace kspace

11 Modifying Running Process Instances in Process Workspace

11.1	Altering the Flow of Running Process Instances in Process Workspace	11-1
11.1.1	How to Alter the Flow of Running Process Instances in Process Workspace	11-1
11.2	Migrating Process Instances in Process Workspace	11-3
11.2.1	How to Migrate a Process Instance in Process Workspace	11-4
11.2.2	How to Resume a Suspended Component	11-5
11.3	What You May Need to Know About Redeploying a Process	11-7
11.4	Moving Running Composite Instances from One Revision to Another	11-8
11.4.1	Overview	11-8
11.4.2	Important Considerations	11-8
11.4.3	How to Perform a Migration	11-9
11.4.4	Restrictions for Components	11-9
11.4.4.	1 BPEL Components	11-9
11.4.4.2	2 BPMN Components	11-10

Human Task Components	11-10
Business Rules Components	11-10
Mediator Components	11-10
Ant Tasks Elements	11-11
Migration Elements	11-12
	Human Task Components Business Rules Components Mediator Components Ant Tasks Elements Migration Elements

Part III Using Oracle Business Process Management Process Spaces

12 Getting Started with Process Spaces

12.1	Understanding Process Spaces	12-1
12.1.1	Oracle WebCenter Portal Group Spaces for Business Process Management	12-1
12.2	Starting and Logging In to Process Spaces	12-2
12.3	Navigating Process Spaces	12-2
12.3.1	About the Process Spaces Process Workspace	12-2
12.3.1.1	The Process Workspace Home Page	12-2
12.3.1.2	2 The Process Workspace Process Tracking Page	12-3
12.3.2	The Modeling Space in Process Spaces	12-4
12.3.3	The Process Instance Space in Process Spaces	12-4

13 Using and Customizing a Process Workspace in Process Spaces

13.1	The Process Workspace Home Page in Process Spaces	13-1
13.2	Acting on Tasks in Process Spaces	13-2
13.3	Filtering Tasks in Process Spaces	13-2
13.4	Adding Members to Process Spaces	13-2
13.5	Adding Documents to Process Spaces	13-2
13.6	Participating in Discussions in Process Spaces	13-2
13.7	Viewing Recent Activity in Process Spaces	13-3
13.8	Initiating a Process Instance in Process Spaces	13-3
13.9	Filtering Process Instances in Process Spaces	13-3
13.10	Canceling a Process Instance in Process Spaces	13-3
13.11	Adding Documents to Process Instances in Process Spaces	13-3
13.12	Using and Customizing Dashboards in Process Spaces	13-4
13.13	Creating an On-Demand Group Space for an Instance in Process Spaces	13-4
13.14	Customizing Process Workspace in Process Spaces	13-4
13.15	Subscribing to a Process Workspace Group Space in Process Spaces	13-4

14 Using and Customizing a Process Modeling Space in Process Spaces

14.1	About Modeling Spaces in Process Spaces	14-1
14.2	The Process Modeling Space Page	14-2
14.3	Customizing a Process Modeling Space in Process Spaces	14-2
14.4	Adding Members to the Modeling Space in Process Spaces	14-3
14.5	Adding Documents to the Modeling Space in Process Spaces	14-3
14.6	Participating in Modeling Space Discussions in Process Spaces	14-3
14.7	Subscribing to a Modeling Space in Process Spaces	14-3

15 Using and Customizing a Process Instance Space in Process Spaces

15.1	The Process Instance Space in Process Spaces	15-1
15.2	Adding Comments and Attachments to a Process Instance in Process Spaces	15-2
15.3	Tracking a Process Instance in Process Spaces	15-2
15.4	Creating a Calendar for a Process Instance in Process Spaces	15-3
15.5	Adding a Document to a Process Instance Space in Process Spaces	15-3
15.6	Participating in a Discussion in a Process Instance Space in Process Spaces	15-3
15.7	Customizing a Process Instance Space in Process Spaces	15-3

16 Creating Custom Dashboards in Process Spaces

16.1	Understanding Custom Dashboards in Process Spaces	16-1
16.2	Creating a Custom Dashboard in Process Spaces	16-1
16.2.1	How to Create a Custom Dashboard in Process Spaces	16-1

17 Performing Administration in Process Spaces

17.1	Understanding Roles and Permissions in Process Spaces	17-1
17.2	Understanding Task Flows in Process Spaces	17-1
17.3	Editing Pages in Process Spaces	17-1
17.4	Configuring a Group Space in Process Spaces	17-2

Part IV Appendices

A Creating Custom ADF Applications with Oracle Business Process Management Workspace Task Flows

A.1	User-Related Task Flows	A-1
A.1.1	Task List Task Flow	A-2
A.1.1.1	federatedMode	A-2
A.1.1.2	federatedServers	A-2
A.1.1.3	showServerColumn	A-3
A.1.1.4	wfCtxID	A-3
A.1.2	Certificates Task Flow	A-3
A.1.3	Reports Task Flow	A-4
A.1.4	Application Preferences Task Flow	A-4
A.1.5	Flex Fields Task Flow	A-4
A.1.6	Rules Task Flow	A-5
A.2	Process Task Flows	A-5
A.2.1	Process Application Task Flow	A-5
A.2.2	Process Instances Task Flow	A-6
A.2.3	Process Instance Details Task Flow	A-7
A.3	Process Analytics Task Flows	A-7
A.3.1	Dashboard Widget Task Flow	A-8
A.3.2	Application Roles Task Flow	A-8
A.3.3	Organization Units Task Flow	A-9
A.3.4	Parametric Roles Task Flow	A-10
A.3.5	Business Calendar and Holiday Lists Task Flow	A-10
A.3.6	Extended User Properties Task Flow	A-10

A.4	.JAR files for the Task Flows	
A.5	Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows A-10	
A.5.1	Sample 1: Creating an Application with an Embedded Tasklist Task Flow	A-10
A.5.2	Sample 2: Creating an ADF Page with Task Flows	A-23
A.6	Sample: Wf_client_config.xml File	A-27
A.7	Sample: Bean with the Method Returning the WorkflowServicesClientConfigurationType Object A-29	
Custo	mizing a Resource Bundle	

B.1 How to Customize a Resource Bundle B-1

C Customizing Applications in Process Workspace and the Worklist Application

В

C 1	How to Customize an Application	C	1
0.1	The customize an application	0	•

D Deploying Images and JAR files as Part of a Shared Library JAR file

D.1	How to Deploy Images and IAR Files as Part of a Shared Library IAR File	D-1

List of Examples

9–1	Implementation Class	9-20
A–1	Wf_client_config.xml	A-27

List of Figures

1–1	Example of Process Workspace Home	1-4
1–2	Process Workspace Tasks Page	1-8
1–3	Example of a Task Details Page	1-10
1–4	History: Graphical View	1-12
1–5	Filters: Assignee, Status, Search, and Advanced Search	1-14
1–6	Worklist Advanced Search: Definition Page	1-16
1–7	Worklist Advanced Search: Display Page	1-17
1–8	Workspace Advanced Search Definition Tab - Display Tab Not Available	1-17
1–9	Saving a View	1-18
1–10	Worklist Advanced Search	1-18
1–11	Adding Filters for an Advanced Search on Tasks	1-19
1–12	Advanced Search	1-21
1–13	Task Type Browser	. 1-21
1-14	Views Menu	1-22
1_15	Creating a Worklist View	1-23
1_16	Displaying Fields in a Worklist View	1 20 1-24
1_17	Customizing Fields in a Worklist View	1 <u>2</u> 4
1_17	Sample Activity Cuide	1-20 1-26
1 10	Process Workspace Process Tracking Dage	. 1-20
1 20	Process Workspace Frocess fracking Fage	1-27
1-20	Frocess workspace Standard Dashboards Fage	. 1-29
1 00	Selecting Accessionity	1-30
1-22	Oploading a Certificate	1-31
2-1	Claiming a Task	2-2
2-2	History Pane - Options	2-3
2-3	Keassigning a Task	2-4
2-4	Uploading an Attachment to Oracle WebCenter Content	2-6
2-5	Possible Destination Folders	2-7
2-6	Saving an Attachment to a Task	2-8
2-7	Requesting Information	2-8
2–8	Requesting Information from Past Approvers or Another User, or Pushing the Task	Back
	2-9	
2-9	Identity Browser	2-9
2–10	Configuration Areas: My Rules Page	2-10
2–11	Creating a Group Rule	2-13
2–12	My Rules - Vacation Period	2-14
2–13	Identity Browser Window	2-15
2–14	Messaging Channels	2-17
2–15	Messaging Filters	2-18
2–16	Adding a Messaging Filter	2-19
2–17	The Create To-Do Task Dialog	2-20
2–18	Creating a To-Do Task	2-21
2–19	Selecting to Create a Sub Task	2-22
2–20	Create Subtask Dialog Box - Task List	2-23
2–21	Selecting to Create a Sub Task from a the Task Details Page	2-24
2–22	Create Subtask Dialog Box - Task Details Page	2-24
2–23	Acting on a Task That Requires a Password	2-25
2–24	Digital Signature Task Details	2-26
2–25	Ad Hoc Routing	2-27
2–26	Routing a Task	2-27
2–27	Providing Consensus Information	2-27
2–28	The Evidence Store	2-29
2–29	Creating a Report	2-31
2–30	Oracle BPM Worklist Reports	2-31
2–31	Unattended Tasks Report—Input Page for Task Analysis	2-31

2–32	Report Display—Table Format, Bar Chart Format, and Report Inputs	2-32
2–33	Unattended Tasks Report	2-32
2–34	Tasks Priority Report	2-33
2–35	Tasks Cycle Time Report	2-34
2–36	Tasks Productivity Report	2-34
3–1	Search Results	3-2
3–2	Work Item Details Panel	3-4
3–3	Selecting the Audit Details View	3-5
3–4	Selecting Activities for Audit Details	3-5
3-5	Displaying Activity Input and Output Messages	3-5
4_1	Sample Activity Cuide	4-2
	Adding an Applications Panol	5-2
5_2	Adding a Dashboards (Charte) Papal	5-2
52	Adding a Dashboards (Charles) I aller	5/
5-3 E 1	Auding a Flocess fistalices Faller	5-4
0-4 c 1	Specifying Access to a Custom Fage	- 0-0 - 0 0
0-1	Participant Workload Dashboard	0-2
6-2	Participant Performance Dashboard	6-3
6-3	Process Workload Dashboard	6-4
6-4	Process Performance Dashboard	6-5
/-1	Creating a Rule	7-2
7–2	Creating a Group Rule	7-5
7–3	Reassigning a Task	7-6
7–4	Reassigning a Task	7-6
7–5	Claiming a Task	7-7
7–6	Oracle Business Process Management Workspace Reports	7-7
8–1	Navigation Bar	8-2
8–2	Case Listing Drop-down List	8-4
8–3	Case Page	8-6
8–4	Case List Panel Advanced Search	8-9
9–1	Setting Application Preferences	9-2
9–2	Specifying Language Local Information	9-3
9–3	User Name Format	9-4
9–4	Show Standard Tabs Selection	9-7
9–5	Task Form	9-9
9–6	Header View in Process Workspace	9-9
9–7	Task Pavload	9-11
9–8	Task History in Process Workspace	9-12
9_9	Task History - Additional Actions	9-12
9–10	Comments and Attachments	9-13
9_11	Elevield Mapping: Protected	9-14
9_12	Process Workspace: Approval Croup Detail	0-15
0_12	Licer Name Search Regulte	0_16
9-13	Create Approval Croup: Solast Statia Croup	0 17
9-14	Approval Group. Select Static Group	0 10
9-15	Approval Group Structure: Multiple Members	9-10
9-10	Code for Dynamic Approval Group Class	9-21
9-17	Specifying a Mapped Attribute	9-23
9–18	Creating a Label	9-24
9–19	Task Configuration: Editable Task Page	9-26
9–20	Task Contiguration: Rules Tab Page	9-26
9–21	Editing a Task Rule	9-34
9–22	Adding a Rule	9-35
9–23	Adding Actions to a Rule	9-37
9–24	Available Actions for Rules	9-37
9–25	Delete, Up, Down, Copy, Cut and Paste Commands	9-38
9–26	Adding Actions to a Decision Table	9-39

9–27	Decision Table with Editable Actions	9-39
9–28	Editing an Action from the Decision Table	9-40
9–29	Action Editor	9-40
9–30	Single Cell Editing	9-40
9–31	Adding a Decision Table	9-41
9–32	Searching the Evidence Store	9-42
10–1	Holiday Rules	10-2
10-2	Calendar Rules	10-3
10-3	Creating Rules for Another User or Group	10-5
10-4	Defining Rules for Another User or Group	10-6
10-5	Setting the Escalation Path	10-8
10-6	Defining Parameters	10-10
10-7	Selecting Defined Parameters as Conditions of Users with Parametric Roles	10-12
10_8	Customizing I DAP Attributes in Enterprise Manager	10-12
10-0	Viowing Attributes Tab for InsProvider	10-13
10-3	Viewing Operations Tab for InsProvider	10-10
10-10	Parameters for add Searchable Lear Attribute Man	10-14
10-11	Farmele Extended Lear Properties Dage	10-14
10-12	Adding on Organization Unit	10-10
10-13	Adding an Organization Unit	10-19
11-1	Altering the Flow of a Process Instance	11-2
11-2	Migrating a Process Instance	11-4
11-3	Migration Confirmation: Failure Message	11-5
11-4	Pending Components Panel in the Process Tracking Tab Page	11-6
11-5	Resume Component Dialog Box	11-6
11–6	Setting the force.deploy Property in Oracle JDeveloper	11-8
11–7	Example of Ant Task	11-12
12–1	Process Spaces Home Page	12-3
12–2	Process Spaces Process Tracking Page	12-3
12–3	Process Spaces Modeling Space Page	12-4
12–4	Process Spaces Process Instance Page	12-5
13–1	Process Spaces Home Page	13-2
14–1	Process Spaces Modeling Space Page	14-2
15–1	Process Spaces Process Instance Page	15-2
A–1	Process Workspace Tasks Page	A-2
A–2	Uploading a Certificate	A-4
A–3	Applications Panel	A-6
A–4	Work Items Details Panel	A-7
A–5	Process Performance Dashboard	A-8
A–6	Role Details Panel	A-9
A–7	Organization Units Details Panel	A-9
A–8	Creating New Oracle Fusion Middleware Application	A-11
A–9	Adding BPM Worklist Component Library.	A-12
A–10	Adding BPM Services Library	A-12
A–11	Adding the WSRP Container Library	A-13
A–12	Creating a a ISPX File	A-14
A–13	Choosing adflibTaskListTaskFlow.jar	A-15
A-14	Oracle IDeveloper with TaskList Task Flow on the ISPX Test Page	A-15
A–15	Passing Task Flow Parameters	A-16
A–16	Editing the WAR Deployment Profile	A-17
Δ_17	Navigating to Secure the Application	Δ_18
Δ_18	Selecting ADF Authentication	Δ_10
Δ_10	Selecting HTTP Basic Authentication	Δ_20
Δ_20	Deploying the Application	A-20
Δ_21	Deploying the Application Server	A-00
A-21	Deproying to Application Server	A-22
A-22	I IUCESS WUIKSPACE TASK LIST	A-23

A–23	Configuring the JSPX Page	A-24
A–24	Split Panel in the JSPX Page	A-24
A–25	Creating SampleController.java	A-25
A–26	Specifying the Managed Bean	A-25
A–27	Adding and Specifying a Switcher	A-27
B–1	Creating an ADF Project in JDeveloper	B-1
B–2	Creating a Java Class	B-2
B–3	Including Worklist Application in the Library	B-3
B–4	Modifying the CustomResourceBundle Class	B-3
B–5	Creating a Deployment Profile for a JAR File	B-4
B–6	Verifying JAR Deployment Profile	B-5
B–7	Editing the JAR Deployment Profile	B-5
B–8	Deploying the Resource Bundle	B-6
C–1	Creating an ADF Project in JDeveloper	C-1
C–2	Including Worklist Application in the Library	C-2
C–3	Links Panel in Process Workspace	C-4
C–4	Customized Columns in Task Details Pane	C-4
D–1	Stopping Worklist Application	D-2
D–2	Restarting Worklist Application	D-2
D–3	Setting Application Preferences	D-3

List of Tables

1–1	Users of Oracle Business Process Management Workspace	. 1-2
1–2	Toolbar for Tasks, Activity Guides, Process Instances, Dashboards, and More	1-4
1–3	Toolbar for Oracle Business Process Management Workspace	. 1-5
1–4	Task State Descriptions	1-6
1–5	Views, Applications, and Favorites List	. 1-8
1–6	Components of the Tasks Page	. 1-9
1–7	Icons for Task History	1-13
1–8	System Task Actions	1-13
1–9	Advanced Search: Conditions	1-19
1–10	Categories in the Administration Areas	1-32
1–11	Preferences	1-33
2–1	Data Types Supported by Messaging Filters	2-15
2–2	Predefined Attributes for Messaging Filters	2-16
2–3	Worklist Report Types	2-30
3–1	Add View Status List	3-3
3–2	Visibility into Process Instances	3-4
4–1	Process List	4-3
4–2	Activity Guide Navigation Panel	. 4-3
6–1	Standard Dimensions Available	. 6-7
6–2	Standard Measurements Available	. 6-7
6–3	Operation Types Allowed	. 6-7
9–1	Header Fields	9-9
9–2	Header Actions	9-10
9–3	Edit Future Participants List Actions	9-12
9–4	Task History Actions	9-13
9–5	Highest Privilege Levels for Users of Task Content	9-33
10–1	Holiday Types	10-2
14–1	Process Spaces Modeling Spaces Panels	14-2
15–1	Panels in the Process Instance Space	15-2
A–1	Process Applications Task Flow	A-6
A–2	Process Instances Task Flow	A-6
A–3	Process Instances Task Flow	A-7
A–4	Dashboard Widget Task Flow	A-8
A–5	JAR Files for Task Flows	A-10

Preface

Oracle Fusion Middleware User's Guide for Oracle Business Process Management describes how to interact with tasks and business processes in both Oracle Business Process Management Workspace and Oracle Business Process Management Process Spaces.

Audience

This document is intended for all users of both Oracle Business Process Management Workspace and Oracle Business Process Management Process Spaces.

Related Documents

For more information, see the following Oracle resources:

- Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management
- Oracle Fusion Middleware Business Process Composer User's Guide for Oracle Business Process Management
- Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite

Conventions

The following text conventions are used in this document:

Convention	Meaning
boldface	Boldface type indicates graphical user interface elements associated with an action, or terms defined in text or the glossary.
italic	Italic type indicates book titles, emphasis, or placeholder variables for which you supply particular values.
monospace	Monospace type indicates commands within a paragraph, URLs, code in examples, text that appears on the screen, or text that you enter.

What's New in This Guide for Release 11.1.1.9

There are no major changes in this guide for Release 11.1.1.9.

For a list of known issues (release notes), see the "Known Issues for Oracle SOA Products and Oracle AIA Foundation Pack" at http://www.oracle.com/technetwork/middleware/docs/soa-aiafp-knownissuesind ex-364630.html.

Screens shown in this guide may differ from your implementation, depending on the skin used. Any differences are cosmetic.

Part I

Using Oracle Business Process Management Workspace

This part contains the following chapters:

- Chapter 1, "Getting Started with Process Workspace"
- Chapter 2, "Working on Tasks in Process Workspace"
- Chapter 3, "Working on Process Instances in Process Workspace"
- Chapter 4, "Working with Activity Guides in Process Workspace"
- Chapter 5, "Creating and Editing Custom Pages in Process Workspace"
- Chapter 6, "Using Dashboards in Process Workspace"
- Chapter 7, "Managing Workloads Among Users in Process Workspace"
- Chapter 8, "Working with Adaptive Case Management in Process Workspace"

1

Getting Started with Process Workspace

This chapter tells you how to start and log in to Process Workspace, navigate its interface, and configure your preferences.

Process Workspace is a customizable web-based interface enabling users to access and manage tasks and process instances according to their roles and responsibilities.

To help you get started with Process Workspace, this chapter contains these topics:

- Section 1.1, "Understanding Process Workspace"
- Section 1.2, "Starting and Logging In to Process Workspace"
- Section 1.3, "Navigating Process Workspace"
- Section 1.4, "Setting Accessibility Preferences in Process Workspace"
- Section 1.5, "Providing a Digital Certificate in Process Workspace"
- Section 1.6, "Accessing Administration Capabilities"
- Section 1.7, "Setting Your Preferences in Process Workspace"

For information about troubleshooting human workflow issues, see *Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite*, "Human Workflow Troubleshooting".

1.1 Understanding Process Workspace

Process Workspace enables process participants to interact with the applications you create using Oracle BPM. The Process Workspace user interface provides tabs for each of the following:

- Tasks: This page enables process participants to view and work with their assigned tasks.
- Process Tracking: This page enables process participants to view running process instances.
- Standard Dashboards: This page provides out-of-the-box dashboards for monitoring process performance, task performance, and workload.
- Custom Dashboards: This page enables process participants to define and use custom dashboards based on the measurement data generated by process instances.
- Case Management: Adaptive Case Management supports the modeling and running of very flexible and data-intensive business processes.

Process Workspace also enables business administrators to configure and maintain organizations and roles. For more information, see "Modeling Your Organization" in the *Modeling and Implementation Guide for Oracle Business Process Management*.

Table 1–1 lists and describes the users for whom Process Workspace is designed.

User Profile	Description
Process Participants	This is the most basic profile, common to all users. It enables you to:
	 Organize, find, claim, and perform work assigned or delegated either to you or to your group
	 Delegate work to another user
	 Create and manage personal rules—for example, whom to send tasks to when you are on vacation
	 Define and share views
	 Manage personal preferences
	 View reports and dashboards
	These users may be spread throughout the organization or work outside the company.
	A process participant may be one who not only initiates the process, but can also track its progress and, if appropriate, withdraw it.
Process Owners	Users who manage one or more business process. They act on tasks belonging to the process but assigned to other users, in addition to their own end-user access. They want to know how process instances are moving in those processes. To get this perspective, they typically use dashboards.
	These users usually reside on the business side of the house as part of the operations organization. They may also be supervisors.
Administrators	Includes two types of administrative users:
	 Business administrators: Users responsible for administering Process Workspace—for example, adding mapped attributes, changing the look and feel, or ensuring that the web application is both running and accessible. Administers tasks that are in an errored state, for example, tasks that must be reassigned or suspended. They can also change application preferences and map attributes, and manage rules for any user or group, in addition to their own end-user access. They typically need access to Fusion Middleware Control Console as well as to the computers where Process Workspace resides.
	 Process administrators: Users responsible for administering

 Table 1–1
 Users of Oracle Business Process Management Workspace

 Process administrators: Users responsible for administering processes—for example, creating shared views; assigning or removing roles, permissions, and groups; and mapping roles.

User Profile	Description
Supervisors	Users with other process participants reporting to them. To ensure that process instances execute as expected, supervisors can:
	 Access staff work
	 Reassign and delegate work as needed
	 Handle escalations from individuals reporting to them
	 Generate reports regarding work distribution and productivity
	 Specify extended user properties

Table 1–1 (Cont.) Users of Oracle Business Process Management Workspace

Process analysts typically use the Oracle Business Process Analysis Suite or Business Process Composer to create process models. They can also use the process analyst role within Oracle BPM Studio.

Process developers typically use Oracle BPM Studio to model and implement the components of a business application. They may occasionally use Business Process Composer for modeling basic processes.

1.2 Starting and Logging In to Process Workspace

To start Process Workspace, you must have installed Oracle SOA Suite, and the Service Integration server must be running. See your installation documentation for more information.

To start and log into Process Workspace:

1. Go to

http://host_name:port_number/bpm/workspace/

The host_name is the name of the host computer on which Oracle SOA Suite is installed.

The port_number is the one used at installation.

2. Enter the user name and password.

You can use the preseeded user name to log in as an administrator. If you have loaded the demo user community in the identity store, then you can use other users such as jstein or jcooper.

The user name and password must exist in the user community provided to JAZN.

3. Click Login.

Notes:

- If you expect to have a large number of users using Process Workspace simultaneously, then set the session time-out value to no more than 5 minutes. To do this, expand the Process Workspace .ear file, open the web.xml file, and update the time-out parameter.
- If user identities are stored in Microsoft Active Directory, then at times authentication can fail because Oracle WebLogic Server, where Process Workspace is running, cannot communicate with the Active Directory installation. This can occur when Oracle WebLogic Server and Active Directory are in separate domains.

To resolve this issue, add the entry for the Active Directory host's address to the /etc/hosts file on machine where Process Workspace is running.

1.3 Navigating Process Workspace

Figure 1–1 shows an example of Process Workspace Home page as it appears by default when you log in.



Figure 1–1 Example of Process Workspace Home

At the very top, you see two toolbars. You use the top-center toolbar to act on tasks, process instances, dashboards, activity guides, reports, and to add custom pages. The links to these parts of Process Workspace are listed and described in Table 1–2.

 Table 1–2
 Toolbar for Tasks, Activity Guides, Process Instances, Dashboards, and More

Toolbar Link	Description
Tasks	The Tasks page displays tasks for the user based on the user's permissions and assigned groups and roles.
Process Tracking	The Process Tracking page enables you to interact with business processes based on the roles assigned to you.

Toolbar Link	Description
Standard Dashboards	Dashboards provide analytical information to monitor process execution as well as user workloads. They provide summary information and enable you to drill down to instance-specific information.
More: Reports	 Reports provide analysis of various aspects of tasks and their lifecycles. The following reports are available: Unattended Tasks Report, Tasks Priority Report, Tasks Cycle Time Report, Tasks Productivity Report, and Tasks Time Distribution Report. For more information about reports, see Section 2.13.1, "How To Create Reports."
	 Activity guides organize your tasks into milestones. Each milestone is a specific set of tasks for you to complete, either on your own or in collaboration with others.
	 Process Workspace provides its own default layout, but you can customize that layout by adding new pages and panels. For more information, see Chapter 5, "Creating and Editing Custom Pages in Process Workspace."

Table 1–2 (Cont.) Toolbar for Tasks, Activity Guides, Process Instances, Dashboards,

The toolbar at the top right is for configuring and acting on Process Workspace itself—for example, logging out, and setting user preferences. If you are an administrator, then an **Administration** link also appears. Administrative tasks can include choosing to show or hide the default tabs, or specifying which tab page is displayed by default when the user logs in.

Table 1–3 describes the pages you access through the **Home**, **Administration**, and **Preferences** links.

Page	Description
Home	As described in Table 1–2, the logged-in user's list of tasks, details for a selected task, and all the functions needed to start acting on a task are provided.
Administration	The following administrative functions are available:
	 Organization administration
	 Specifying mapped attributes
	 Specifying application preferences
	 Administering tasks, including configuring tasks, performing evidence search, and administering approval groups
Preferences	Preference settings include:
	 Setting rules for users or groups, including vacation rules, and setting vacation periods
	 Uploading certificates
	 Specifying user notification channels and message filters
	Specifying accessibility for screen readers, contrast colors, and font size

Table 1–3 Toolbar for Oracle Business Process Management Workspace

For more information about creating custom pages, see Chapter 5, "Creating and Editing Custom Pages in Process Workspace".

For information on specifying, showing, and hiding default tabs, see Section 9.2.10, "How to Specify, Show, or Hide the Default Tab Pages in Process Workspace".

This section contains these topics:

- Section 1.3.1, "About Task States"
- Section 1.3.2, "About the Process Workspace Tasks Page"
- Section 1.3.3, "About the Process Workspace Activity Guides Page"
- Section 1.3.4, "About the Process Workspace Process Tracking Page"
- Section 1.3.5, "About the Process Workspace Standard Dashboards Page"

1.3.1 About Task States

Every task has a state value assigned to it at all times to indicate the status of the task. As tasks are created, processed, and closed, their state value changes appropriately, depending on what is happening with the task.

Task State	Description	Common Action From State
ALERTED	When there is a recoverable fault that has happened	Reassign
	with task assignment, the task is moved to ALERTED state and assigned to the error assignee	Route
	Suite and assigned to the error assignee.	Error
		Withdraw
ASSIGNED	Indicates that a task has been assigned to a user, role or group and its ready for action by a user.	Claim (if assigned to a group/role/set of users).
		Reassign
		Custom actions, such as Approve, Reject
		Escalate
		Route
		Delegate
		Request information
DELETED	The task is complete. This is a terminal state.	
ERRORED	The task is past its expiration date and has expired. This is a terminal state.	
INFO_REQUESTED	Information is requested on a task.	Submit information, which moves the task back to ASSIGNED state.
OUTCOME_UPDATED	A short-lived state to denote that a user has acted on a task. This state is used primarily to keep a historical record of the action.	
STALE	When the owning SOA composite or BPM project is undeployed, all instances of that version of the composite or project are moved to STALE state.	
SUSPENDED	Processing has stopped on the task. No action can be performed on the task in this state, and timers are frozen.	Resume. This moves the task back to ASSIGNED state.
WITHDRAWN	The task is no longer needed. This is a terminal state.	

Table 1–4 Task State Descriptions

1.3.2 About the Process Workspace Tasks Page

After a user logs in, the Home page displays the My Tasks page by default.

You can configure Process Workspace to display another page when a user logs in. This is explained in Section 9.2.10, "How to Specify, Show, or Hide the Default Tab Pages in Process Workspace" on page 9-7.

The Tasks page displays tasks for the user based on the user's permissions and assigned groups and roles.

The Tasks page enables you to do the following:

- Perform authorized actions on tasks in the worklist, acquire and check out shared tasks, define personal to-do tasks, and define subtasks.
- Filter tasks in a worklist view based on various criteria.
- Work with standard work queues, such as high priority tasks, tasks that are due soon, and so on. Work queues enable users to create a custom view to group a subset of tasks in the worklist—for example, high priority tasks, tasks due in 24 hours, expense approval tasks, and more.
- Define custom work queues.
- Gain proxy access to part of another user's worklist.
- Define custom vacation rules and delegation rules.
- Enable group owners to define task dispatching rules for shared tasks.
- Collect a complete workflow history and audit trail.
- Use digital signatures for tasks.
- Display a pop-up showing details about the process instance related to this task
- The admin can map the task outcomes of all the task types (approve, reject, complete, resolve, and so forth) to the images available through the workflow global configuration in Enterprise Manager. After mapping the task outcome to the images, you can take task outcome actions from the worklist task list tool bar by clicking on the available images.

Note: If you change a user's privileges in Fusion Middleware Control Console while the user is logged in to Process Workspace, the changes take effect only after a subsequent login by the user. This is true for situations in which there are two active worklist sessions: one in which the user is logged in before the privileges are changed, and another in which the same user logs in after the privileges are changed. In the first case, the changes to the user's privileges do not take effect while the user is logged in. In the second case, when the user logs in to the second instance of Process Workspace, the changes to the user's privileges do take effect.

1.3.2.1 Navigating the Process Workspace Tasks Page

Figure 1–2 shows a sample Process Workspace tasks page as it appears by default when a user first logs in.

CRACLE Business Process Workspace Tasks Process Trading Standard Dashboards Tore Advanced Advanced Advanced Tasks Process Name Creator Priority Assigned Flease select a task to see the details

Figure 1–2 Process Workspace Tasks Page

Copyright © 2005, 2012, Oracle and/or its affiliates. All rights reserved.

This page lists all the tasks and work items assigned to you, depending on your role.

At the far left is a list with **My Tasks** selected. Expand this list to select:

- A particular view showing the number of open tasks for each view. Selecting a
 particular view refreshes the task count to the latest number.
- A list of applications deployed to Process Workspace
- Any favorite links or applications you may have specified

To keep this list visible while you work on tasks, click **Pin**. To hide it, click **Unpin**.

Table 1–5 lists and describes the items in the list of views, applications, and favorites.

Table 1–5 Views, Applications, and Favorites List

Component	Description
Views	 Inbox—Shows all tasks that result from any filters you used. The default shows all tasks. You can set the priority for each task in your inbox by selecting the priority number from the inline list that appears in the Priority column.
	 Standard Views—Due Soon, High Priority, Past Day, Past Week, Past Month, Past Quarter, and New Tasks.
	The items displayed in the inbox depend on the role granted to the logged-in user.
	• Everyone (the user role) sees My Tasks, Initiated Tasks and Administrative Tasks.
	 Users who are also supervisors see the My Tasks, Initiated Tasks, My Staff Tasks and Administrative tabs
	 Users who are also administrators (the BPMProcessAdmin), but not supervisors, see the My Tasks, Initiated Tasks, and Administrative Tasks
	 Users who are supervisors and administrators see all the tabs— My Tasks, Initiated Tasks, My Staff Tasks, and Administrative Tasks
Applications	The processes the user is authorized to initiate. This component enables you to search for the authorized applications.
Favorites	Links to external websites or applications you may have specified. For more information, see Section 9.2.8, "How to Enable Customized Applications and Links in Process Workspace" on page 9-6.

In addition to the Views, Applications, and Favorites list, the My Tasks page includes other components listed and described in Table 1–6.

|--|

Component	Description
Display Filters	Specify search criteria from the Search, Assignee, or Status fields. The category filters available depend on which view is selected.
	• From the My Tasks page, the Assignee filters are:
	 Me
	 My Group
	 Me & My Group
	 Me (Previously) - tasks worked on in the past. Use Me (Previously) with the Status filter set to Completed to see tasks that are completed.
	 Me (Review Only)
	For the following pages, tasks are filtered using the Search function.
	 From the Initiated Tasks page, Search filters on Creator
	 From the My Staff Tasks page, Search filters on Reportees
	 From the Administrative Tasks page, Search filters on Admin
	The Status filters include:
	 Any
	 Assigned
	 Completed
	 Suspended
	 Withdrawn
	 Expired
	 Errored
	 Alerted
	 Information Requested
	You can perform an advanced search using several other criteria. For more information about searches, see Section 1.3.2.4, "Searching for Tasks in Process Workspace" on page 1-25.
Actions List	Select a group action (Claim) or a custom action (for example, Approve or Reject) that was defined for the human task. Claim appears for tasks assigned to a group or multiple users; one user must claim the task before it can be worked. Other possible actions for a task, such as system actions, are displayed on the Task Details page for a specific task. You can also create ToDo tasks and subtasks here. When you select an action from Actions , a comment window appears. You must enter a comment and click OK to update the task and refresh the task list table. Click Cancel to cancel the action.

Component	Description
Task Status	A bar chart shows the status of tasks in the current view.
Default Columns	Title —The title specified when the human task was created. Tasks associated with a purged or archived process instance do not appear.
	Process Name —If this task is associated with a running process instance, the name of that instance. Clicking this link opens a pop-up with process details.
	Creator —The user who created the task. For business user tasks, the creator field shows as <i>null</i> .
	Priority —The priority specified when the human task was created. The highest priority is 1; the lowest is 5.
	Assigned—The date that the task was assigned.
Task Details	Task details can be viewed in the lower half of the worklist by selecting the task in the Inbox. You can also view them in a pop-up browser window by double-clicking the task.
	Buttons on the Task Details page indicate available actions. After you complete a task:
	 The Task Details page for the completed task disappears.
	 The task list refreshes to show only the remaining tasks.
	 The details of the next open task appear.

Table 1–6 (Cont.) Components of the Tasks Page

1.3.2.2 Using the Process Workspace Task Details Page

Figure 1–3 shows an example of a Task Details page. This page is invoked by selecting a task from the inbox on the Process Workspace Tasks page. You can also view it in a pop-up browser window by double-clicking the task.

The Task Details page may appear differently depending on the tool used during design time to develop the task form it displays.

Figure 1–3 Example of a Task Details Page

sk 1			🔀 🧼 🧼 Actions 👻 Complete
🗹 😤 Details			
Assignees weblogic John Steinbeck James Cooper	Acquired By weblogic	Task Number 200000	
Creator weblogic	Category	Priority 1	
Created Aug 17, 2012 5:32 AM	Percentage 50.0	State Assigned	
Updated Aug 21, 2012 7:47 AM	Complete		
Expration Date	Due Date Oct 19, 2012 11:59 PM 40		
	Start Date Aug 17, 2012 11:59 PM		
⊻ History			
weblogic, John Steinbeck, James Coo 1 Assigned Aug 17, 2012 5:32 / weblogic	AM Ø		
2 Assigned Aug 17, 2012 5:33 J	oper- AM Ø		
🔏 weblogic, John Steinbeck, James Coo	oper -		
3 Acquired Aug 21, 2012 7:47 A weblogic	AM 🗑		
a			
÷			
a weblogic, John Ste			
+			
🐧 weblogic, John Ste			
*			
👌 weblogic, John Ste			
È			
⊻ Comments	9	⊻ Attachments	4%

The task details page has the following components:

- Basic information including the title of the task—for example, task 1—task number, creator, assignees, state, created date, the person who acquired the task, priority, expiration date, and other flex fields.
- Previous Task/Next Task icons—For moving from one task to the previous or next task in the task list.
- Action buttons—For custom actions defined for this human task during design time—for example, **Resolved** and **Unresolved** for a help desk request, or **Approve** and **Reject** for a loan request. After you complete your role in a task, for example, by making an approval, the task list refreshes accordingly and moves you to the next task in the task list.

For the administrator, the task initiator, or the manager, **Withdraw** may also appear.

The available task actions depend on the user role and the task. The task actions do not depend on the which view you have selected.

- Other Actions—System task actions, beyond the custom actions defined for the task—as described in Section 1.3.2.2.2, "Performing System Actions in Process Workspace".
- Content—The payload. The fields displayed are specific to how the human task was created during design time.
- History Details—These are described in Section 1.3.2.2.1, "Task History.".
- Comments and Attachments:
 - Comments entered by various users who have participated in the workflow. A
 newly added comment and the commenter's user name are appended to the
 existing comments. A trail of comments is maintained throughout the life
 cycle of the task. To add a comment, you must have permission to update the
 task.

For information about how to add a comment, see Section 2.4, "Adding Comments and Attachments to Tasks in Process Workspace"

 Attachments are documents or reference URLs associated with a task. These are typically associated with the workflow as defined in the human task during design time, or attached and modified by any of the participants using the task list. To add or delete an attachment, you must have permission to update the task. When adding file attachments, you can use an absolute path name or browse for a file.

1.3.2.2.1 Task History The task history maintains an audit trail of the actions performed by the participants in the workflow and a snapshot of the task payload and attachments at various points in the workflow. The short history for a task lists all versions created by the following tasks:

- Initiate task
- Reinitiate task
- Update outcome of task
- Completion of task
- Erroring of task
- Expiration of task
- Withdrawal of task

• Alerting of task to the error assignee

You can include the following actions in the short history list by modifying the shortHistoryActions element.

- Acquire
- Ad hoc route
- Auto release of task
- Delegate
- Escalate
- Information request on task
- Information submit for task
- Override routing slip
- Update outcome and route
- Push back
- Reassign
- Release
- Renew
- Resume
- Skip current assignment
- Suspend
- Update

The history provides a graphical view of a task flow, as shown in Figure 1–4.

Figure 1–4 History: Graphical View

History			
	🔏 weblogic, John 9	Steinbeck, James Cooper	
1	Assigned weblogic	Aug 17, 2012 5:32 AM	٥
	👌 weblogic, John 🤋	Steinbeck, James Cooper	
2	Assigned	Aug 17, 2012 5:33 AM	
2	🔱 weblogic, John 🤉	Steinbeck, James Cooper	
5	Released	Aug 18, 2012 8:08 AM	Ø
<u>a</u> ,	reblogic, John Ste		
8 v	reblogic, John Ste		
	ė.		


Description
Indicates an approver in an ad hoc routing scenario.
Indicates that the task has been approved
Indicates that the participant just receives a notification task and the business process does not wait for the participant's response. Participant cannot directly impact the outcome of a task, but in some cases can provide comments or add attachments.
Indicates that a set of people must work in parallel. This pattern is commonly used for voting.
Indicates that the participant belongs to a management chain.
Indicates the simple case in which a participant maps to a user, group, or role.
Indicates that the task is untouched
Indicates the end of the task flow.

Table 1–7 Icons for Task History

Note: The history of a parent task also displays the history of any subtasks it contains.

1.3.2.2. Performing System Actions in Process Workspace After you have selected a task from the task list, the Actions list of the Task Details page displays system actions available on all tasks based on the user's privileges. Table 1–8 lists system actions.

Action	Description
Claim	Use this action to claim a task that is assigned to a group or multiple users. Claim is the only action available in the Task Action list for group or multiuser assignments. After a task is claimed, all applicable actions are listed.
Reassign	Use this action to assign a task to someone else. If you are a manager, you can delegate a task to reportees. A user with BPMWorkflowReassign privileges can delegate a task to anyone.
Escalate	Use this action to escalate a task that is assigned to a group or multiple users. You can add an optional comment in the comments area. The task is reassigned to whomever is specified by the administrator. If no escalation path is specified, then the task is escalated to your manager (up one level in a hierarchy).
Release	Use this action to release a claimed task. If a task is assigned to a group or multiple users, it can be released if the user who claimed the task cannot complete the task. Any of the other assignees can claim and complete the task.

Table 1–8 System Task Actions

Description
Use these actions to suspend or restart a task. These options are available only to users who have been granted the BPMWorkflowSuspend role. Other users can access the task by selecting Previous in the task filter or by looking up tasks in the Suspended status. A suspension is indefinite. It does not expire until Resume is used to resume working on the task.
Use this action to cancel a task. If you are the creator of a task and do not want to continue with it, for example, you want to cancel a vacation request, you can withdraw it and add an optional comment in the Comments area. The business process determines what happens next. You can use the Withdraw action on the home page by using the Creator task filter.
Use this action to send a task down one level in the workflow to the previous assignee.
Use this task to extend a task's expiration date. You can add an optional comment in the comments area. The task expiration date is extended 1 week. A renewal appears in the task history. The renewal duration for a task can be controlled by an optional parameter. The default value is P7D (7 days).
Use these actions if another user requests that you supply more information or to request more information from the task creator or any of the previous assignees. If reapproval is not required, then the task is assigned to the next approver or the next step in the business process.

Table 1–8 (Cont.) System Task Actions

For more information about the Task Details page, see Section 1.3.2.2, "Using the Process Workspace Task Details Page".

1.3.2.3 Customizing the Task List in Process Workspace

You can customize your task list in several ways, including adding worklist views, selecting which columns to display, setting the sorting order, and displaying a subset of the tasks based on filter criteria. You can customize your inbox view as well as any other view including views you have created yourself.

To increase the number of tasks retrieved, resize the task list display area.

This section contain these topics:

- Section 1.3.2.3.1, "How to Filter Tasks"
- Section 1.3.2.3.2, "How to Create and Customize Worklist Views"

1.3.2.3.1 How to Filter Tasks Figure 1–5 shows the filter fields.

Figure 1–5 Filters: Assignee, Status, Search, and Advanced Search

C	ORACLE Business Process Workspace					Tasks Process Tracking Standard Dashboards More 👻										
r	My Ta:	sks(6) 🗸 🛛 🖾 🌔	Actions 🗸	নি <u>য়</u> Ass	signee M	le & My Group	~	State	Assigned	6	 Search 		•	Advanced		
	•	Title													Pro	6
:		simple task													Sin	n
1		LoanApplication_Lab1													Lo	a
8:	1	Submit request													Ap)p

Filters are used to display a subset of tasks and can be applied to your inbox view or other views. Filters are based on the following criteria:

Assignee

Select one of the following:

Me: Retrieves tasks directly assigned to the logged-in user

- My Group: Retrieves the following:
 - * Tasks that are assigned to groups to which the logged-in user belongs
 - Tasks that are assigned to an application role that the logged-in user is assigned
 - * Tasks that are assigned to multiple users, one of which is the logged-in user
- Me & My Group: Retrieves all tasks assigned to the user, whether through direct assignment, or by way of a group, application role, or list of users
- Me (Previously): Retrieves tasks that the logged-in user has updated
- Me (Review Only): Retrieves task for which the logged-in user is a reviewer

From the Initiated Tasks page, select Creator.

From the **My Staff Tasks** page, select **Reportees**.

From the **Administrative Tasks** page, select **Admin**.

- State: Select from the following: Any, Assigned, Completed, Suspended (can be resumed later), Withdrawn, Expired, Errored (while processing), Alerted, or Information Requested.
- **Search**: Enter a keyword to search task titles, comments, identification keys, and the flex string fields of tasks that qualify for the specified filter criterion.
- Advanced: Provides additional search filters.

Note: If a task is assigned separately to multiple reportees, when a manager looks at the My Staff Tasks list, the manager sees as many copies of that task as the number of reportees to whom that task is assigned.

Seeing your completed tasks:

When you mark a task as Completed, it is no longer displayed in your Task List. To see all completed tasks, select Me (Previously) from the **Assignee** filter and **Completed** from the **Status** filter. The task list is automatically updated to show your completed tasks.

To filter tasks based on assignee or state:

Select options from the **Assignee** and **State** lists. The task list is automatically updated based on the filter selections.

To filter tasks based on keyword search:

- 1. Enter a keyword to search task titles, comments, identification keys, and the flex string fields of tasks that qualify for the specified filter criterion.
- 2. Press Enter or click Refresh.

To filter tasks based on an advanced search:

Mapped attribute labels can be used in an advanced search if you select task types for which mapped attribute have been defined.

See Section 9.4.1, "How to Specify Mapped Attributes" for more information.

1. Click Advanced.

2. (Optional) Select Save As View, provide a view name, and use the Display page to provide other information, as shown in Figure 1–6 and Figure 1–7. If you do not select Save As View, the Display page is not available and the Definition page is as seen in Figure 1–8.

Note: Selecting **Save As View** creates a customized view. The saved view appears under **My Views**.

Advanced Sear	ch						1
Definition	Display						
Save Sear	th as View						
Name use	r view						
Assignee Me	& My Group All 💂	•					
		Match 💿	All 🍙 Any 🖻	ŀ			
Task Type	equals	-		_	Q	×	
	L						
Share View	Definition only	o Data					
Us	ers						
Gro	lps		4				

Figure 1–6 Worklist Advanced Search: Definition Page

	would aways be present	and would be first in the task table	h.
lect View Co Available	olumns		
Start Date Task Defin Owner Rol Custom Da Updated D Custom Ba Custom St Created Composite Custom St From User	tion Name e te 1 ate te 2 ring 2 Version ring 1	Title Creator Priority Assigned	ム マ マ
Sort by	Created	▼ ▼△	
Then by	None	▼ ▼△	
Then by	None	▼ ▼△	
Then by	None	× 🛆	

Figure 1–7 Worklist Advanced Search: Display Page



Advanced Searc	ch					×
Definition	Display					
Save Search	h as View					
		Match 💿 All	🎯 Any 🕂			
Task Type	equals	•		9	×	
				Reset	Search	Cancel

Table 1–9 describes the advanced search view columns available in the **Display** page.

Note: The titles: **User Conditions**, **Advanced Condition**, **Time Conditions**, **Basic Conditions** and **Custom Conditions** correspond to the menu categories for adding conditions (see Figure 1–11).

The saved view appears in the **Views** panel under **My Views**, as shown in Figure 1–9.

Figure 1–9 Saving a View



3. Select an assignee, as shown in Figure 1–10.

Figure 1–10 Worklist Advanced Search

Advanced Search	×
Definition Display	
Save Search as View	
Name My Saved Search	
Assignee Me & My Group	
Match 🔷 All 🖲 Any 🖶	

- 4. Select Any or All for matching multiple filters.
- **5.** Add conditions (filters), as shown in Figure 1–11.

Definition Display Save Search as View Match All • Any • User Acquired By Advanced Approvers Basic Creator Time From User Custom Original Assignee User Owner Group Owner Role Owner User Updated By	Advanced Search					
■ Save Search as View Match All All All Ar User Acquired By Advanced Advanced Advanced Acquired By Creator Time Custom Original Assignee User Owner Group Owner Clee Updated By	Definition Display					
Match All O Ary User Acquired By Advanced Approvers Basic Creator Time From User Original Assignee User Owner Group Owner Role Owner User Updated By Dydated By	Save Search as View					
User Acquired By Advanced Approvers Basic Creator Time From User Custom Original Assignee User Owner Group Owner Role Owner User Updated By		Match	🔿 All 💿 Any 🤇	÷		
Advanced + Approvers Basic + Creator Time + From User Custom + Original Assignee User Owner Group Owner Role Owner User Updated By				User	•	Acquired By
Basic · Creator Time · From User Custom · Original Assignee User Owner Group Owner Role Owner User Updated By				Advanced	•	Approvers
Time + From User Custom + Original Assignee User Owner Group Owner Role Owner User Updated By				Basic	•	Creator
Custom Original Assignee User Owner Group Owner Role Owner User Updated By				Time	•	From User
Owner Group Owner Role Owner User Updated By				Custom	•	Original Assignee User
Owner Role Owner User Updated By						Owner Group
Owner User Updated By						Owner Role
Updated By						Owner User
						Updated By
Dust Court Court					_	

Figure 1–11 Adding Filters for an Advanced Search on Tasks

Table 1–9 describes the available conditions.

Condition	Description
User Conditions	
Acquired By	The name of the user who claimed the task in the case when the task is assigned to a group, application role, or to multiple users, and then claimed by the user.
Approvers	The approvers of the task.
Assignees	The assignees of the task.
Creator	The name of the creator of the task.
From User	The from user for the task.
Original Assignee User	The name of the user who delegated the task in the case when the user delegates a task to another user.
Owner Group	The group (if any) that owns the task instance. Task owners can be application roles, users, or groups. If the owner of the task is a group, this field is set.
Owner Role	The application role (if any) that owns the task instance. Task owners can be application roles, users, or groups. If the owner of the task is an application role, this field is set.
Owner User	The user (if any) that owns the task instance. Task owners can be application roles, users, or groups. If the owner of the task is a user, this field is set.
Updated By	The user who last updated the task.
Advanced Conditions	

 Table 1–9
 Advanced Search: Conditions

Condition	Description
Application Context	The application to which any application roles associated with the tasks (such as assignees, owners, and so on) belong.
Component	The name of the task component that defines the task instance.
Composite	The name of the composite that contains the task component that defines the task instance.
Composite Distinguished Name	The unique name for the particular deployment of the composite that contains the task component that defines the task instance.
Composite Version	The version of the composite that contains the task component that defines the task instance.
Partition	The domain to which the composite that contains the task component that defines the task instance belongs.
Task Display URL	The URL to display the details for the task.
Basic Conditions	
Category	The category of the task.
Identifier	The (optional) custom unique identifier for the task. This is an additional unique identifier to the standard task number.
Number	An integer that uniquely identifies the task instance.
Outcome	The outcome of the task, for example approved or rejected. This is only set on completed task instances.
Percentage Complete	The percentage of the task completed (used with ToDo tasks).
Priority	An integer that defines the priority of the task. A lower number indicates a higher priority; typically numbers 1 to 5 are used.
State	The state of the task instance.
Task Definition Name	The name of the task component that defines the task instance.
Task Type	The type of the task.
Title	The title of the task.
Time Conditions	
Assigned	The date that this task was assigned.
Created	The date that the task instance was created.
Due Date	The due date of the task (used with ToDo tasks).
End Date	The end date of the task (used with ToDo tasks).
Expires	The date on which the task instance expires.
Start Date	The start date of the task (used with ToDo tasks).
Updated Date	The date that the task instance was last updated.
Custom Conditions	
Custom Date 1	Custom flex field 1 with Date datatype
Custom Date 2	Custom flex field 2 with Date datatype
Custom String 1	Custom flex field 1 with String datatype
Custom String 2	Custom flex field 2 with String datatype
Custom Number 1	Custom flex field 1 with Number datatype

Table 1–9 (Cont.) Advanced Search: Conditions

Condition	Description
Custom Number 2	Custom flex field 2 with Number datatype

6. Add parameter values as exemplified in Figure 1–12.

Figure 1–12 Advanced Search

Advanced Search	×
Definition Display	
✓ Save Search as View	
Name user view	1
Assignee Me & My Group All	
Match 💿 All 💿 Any 🐥	
Task Type equals	
Start Date on 💽	
🔇 August 💌 2012 🖨 🕨	
SUN MON TUE WED THU FRI SAT	
29 30 31 1 2 3 4 5 6 7 8 9 10 11	
Share View Definition 12 13 14 15 16 17 18 19 20 21 22 23 24 25	
Users 26 27 28 29 30 31 1	
Groups	
Reset Save as View Search Cance	el

7. Browse for a task type or leave the field blank for all types. Mapped attribute labels can be selected in the query and display columns dialog boxes if the selected task types have mapped attribute mappings defined. Use the search option to search by task type, process name or category as shown in Figure 1–13.

Figure 1–13 Task Type Browser

	Search		,
Task Type	Process Name	Description	Outcomes
To-do task	None	Ad hoc To-do task	Complete
Simple Approval	default/SimpleApproval/1.0	Built-in simple approval task	Approve,Reject

- **8.** Specify whether to share either this view's definition or its data, and the users or groups to share it with.
- 9. Click Search.

The task list appears with the tasks filtered according to your criteria.

1.3.2.3.2 How to Create and Customize Worklist Views The Views menu, shown in Figure 1–14, displays your inbox, standard views, and any custom views you have created. Use it to create, share, and customize views.

Figure 1–14 Views Menu

ORACLE Business Process Works



To create a worklist view:

1. In the Views section, click Add View. The Create User View dialog box appears as shown in Figure 1–15

Create User View	X
Definition Display	
Create View Use Public View	
Name user view	Add to Standard Views
Assignee Me & My Group All	
Match 💿 All 🍥 Any 🕂	
Task Type equals	۹ 🗙
Chara Maria - O Definition only - O Data	
Groups	
	OK Cancel

Figure 1–15 Creating a Worklist View

- 2. Use the **Definition** page of the Create User View dialog box to do the following:
 - Create View or Use Public View—Create your own view or browse for a public view to copy.
 - Name—Specify a name for your view.
 - Add to Standard Views—This option applies to administrators only. Administrators select this option to create the view as a standard view, which then appears in the Standard Views list for *all* worklist users.
 - Assignee—Select Me, My Group, Me & My Group, Me & My Group All, Me (Previously), Reviewer, Creator, Reportees, Admin, Owner
 - Add Condition—Select a filter from the list and click Add. For example, if you select startDate, and click Add, then a calendar and a list including on, equals, not equals, greater than, less than, and so on appears.
 - Match—Select All or Any to match the conditions you added.
 - Task Type—Browse for a task type or leave the field blank for all types. Mapped attribute labels can be selected in the query and display columns dialog boxes if the selected task types have mapped attribute mappings defined.
 - Share View—You can grant access to another user to either the definition of this view, in which case the view conditions are applied to the grantee's data, or to the data itself, in which case the grantee can see the grantor's worklist view, including the data. Sharing a view with another user is similar to delegating all tasks that correspond to that view to the other user; that is, the other user can act on your behalf. Shared views are displayed under My Views.
 - Assignees—Specify the users (grantees) who can share your view.

3. Use the **Display** page of the Create User View dialog box, shown in Figure 1–16, to customize the fields that appear in the view.

Create User ¥	/iew				×
Definition Column 'Title' (Display would always be present a	and would	be first in the task table		
Select View Co	olumns				
* Available			Selected		
Start Date Assignees Task Defini Owner Rol Custom Da Updated D Custom Sta Created Composite	ition Name e ate 1 ate 2 ring 2 Version ving 1	> >> < < <	Title Priority Creator Assigned	4 A A	
Sort by	Created				
Then by	None		V VA		
Then by	None		▼ ▼ △		
Then by	None		▼ △ ▲		
Sort Order	Ascending 💌				
					OK Cancel

Figure 1–16 Displaying Fields in a Worklist View

- Select View Columns—Specify which columns you want to display in your task list. They can be standard task attributes or mapped attributes that were mapped for the specific task type. The default columns are the same as the columns in your inbox.
- **Sort by** —Select a column or multiple columns on which to sort. Use the arrow buttons to specify the preferred sequence.
- Sort Order—Select ascending or descending order.

The sorting options selected are saved as a user preference.

4. Click OK.

The saved view appears in the Views panel under My Views.

To customize an Inbox view:

Note: If you select to display the **Task Display URL** column, you must have an associated taskflow for the deployed task. The taskflow registration is performed during deployment. At that time, the **Task Display URL** column value is set. If the taskflow is not associated during deployment, the **Task Display URL** column does not display generic URL details for a task.

1. In the Views, Applications, and Favorites pane, select Views, then select the view you want to customize.

2. Select Edit User Preferences. The Edit User Preferences dialog box appears as shown in Figure 1–17.

Edit User Prefe	rences		\boxtimes
* Stor Columns	Available Columns	Selected Columns	
Ŭ	Start Date	Title	
	Task Definition Name	Creator	
	Owner Role	Priority	
	Custom Date 1	Assigned	$\overline{\Delta}$
	Updated Date		\bigtriangleup
	Custom Date 2	8	\bigtriangledown
	Custom String 2	1	$\overline{\mathbf{A}}$
	Createu Composite Version		
	Custom String 1		
	Erom Licer		
Title will always a	appear as first column irrespective of its	position chosen above	
	Sort by Number	▼ ▼△	
	Then by Created	▼ ▼ △	
	Title	\checkmark \checkmark \triangle	
	Then by Priority	🖌 🗠 🖌	
	Sort Order Ascending		
	bort of doi in bearlang		
Number of task	s per fetch 🛛 20 🛛 💌		
Hide Task D	etails Panel 📃		
Display ta	sk details in 💿 External Window 🔵 San	ne Window	
Use language	settings of 🔘 Browser 💿 Identity Pro	vider	
C	efault View Past Week 🔽		
			OK Cancel

Figure 1–17 Customizing Fields in a Worklist View

3. Use the Edit User Preferences dialog box to customize the view.

When you select and move items from the **Available Columns** list to the **Selected Columns** list (or vice-versa), the items remain selected. Therefore, if you select items to move back, the previously selected items are also moved. Ensure that you deselect items after moving them between the lists if you intend to move additional columns.

To specify the number of tasks to display in the view, set the number in the **Number of tasks per fetch** list.

To hide the details panel so that it appears only when you select a task, select **Hide Task Details Panel**.

To specify how the Task Details Panel appears, in the **Display task details in** field, select either **External Window** or **Same Window**.

To specify which language settings to use, in the Use Language Settings of field, select either **Browser** or **Identity Provider**. For more information about specifying language, see Section 9.2.3, "How to Specify the Language Locale Information in Process Workspace".

When you have finished making your settings, click **OK**.

1.3.2.4 Searching for Tasks in Process Workspace

For instructions about searching for tasks, see:

- "To filter tasks based on keyword search:" on page 1-15
- "To filter tasks based on an advanced search:" on page 1-15

1.3.3 About the Process Workspace Activity Guides Page

Activity guides organize your tasks into milestones. Each milestone is a specific set of tasks for you to complete, either on your own or in collaboration with others. For each milestone, the activity guide tells you which tasks to complete and in which order. You complete a milestone by completing all the tasks it contains for you. Depending on the nature of the task, you can save an unfinished task and resume it later.

Figure 1–18 shows an example of an activity guide.



Activity Guide List						
🝓 Assignee My 💽 Status In Prog	ress 🔽					
Title	Status	Initiated	Instance #	Process Name	Composite Name	
OnBoarding guided process-6	In Progress	Jun 26, 2011 2:59 PM	6	OnBoarding guided process	OnBoardingProcess!1.0	~
OnBoarding guided process-5 View the status and	In Progress	Jun 26, 2011 2:59 PM	5	OnBoarding guided process	OnBoardingProcess!1.0	
OnBoarding guided process-4 the overall business	In Progress	Jun 26, 2011 2:59 PM	4	OnBoarding guided process	OnBoardingProcess!1.0	
OnBoarding guided process-3 process in which yo	u In Progress	Jun 26, 2011 2:59 PM	3	OnBoarding guided process	OnBoardingProcess!1.0	~
< participate.					1	>
Activity Guide 🛛 🖓 🚽						÷,
Activity Guide Completion	Validate personal info				Actions - Ok No	
095 10095	✓ Contents					
Completed Tacks (0/1)	Employee Id jcooper					
Completed lasts (0/1)	First Name John					
Task Filter All	Last Name Cooper					
18.55	User Info - Address					
view +	Street 1156 High Stre	et				
OnBoarding guided process	City Santa Cruz					
	State California					
V O * Personal Info	Zip Code 95064					
🗄 * Validate personal info	Country United States					
	Comments		9	✓ Attachments	÷×	
Logistics	No data to display			No data to display		
Expand a miles	stone to see and act on sks it includes.					
						~
-						_
ovright © 2005, 2011, Oracle and/or its affiliates. All ric	hts reserved.					

Using an activity guide, you can:

- Expand a milestone to see and act on the tasks it includes.
- View the status and percent completion of the overall business process in which you participate.

For more information about navigating the Activity Guides page, see Section 4.2, "Navigating Activity Guides".

1.3.4 About the Process Workspace Process Tracking Page

Use the Process Tracking page to interact with business processes based on the roles assigned to you.

Using the Process Tracking tab, you can do the following:

- View the list of process instances pending in your inbox.
- Execute tasks. That is, carry out a human task on a particular instance.
- Search for instances by different criteria.
- Save the searches as a view.
- View the details of an instance, including the audit trail.
- Create a new instance, if you have permission to do so.

- Attach a file to an instance.
- Add notes to an instance.

1.3.4.1 Navigating the Process Workspace Process Tracking Page

Title						
Theo			Number	Initiator	Process Name	In
New title modify I	DO		33		[Project3 v1.0] ReviewPr	rocess O
Instance #50 of	OnboardingProcess		50		[M4HCMDemo v1.0] Onb	oardinç O
Instance #51 of	OnboardingProcess		51		[M4HCMDemo v1.0] Onb	oardinç O
Instance #52 of	ReviewProcess		52		[Project3 v1.0] ReviewPr	rocess O
<						
w title modify	y DO					Action:
Dotaile						1
Decails					\searrow	
Priority	/ normal	Process [Project3 -	v1.0]	Creat	ion time 10/10/2011	
Status	Running	Number 33	ocess			
		Hambor 00				
Open Activitie						
	•					
Activity	• Title	Assigned to		Priority	State	Act
Activity VeryExpensiveUse	Title rReVery Expensive Re	Assigned to vieProject3.ProcessO	Nner	Priority	State	Acl
Activity /eryExpensiveUse ;impleUserReview	Title rReVery Expensive Re SimpleUserReview	Assigned to vieProject3.ProcessOv Project3.ProcessOv	wner	Priority	State Running Running	Acl Ru Ru
Activity /eryExpensiveUse 5impleUserReview	Title rReVery Expensive Re SimpleUserReview	Assigned to vie Project3.ProcessOu Project3.ProcessOu	wner wner	Priority	State Running Running	Acl Ru Ru

Figure 1–19 Process Workspace Process Tracking Page

The Process Tracking page has these sections:

The Applications panel, which shows the list of applications available to you. Here
you can perform basic Process Workspace actions, such as creating a new instance
performing a given activity, or sorting applications.

Hovering over the name of an application in the Applications panel displays a tool tip with the application's full name and revision number.

The applications available in Process Workspace depend on the process design and the roles assigned to the user currently logged in. The list of available applications also depends on the BPM processes that are deployed and running. If no processes are running, or if you are not a participant able to execute an application in any of the processes currently running, then the applications panel is empty.

In the applications panel, you can sort either of the following:

- Only the default versions of applications
- All active versions of applications

You can sort either by name or by process version. You can filter the same applications by entering in the **Search** field either full or partial names or full or partial versions.

- The Views panel, on the lower left half, displays the views available to you in an expandable tree
- The Pending Components panel shows all suspended processes. If a new version of a process is deployed, and the previously deployed process has instances not yet migrated to the new version, then the previously deployed process is suspended. To resume it, either migrate or suspend any non-migrated instances. For information about resuming suspended processes, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*, "Resuming a Suspended Process".
- The Work Items panel, on the upper right half, shows the task items corresponding to the view you have selected in the Views panel. In this panel, you can choose the columns that are presented in a given view. A particular column arrangement is known as a presentation. Clicking **Add/Remove columns** displays the Presentation dialog box in which you can define columns.

The following columns are available:

- Actions
- Activity
- Activity Due Date
- Copy
- Creation Time
- Deadline
- Description
- Has Attachments
- Has Notes
- Initiated
- Initiator
- Initiator Name
- Instance #
- Parent Copy
- Participant
- poID
- Priority
- Process Name
- Process Due Date
- Received
- Status
- Task Participant (that is, the participant that is currently executing an instance.)
- Title
- The Work Item Detail panel, on the lower right half, shows detailed information about a work item that you have selected in the Work Items panel. It has these sections:

- Details: Shows the priority, process name, when the instance is due to complete, status, creation time, and the assignee if any.
- Open Activities: Shows the status of any gateways with threads that are still running.
- Audit Trail: Shows audit details in either a list view, a tree view, or graphical view. You can select the type of activity or sub-process you want to audit: human activities, service activities business rule activities, subprocesses, events, gateways, script activities, other activities, or all of the above.

If the process includes an input or output payload, then you can display the audit details for that message in a pop-up. Clicking on a particular activity displays the task history.

For more information about viewing audit details, see Section 3.3.1, "Viewing the Audit Trail in Process Workspace."

- Comments: Enables you to add Comments to the process instance.
- Attachments: Enables you to add attachments to the process instance.

For more information about adding comments and attachments to a process instance, see Section 3.3.2, "Associating Comments and Attachments with a Process Instance in Process Workspace."

1.3.5 About the Process Workspace Standard Dashboards Page

Dashboards provide analytical information to monitor process execution as well as user workloads. They provide summary information and enable you to drill down to instance-specific information.

Figure 1–20 shows the Standard Dashboards page.

Figure 1–20 Process Workspace Standard Dashboards Page



For additional information about dashboards, see Chapter 6, "Using Dashboards in Process Workspace".

1.4 Setting Accessibility Preferences in Process Workspace

To set accessibility preferences:

- 1. In the upper right corner of Process Workspace, click Preferences.
- **2.** In the Configuration Areas panel, select **Accessibility**. The Select Accessibility page appears in the right pane as shown in Figure 1–21.

Figure 1–21 Selecting Accessibility

ORACLE: Business Pro	ocess Workspace	62	Logged in as weblogic	Home	Administration	Preferences	Help	Logout	2
Preferences	Select Accessibility								Ī
Vreterences User Preferences Rules Notification Preferences Accessibility	Select Accessibility							I	ţ.
Copyright © 2005, 2010, Oracle and/or its :	J								1

3. Select your accessibility preferences, then click OK.

1.5 Providing a Digital Certificate in Process Workspace

A digital certificate establishes the participant's credentials. It is issued by a certification authority (CA). It contains your name, a serial number, expiration dates, a copy of the certificate holder's public key (used for encrypting messages and digital signatures), and the digital signature of the certificate-issuing authority so that a recipient can verify that the certificate is real.

To provide a digital certificate:

- 1. In the upper right corner of Process Workspace, click **Preferences**.
- 2. In the navigation bar on the left, click Certificates.
- 3. Upload the certificate to use to sign your decision, as shown in Figure 1–22.

When signing a task outcome using your certificate, you must upload the entire chain of certificates through Process Workspace as a . P7B (PKCS7 format) file, not just the one certificate issued to you by the certificate issuer. The entire chain can be exported through Internet Explorer. Mozilla Firefox does not let you export the chain as a . P7B file. Therefore, you can perform the following steps:

- **a.** Export the chain from Mozilla Firefox as a . P12 file (PKCS12 format that also contains your private key).
- **b.** Import the . P12 file in Internet Explorer.
- c. Export it again from Internet Explorer as a . P7B file.
- d. Upload it through Process Workspace.

Preferences	Upload Certificate			
Jser Preferences Certificates Rules Notification Preferences Accessibility	Browse for Certificate Certificate file type PKC57 PKC512 Select Certificate File Certificate Password Type or Paste Certificate Contents	Browse	L2∕	

Figure 1–22 Uploading a Certificate

Note the following important points when providing your certificate to the system. Otherwise, you cannot use your certificate to sign your decisions on tasks.

- The PKCS7 file format is a binary certificate format. Select this option if you
 have a standalone certificate file stored on your disk.
- The PKCS12 file format is a keystore format. Select this option if you have your certificate stored inside a keystore.
- If you want to copy and paste the contents of the certificate, select Type or Paste Certificate Contents and paste the BASE64-encoded text into the field. Do not paste a certificate in any other format into this field. Likewise, if you choose to upload a certificate, do not upload a BASE64-encoded certificate. Only PKCS12- and PKCS7- formatted files are supported for uploads.
- **4.** Return to the task list by clicking the **Home** link in the upper right corner of Process Workspace.

5. Click a task to approve or reject.

The task details are displayed.

- Click either Approve or Reject.
 Details about the digital signature are displayed.
- 7. For a task that has a signature policy, click **Sign**.

The Text Signing Report dialog box appears.

- 8. Select the certificate from the list to use to sign your decision.
- 9. Enter the master password of the Web browser that you are using.
- **10.** Click **OK**.

The Web browser signs the string displayed in the upper half of the Text Signing Request with the certificate you selected and invokes the action (approval or rejection) that you selected. The task status is appropriately updated in the human workflow service.

If you are a developer and want more information about how certificates are uploaded and used, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

1.6 Accessing Administration Capabilities

The upper right toolbar contains an **Administration** link that takes you to the Administration Areas.

Table 1–10 lists and describes the categories in the Administration Areas and tells where to find more information about each one.

Category	Description	Information
Organization	Model an organization by defining organization units, business calendars, business holidays, roles, and other user properties.	Section 10, "Managing Your Organization in Process Workspace"
	Note: If you are logged into Process Workspace, and changes are made to your organizational membership, you will not see these changes reflected in the interface until you log out and log back in again. This is because, for better performance, the organization unit information is cached when you log into Process Workspace.	
Flex Fields (Mapped Attributes)	Create mappings needed for populating mapped attributes.	Section 9.4, "Managing Mapped Attributes (Flex Fields) in Process Workspace"
Task Administration	Administer approval groups, task configuration, evidence search, and exceptions	Section 9.5, "Using Task Configuration in Process Workspace"
Application Preferences	Customize the appearance of the worklist.	Section 9.2, "Setting the Display of the Work Items Panel in Process Workspace"
Accessing Process Workspace in local languages	Specify your preferred language and time zone.	Section 9.2.3, "How to Specify the Language Locale Information in Process Workspace"

Table 1–10 Categories in the Administration Areas

1.7 Setting Your Preferences in Process Workspace

The upper right toolbar contains a **Preferences** link.

Table 1–11 lists and describes the preferences you can set and tells you where to find information for setting each preference.

Description	Information
Provide a digital certificate from a certification authority to verify your credentials.	Section 1.5, "Providing a Digital Certificate in Process Workspace".
Set rules for tasks assigned to a user or group.	Section 2.6.1, "How to Create User Rules in Process Workspace"
	Section 2.6.2, "How to Create Group Rules in Process Workspace"
Configure the notification settings to control how, when, and where you receive messages in cases when you have access to multiple communication channels (delivery types).	Section 2.6.4, "Specifying Notification Settings".
Select accessibility preferences such as using a screen reader, using high contrast colors, and large fonts.	Section 1.4, "Setting Accessibility Preferences in Process Workspace"
	Provide a digital certificate from a certification authority to verify your credentials. Set rules for tasks assigned to a user or group. Configure the notification settings to control how, when, and where you receive messages in cases when you have access to multiple communication channels (delivery types). Select accessibility preferences such as using a screen reader, using high contrast colors, and large fonts.

Table 1–11 Preferences

In addition to the preferences mentioned in this section, administrators can also set various preferences for how the Work Items panel is displayed in Process Workspace. For information about setting these application preferences, see Section 9.2, "Setting the Display of the Work Items Panel in Process Workspace".

Working on Tasks in Process Workspace

This chapter describes how to set user preferences and act on tasks with Oracle Business Process Management Workspace, including claiming and releasing tasks; reassigning, delegating, and routing tasks; associating comments and attachments to tasks; and setting rules and vacation periods.

This chapter contains these topics:

- Section 2.1, "Claiming and Releasing Tasks in Process Workspace"
- Section 2.2, "Modifying History Approval Options in the Task Details Page"
- Section 2.3, "Reassigning and Delegating Tasks in Process Workspace"
- Section 2.4, "Adding Comments and Attachments to Tasks in Process Workspace"
- Section 2.5, "Requesting Information in Process Workspace"
- Section 2.6, "Setting Rules, Vacation Periods, and Notification Preferences in Process Workspace"
- Section 2.7, "Creating ToDo Tasks in Process Workspace"
- Section 2.8, "Creating Subtasks in Tasks in Process Workspace"
- Section 2.9, "Acting on Tasks that Require Passwords in Process Workspace"
- Section 2.10, "Acting on a Task that Requires a Digital Signature in Process Workspace"
- Section 2.11, "Routing Tasks in Process Workspace"
- Section 2.12, "Evidence Store Service and Digital Signatures"
- Section 2.13, "Creating Worklist Reports"

For information about troubleshooting human workflow issues, see Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite, "Human Workflow Troubleshooting".

2.1 Claiming and Releasing Tasks in Process Workspace

If a task is assigned to a group or multiple users, then the task must be claimed first. Claim is the only action available in the Task Action list for group or multiuser assignments. After a task is claimed, all applicable actions are listed.

Note: If the task has the auto claim option set, the user does not need to claim the task. The custom action is available for the user.

If a task is assigned to a group or multiple users, it can be released by the user who claimed it. Any of the other assignees can then claim and complete the task.

To claim and release tasks:

To claim a task, open it in the Task Details page, then click the Claim button.

Similarly, to release a task, open it in the Task Details page, then click the **Release** button.

Figure 2–1 shows part of a Task Details page with the **Claim** button in the upper right corner.

Figure 2–1 Claiming a Task

Solution Details Assignees weblogic John Steinbeck James Cooper Acquired By Creator weblogic John Steinbeck James Cooper Creator Category Pricentage 50.0 Complete State Due Date State Start Date State	
Assignees weblogic John Steinbeck James Cooper Acquired By Task Number 200020 Creator weblogic Category Priority Image: Cooper Category Creator weblogic John Steinbeck, James Cooper - State Assigned State Assigned Weblogic, John Steinbeck, James Cooper - Image: Cooper - Image: Cooper - Image: Cooper -	
✓ History & weblogic, John Steinbeck, James Cooper	
🛔 weblogic, John Steinbeck, James Cooper -	
1 Asigned Aug 21, 2012 8:33 AM Weblogic Image: State of the stat	+%

2.2 Modifying History Approval Options in the Task Details Page

This section includes the following topics:

- Section 2.2.1, "Combining Repeated Approvals of a Task in the History"
- Section 2.2.2, "Excluding System Approvals from a Task"
- Section 2.2.3, "Including New Participants in an Existing Task"

2.2.1 Combining Repeated Approvals of a Task in the History

At times you may want to combine tasks that have identical approval sequences into one combined approval—for example, to approve a purchase order that has many identical line items each requiring its own approval. Rather than approving each line item one-by-one, you can combine them into a single approval and then approve that combined task.

After you combine tasks into one repeated approval, you cannot alter the sequence of those tasks; for example, you cannot add, edit, or delete a participant in that task.

To combine repeated approvals of a task:

- **1.** In the Process Workspace task list, select the task. The Task Details window appears.
- 2. In the Task Details window, expand the History section.
- **3.** Click **Options** as shown in Figure 2–2.

Figure 2–2 History Pane - Options

∠ History	
+ /	X Options V Apply Reset V
1	▽ [m] Project Manager Approval Stage □ Show all \$33 John Steinbeck - Project Manager Action □ Exclude system approvals
1.1	Assigned Aug 31, 2012 3:
2	
2.1	V 🔒 Branch 2
2.1.1	
2, 1, 1, 1	⊽ 👬 Branch 2.1.1
2.1.1.1.1	▽ 🤮 Branch 2.1.1.1 (Aggregated)
2.1.1.1	
2.1.1.1	
2.1.1.1	🖀 Mark Twain - Expense Report Approvers
2.2	🗸 🖁 Branch 2
2.2.1	
2.2.1.1	⊽ 👬 Branch 2.2.1
2.2.1.1.1	🗸 👌 Branch 2.2.1.1 (Aggregated)
2.2.1.1	ិដ្ឋិ Irving Stone - Expense Report Submitter Participants

4. Verify that Combine Repeated Approvals is selected. This is the default setting.

To separate the tasks, clear the **Combine Repeated Approvals** check box.

5. Click Apply.

2.2.2 Excluding System Approvals from a Task

To remove all system generated updates from the History of a task, you must select the option to exclude system approvals. To select **Exclude system approvals**:

- **1.** In the Process Workspace task list, select the task. The Task Details window appears.
- 2. In the Task Details window, expand the History section.
- **3.** Click **Options** as shown in Figure 2–2.
- 4. Select Exclude system approvals.
- 5. Click Apply.

2.2.3 Including New Participants in an Existing Task

To include new participants in a task, you must select the **Including Future Approvals** option in the History pane. This option adds participants to the task history and displays the actions they perform on the task.

To include new participants:

1. In the Process Workspace task list, select the task for which you want to include new participants. The Task Details window appears.

- 2. In the Task Details window, expand the History section.
- **3.** Click **Options** as shown in Figure 2–2.
- Verify that Include Future approvals is selected. This is the default setting. To disable this option, clear the Include Future approvals check box.
- 5. Click Apply.

2.3 Reassigning and Delegating Tasks in Process Workspace

To reassign or delegate a task:

- **1.** In the Task Details window, from the **Actions** list, select **Reassign**. The Reassign Task dialog box appears.
- 2. Select either Reassign or Delegate.

Delegate differs from **Reassign** in that the privileges of the delegatee are based on the delegator's privileges. This function can be used by managers' assistants, for example.

3. Enter a user or group name or browse for a user or group name, as shown in Figure 2–3. You can use an asterisk (*) as a wildcard.



Reas Rea This C	sign will t) Rea) Dele	Task tasks to one or more users ransfer ownership of the ta ssign (transfer task to anot sigate (allow specified user t j#	:, isk and remove it ther user or group to act on my beha	from your workli)) IF)	st. Search	Reset			
	ID		First Name	Last Name	Email	Title	Manager	Organization	
	å	jausten	Jane	Austen	jausten@emailExample.com	Loan Consultant	fkafka	Demo User	
V	å	jcooper	James	Cooper	jcooper@emailExample.com	Loan Agent 1	jstein	Demo User	
	å	jlondon	Jack	London	jlondon@emailExample.com	Loan Agent 1	sfitzger	Demo User	
V	i i	jstein	John	Steinbeck	jstein@emailExample.com	Manager2	wfaulk	Demo User	
	å	jverne	Jules	Verne	JVerne@emailExample.com	CIO	cdickens	Demo User	
Selec	ted	🌡 jverne 🗱 십 ji	cooper 🗱 💧	jstein 🗱					OK <u>C</u> ancel

A supervisor can reassign tasks to any of his or her subordinates at any time.

4. Select the users or groups to which you are reassigning and click OK.

You can reassign to multiple users or groups. One of the assignees must claim the task.

Note: When task details have been upgraded from an earlier release, you may see a "Request Failed" error when executing the **Reassign** action. Actually, the reassign completes, and when you click OK again, a popup says the task is already assigned.

To eliminate the error message, upgrade your taskflow applications by opening them in Oracle JDeveloper, then redeploy the task form.

2.4 Adding Comments and Attachments to Tasks in Process Workspace

You can add comments and attachments to a task by using the Task Details page.

For details about the Task Details page, see Section 1.3.2.2, "Using the Process Workspace Task Details Page"

To add a comment to a task:

- 1. In the My Tasks page, from the worklist pane, select the task to which you want to add a comment. The details for that task appear.
- **2.** In the comments section of the Task Details page, specify whom you want to see the comment by clicking either:
 - Only task participants. This option shares the comment with previous and future assignees of this particular task.
 - All process participants. This option shares the comment with all participants in the process, regardless of the tasks assigned to them.
- 3. In the Create Comment dialog box, enter your comment.

Note:

- If you are the initiator of the task, then your comment is shared with all process participants and not only with task assignees. The option to share with only task participants is not available to you.
- Comments added to a task also appear in any subtasks of that task.
- 4. Click OK.

To add an attachment to a task:

- 1. In the Tasks page, from the worklist pane, select the task to which you want to add an attachment. The details for that task appear.
- **2.** In the Attachments section of the Task Details page, click **Add**. The Add Attachment dialog box appears.
- **3.** In the Add Attachment dialog box, select to share the attachment with either of the following:
 - Only task participants. This option shares the attachment with previous and future assignees of this particular task.
 - All process participants. This option shares the attachment with all participants in the process, regardless of the tasks assigned to them.
- **4.** Select the attachment type.

If the task is *not* using Oracle WebCenter Content, then you see the following options:

- URL. Selecting this option opens fields in which you specify the name and URL of the file.
- Desktop File. Selecting this option opens a search field from which to browse for the file on your computer.

If the task is using Oracle WebCenter Content, then you see the following options:

- URL. Selecting this option opens fields in which you specify the name and URL of the file.
- Upload File to UCM. Selecting this option uploads the desktop file to Oracle WebCenter Content. When you select it, another dialog box prompts you for some or all of the following depending on how the task is modeled:
 - Description (optional)
 - Folder name: The folder in WebCenter Content server to which the attachment is uploaded. You must type the full folder path in WebCenter Content server. To display the Contribution Folders level in WebCenter Content, enter a forward slash (/), as shown in Figure 2–4. To display a list of destination folder within the Contribution Folders level, enter a forward slash (/), as shown in Figure 2–5.

Figure 2–4 Uploading an Attachment to Oracle WebCenter Content

Add Attachment		×
Share wit	h 🔘 Only task participar 💿 All process participa	nts ants
* Attachment Typ	e 🔿 URL ⓒ Upload file to UCM ⓒ Associate UCM docu	ument
* File Name C:\IN	STALL.LOG	Browse
Description		
* Folder Name	1 I	
* Document Type	/Contribution Folders	~
Account	~	
Revision		
Release Date		20
Expiration Date		20
* Security Group	*	
		O <u>K</u> <u>C</u> ancel

Add Attachment	
Share with	 ○ Only task participants ⊙ All process participants
* Attachment Type	O URL O Upload file to UCM Associate UCM document
* File Name C:\IN Description	STALL LOG Browse
* Folder Name	/Contribution Folders/
* Document Type	×
Account	×
Revision	
Release Date	/Contribution Folders
Expiration Date	/Contribution Folders/DRA_UCM_Test
* Socurity Croup	/Contribution Folders/EURent
becancy Group	Contribution Folders/MyNewFolder1
	Contribution Folders/MyNewFolder2
	Contribution Folders/RTM1_Test
	/Contribution Folders/abcfolder
	/Contribution Folders/ccasares
	/Contribution Folders/ccasaresXX
	/Contribution Folders/douglasfolder
	/Contribution Folders/dthakkar
	/Contribution Folders/gsah
	/Contribution Folders/gsah2
	/Contribution Folders/liu1
	/Contribution Folders/myfolder
	/Contribution Folders/mvfolder1

Figure 2–5 Possible Destination Folders

- Document type: From this list, select the WebCenter Content document type.
- Account (optional): From this list, select the account name.
- Revision (optional): Type an integer for revision.
- Release date (optional): Select the date at which the document is released.
 Prior to this date, the WebCenter Content document is not visible to
 WebCenter Content users.
- Expiration date (optional): The date when the document expires. After this date, the WebCenter Content document is no longer visible to Oracle WebCenter Content users.
- Associate UCM document. Selecting this option opens another dialog box prompting you for the document identifier. You can search for it by clicking the spyglass icon and, after you have located it, clicking OK.

After you have selected the attachment type, click **OK**.

5. In the Task Details pane, from the Actions list, select **Save** as shown in Figure 2–6. The attachment is deleted from the task.

Note: If the attachment is stored in a content server such as WebCenter Content, then it is not deleted from the content server. If the attachment is stored in the SOA workflow repository, then you can access it by viewing the task snapshot from the task history.

Figure 2–6	Saving a	n Attachment	to a	Task
------------	----------	--------------	------	------

RACLE [®] Business Process	Workspace	Tasks Pro	cess Tracking Standar	d Dashboards M	Dre ↓	62	Home Prefere	nces Help Lo	gout Jo	ohn Stei	inbeck 9
ly Tasks(4) 🗸 🛛 🛛 Actions 🗸 🧯	Assignee Me & My Group	State Assigned	Search	n	Advanced	E					
Title						Process Name	Creator	Priority	Assig	ned	
Approve						ApproveRequest		3 💌	Jul 3	, 2012	3:29 PM
simple task						SimpleProcess		3	Jul 3	, 2012	9:36 PI
LoanApplication_Lab1						LoanRequestPro	tess	3	Jul 4	, 2012	10:19
Submit request						ApproveRequest	Proci	3	Jul 6	, 2012	9:23 Pf
Approve (i) Approve Reje	t Save Claim Retion Re a. To remove this messa Su Notes: With	set Information assign Rate Subtask calate spend hdraw						History	«	>	×
denoral Address	Ski	p Current Assignment									
Name *	Priority * Sa	/e									
	i 🔿 High 🚬										
	🔘 Medium										
Email *	O Low										
	Ctoture #										
Id*	Status	~									
· · · ·	· · · •								_	_	

To delete an attachment from a task:

- 1. In the Tasks page, from the worklist pane, select the task from which you want to delete an attachment. The details for that task appear in the Task Details pane.
- **2.** In the Task Details pane, in the Attachments section, select the attachment you want to delete and click **Delete**. The Delete Attachment dialog box prompts you to confirm the deletion from UCM.
- **3.** Confirm the deletion and click **OK**.

2.5 Requesting Information in Process Workspace

To request information:

1. From the Task Details page, from the **Actions** list, select **Request Information**, as shown in Figure 2–7.

Figure 2–7 Requesting Information

Actions 👻 🛛 🙀 Assignee My 8	Group 💌 Status 4	Assigned 📉	Search	Advanced	
Title	Number	Priority Assignee	s State	Created	Expires
Vacation Request for jcooper	200203 3	3 jstein (U)	Assigned	Mar 16, 2009 2:16 PM	
Vacation Request for jcooper Details Assignees Creator Created Mar 16, 2009 2:16 PM Updated Mar 16, 2009 2:28 PM Econtents	Expiration date Acquired By Task Number 20020 Priority 2	State .	Assigned		Task Actions Reject Ar Request Information Reassign Escalate Suspend Save Save

2. Request information from a task participant or from another user, as shown in Figure 2–8.

Figure 2–8 Requesting Information from Past Approvers or Another User, or Pushing the Task Back

Request More	Information
From Othe	r users
Comments:	.::
Return Options	 ○ Route directly back to me ⑦ Require subsequent participants to retake action

If you use the **Search** icon to find a user name, the Identity Browser appears, as shown in Figure 2–9.

Figure 2–9 Identity Browser

Identity Browser							×
Routing type Single Approver	~						
All 💙 istein			Search	Reset			
ID	First Name	Last Name	Email	Title	Manager	Organization	
🔲 🐧 jstein	John	Steinbeck	jstein@emailExample.com	Manager2	wfaulk	Demo User	
							L _R
Selected							OK Cancel

In Figure 2–9, in the ID column, the identifier, jstein, is a link. Clicking this link displays a details window with more information about this user.

Note: If you are in a multi-tenancy environment, search for a user simply by the user identifier and not by the tenant identifier. For example, if the user identifier is jstein and the tenant identifier is company_name.jstein, you would search by using jstein.

- **3.** Add your comment.
- **4.** Choose one of the following:
 - Route directly back to me
 - Require subsequent participants to retake action (Previous approvers must reapprove the task after the user submits the information)
- 5. Click OK.

2.6 Setting Rules, Vacation Periods, and Notification Preferences in Process Workspace

Rules act on tasks, either a specific task type or all the tasks assigned to a user or group. To set rules, from the Process Workspace toolbar, select **Preferences**. The Preferences window appears as shown in Figure 2–10.

Figure 2–10 Configuration Areas: My Rules Page

	rocess Workspace	🝓 Home Administration Preferences Help Logout weblogic 🧲
Preferences	My Rules Other Rules	
User Preferences Certificates	✓ Rules ▽ △ ⊕ ※ ✓ Vacation Period (Disabled)	Vacation Period : Revert
Rules Notification Preferences Accessibility	My Rules	Remove yourself from automatic task assignment by enabling a vacation date range. Optionally, more specific vacation rules can be created under "My Rules".
		Enable vacation
		Start Date 5
		O Reassign to: User V Q
	•	4
	4	۲

Copyright © 2005, 2012, Oracle and/or its affiliates. All rights reserved.

A rule cannot always apply to all circumstances in which it is used. For example, if a rule applies to multiple task types, it may not be possible to set the outcome for all tasks, because different tasks can have different outcomes.

Rules are executed in the order in which they are listed.

If a rule meets its filter conditions, then it is executed and no other rules are evaluated. For your rule to execute, you must be the only user assigned to that task. If the task is assigned to multiple users (including you), the rule does not execute.

You cannot specify business rules for ToDo tasks.

This section contains these topics:

- Section 2.6.1, "How to Create User Rules in Process Workspace"
- Section 2.6.2, "How to Create Group Rules in Process Workspace"
- Section 2.6.3, "How to Enable a Vacation Period in Process Workspace"
- Section 2.6.4, "Specifying Notification Settings"

2.6.1 How to Create User Rules in Process Workspace

Note: The task type browser does not support a search using NLS strings for task name and category.

To create a user rule:

1. From the Process Workspace toolbar, select **Preferences**. The **User Preferences** pane is displayed.

- **2.** In the User Preferences area, click **Rules**. The My Rules and Other Rules pages appear in the right pane. The My Rules page is selected by default.
- **3.** In the **My Rules** page, in the Rules section, select **My Rules**, then click **Add New Rule**. The My Rule screen appears in the right pane.
- 4. Provide a name for the rule.
- **5.** If the rule is a vacation rule, select **Use as a vacation rule**. The start and end dates of the rule are automatically synchronized with the vacation period.

For instructions about setting a vacation period that is synchronized with the vacation rule, see Section 2.6.3, "How to Enable a Vacation Period in Process Workspace."

6. From the Tasks list, select either All Tasks or Tasks matching these conditions.

If you select **All Tasks**, then no conditions need to be specified. If you select **Tasks matching these conditions**, then specify the conditions by clicking **Add Condition**. The conditions are described in Table 1–9 on page 1-19. You cannot search by translated text.

These conditions further define the rule, such as specifying that a rule acts on priority 1 tasks only, or that a rule acts on tasks created by a specific user. The conditions can be based on standard task attributes and any mapped attributes that were mapped for the specific tasks. See Section 9.4.1, "How to Specify Mapped Attributes" for more information.

- 7. Select Execute rule only between these dates and provide rule execution dates.
- **8.** In the **ACTION** area, select actions to be taken. User rules do the following actions:
 - Reassign to—You can reassign tasks to subordinates or groups you manage. If you were granted the BPMWorkflowReassign role, then you can reassign tasks to any user or group.
 - Delegate to—You can delegate to any user or group. Any access rights or privileges for completing the task are determined according to the original user who delegated the task. (Any subsequent delegations or reassignments do not change this from the original delegating user.)
 - Set outcome to—You can specify an automatic outcome if the workflow task was designed for those outcomes, for example, accepting or rejecting the task. The rule must be for a specific task type. If a rule is for all task types, then this option is not displayed.
 - Take no action—Use this action to prevent other more general rules from applying. For example, to reassign all your tasks to another user while you are on vacation, except for loan requests, for which you want no action taken, then create two rules. The first rule specifies that no action is taken for loan requests; the second rule specifies that all tasks are reassigned to another user. The first rule prevents reassignment for loan requests.
- 9. Click Save.

The new rule appears under the **My Rules** node.

2.6.2 How to Create Group Rules in Process Workspace

Note: The task type browser does not support a search using NLS strings for task name and category.

Creating a group rule is similar to creating a user rule, with the addition of a list of the groups that you (as the logged-in user) manage. Examples of group rules include:

- Assigning tasks from a particular customer to a member of the group
- Ensuring an even distribution of task assignments to members of a group by using round-robin assignment
- Ensuring that high-priority tasks are routed to the least busy member of a group

Group rules do the following actions:

- Assign to member via—You can specify a criterion to determine which member of the group gets the assignment. This dynamic assignment criterion can include round-robin assignment, assignment to the least busy group member, or assignment to the most productive group member. You can also add your custom functions for allocating tasks to users in a group.
- Assign to—As with user rules, you can assign tasks to subordinates or groups you
 directly manage. If you have been granted the BPMWorkflowReassign role, then
 you can reassign tasks to any user or group (outside your management hierarchy).
- Take no action—As with user rules, you can create a rule with a condition that
 prevents a more generic rule from being executed.

To create a group rule:

- 1. From the Process Workspace toolbar, select **Preferences**. The Preferences window is displayed.
- **2.** In the User Preferences area, click **Rules**. The My Rules and Other Rules pages appear in the right pane. The My Rules page is selected.
- **3.** Click the **Other Rules** tab.
- 4. Select My Group from the list.
- 5. Enter a group name, or click the **Search** icon to browse for a group.

The Identity Browser dialog box opens for you to find and select a group.

6. Select the group name under the **Group Rules** node and click **Add New Rule**, as shown in Figure 2–11.

Figure 2–11 Creating a Group R	ule
--------------------------------	-----

ORACLE Business Pr	rocess Workspace			Real Home Administration	Preferences Help Logout weblogic 🤶
Preferences	My Rules Other Rules				
User Preferences	✓ Rules	🔻 🛥 🕂 🗙 🛛 Group Rules			Save Revert
Certificates	My Group 💙 California 🔍	Show Rules Name * Californ	ia Group Rule		<u>~</u>
Rules Notification Preferences Accessibility	✓ Group Rules ✓ ∰ California ♦ California Group Rule	Execute rul	Execute rule only between these dates: Start Date		
		End Date	26		
		TASKS Tasks mate	hing these conditions 💟		
			R	Match 💫 All 💿 Any 🛖	=
		ACTION			
	4	J	Assign to member via: Assign to: Take no action	Least Busy 💙 User 💙	,
opyright @ 2005_2012_Oracle and/or.its	affiliates All rights reserved				

- 7. Select **Execute rule only between these dates** and provide rule execution dates.
- 8. From the Tasks list, select either All Tasks or Tasks matching these conditions.

If you select **All Tasks**, then no conditions need to be specified. If you select **Tasks matching these conditions**, then specify the conditions by clicking **Add condition**. The conditions are described in Table 1–9 on page 1-19.

These conditions further define the rule, such as specifying that a rule acts on priority 1 tasks only, or that a rule acts on tasks created by a specific user. The conditions can be based on standard task attributes and any mapped attributes that were mapped for the specific tasks. See Section 9.4.1, "How to Specify Mapped Attributes" for more information.

- **9.** In the **ACTION** area, select actions to be taken. User rules do the following actions:
 - Reassign to—You can reassign tasks to subordinates or groups you manage. If you were granted the BPMWorkflowReassign role, then you can reassign tasks to any user or group.
 - Delegate to—You can delegate to any user or group. Any access rights or privileges for completing the task are determined according to the original user who delegated the task. (Any subsequent delegations or reassignments do not change this from the original delegating user.)
 - Set outcome to—You can specify an automatic outcome if the workflow task was designed for those outcomes, for example, accepting or rejecting the task. The rule must be for a specific task type. If a rule is for all task types, then this option is not displayed.
 - Take no action—Use this action to prevent other more general rules from applying. For example, to reassign all your tasks to another user while you are on vacation, except for loan requests, for which you want no action taken, then create two rules. The first rule specifies that no action is taken for loan requests; the second rule specifies that all tasks are reassigned to another user. The first rule prevents reassignment for loan requests.
- 10. Click Save.

2.6.3 How to Enable a Vacation Period in Process Workspace

You can set a vacation period so that you are removed from automatic task assignment during the dates you specify.

Vacation rules are not executed for ToDo tasks.

For instructions on setting a vacation rule that is synchronized with the vacation period, see Section 2.6.1, "How to Create User Rules in Process Workspace".

To enable a vacation period:

- 1. Click the **Preferences** link. The Preferences page appears.
- **2.** In the User Preferences panel, click **Rules**. The middle pane displays the Rules panel with the My Rules page selected. Adjacent and to the right of the My Rules page, the Vacation Period panel appears as shown in Figure 2–12.

Figure 2–12 My Rules - Vacation Period

ORACLE	Business	Process	Workspace
--------	----------	---------	-----------

Preferences	My Rules		
User Preferences	🗹 Rules 🗸 🕂 💥	Vacation Period :	
Certificates	Vacation Period (Disabled)		
Rules	User Rule 1	Remove yourself from systematic task assignment by epobling a vacation data range	
Notification Preferences		Optionally, more specific vacation rules can be created under "My Rules".	
Accessibility	EasternRegion		
	🖓 WesternRegion	Enable vacation period	
	LargeTestGroup		
		Start Date	
		End Date	
		Delegate to:	

- **3.** Select Enable vacation period.
- 4. Provide start and end dates.
- **5.** Select either **Reassign to:** or **Delegate to:** If you use the search icon to find a user the Identity Browser windows appears, as shown in Figure 2–13.


Figure 2–13 Identity Browser Window

Search for and select a user and click **OK**.

6. Click Save. The vacation period is enabled.

2.6.4 Specifying Notification Settings

You can configure the notification settings to control how, when, and where you receive messages in cases when you have access to multiple communication channels (delivery types). Specifically, you can define messaging filters (delivery preferences) that specify the channel to which a message should be delivered, and under what circumstances.

For example, you might want to create filters for messages received from customers with different Service Level Agreements (SLA), specifying to be notified through business phone and SMS channels for customers with a premium SLA and by EMAIL for customers with a nonpremium SLA.

2.6.4.1 Messaging Filter Rules

A messaging filter rule consists of *rule conditions* and *rule actions*. A rule condition consists of a rule attribute, an operator, and an associated value. A rule action is the action to be taken if the specified conditions in a rule are true.

2.6.4.1.1 Data Types Table 2–1 lists data types supported by messaging filters. Each attribute has an associated data type, and each data type has a set of predefined comparison operators.

Data Type	Comparison Operators
Date	isEqual, isNotEqual, isGreaterThan, isGreaterThanOrEqual, isLessThan, isLessThanOrEqual, Between, isWeekday, isWeekend
Time	isEqual, isNotEqual, Between
Number	isEqual, isNotEqual, Between, isGreaterThan, isGreaterThanOrEqual, isLessThan, isLessThanOrEqual
String	isEqual, isNotEqual, Contains, NotContains

Table 2–1 Data Types Supported by Messaging Filters

Note: The String data type does not support regular expressions.

2.6.4.1.2 Attributes Table 2–2 lists the predefined attributes for messaging filters.

Attribute	Data Type	
Total Cost	Number	
From	String	
Expense Type	String	
То	String	
Application Type	String	
Duration	Number	
Application	String	
Process Type	String	
Status	String	
Subject	String	
Customer Type	String	
Time	Time	
Group Name	String	
Processing Time	Number	
Date	Date	
Due Date	Date	
User	String	
Source	String	
Amount	Number	
Role	String	
Priority	String	
Customer Name	String	
Expiration Date	Date	
Order Type	String	
Organization	String	
Classification	String	
Service Request Type	String	

 Table 2–2
 Predefined Attributes for Messaging Filters

2.6.4.2 Rule Actions

For a given rule, a messaging filter can define the following actions:

- Send No Messages: Do not send a message to any channel.
- Send Messages to All Selected Channels: Send a message to all specified channels in the address list.
- Send to the First Available Channel: Send a message serially to channels in the address list until one successful message is sent. This entails performing a send to the next channel when the current channel returns a failure status. This filter action is not supported for messages sent from the human workflow layer.

2.6.4.3 Managing Messaging Channels

In Oracle BPM Worklist, messaging channels represent both physical channels, such as business mobile phones, and also email client applications running on desktops. Specifically, Oracle BPM Worklist supports the following messaging channels:

- EMAIL
- IM
- MOBILE
- SMS
- VOICE
- WORKLIST

Note the following about message channels:

- Addresses for messaging channels are fetched from the configured identity store.
- SMS and MOBILE notifications are sent to the mobile phone number.
- VOICE notifications are sent to the business phone number.
- No special notification is sent when the messaging channel preference is WORKLIST. Instead, log in to Oracle BPM Worklist to view tasks.
- EMAIL is the default messaging channel preference when a preferred channel has not been selected.

You can use the **Messaging Channels** tab to view, create, edit, and delete messaging channels.

2.6.4.3.1 Viewing Your Messaging Channels You can display your existing messaging channels.

To view messaging channels:

- 1. Click the **Preferences** link.
- 2. Click Notification Preferences.
- 3. Select the Messaging Channels tab.

The My Messaging Channels list appears (Figure 2–14) and displays the following information:

- **Name**: The name of the messaging channel.
- **Type**: The type of messaging channel, such as EMAIL or SMS.
- Address: The address for the channel, such as a phone number or email address.
- **Default**: Specifies whether this channel is the default messaging channel.

Figure 2–14 Messaging Channels

Messagin	g Channels	Mes	saging Filter:	5		
My Messa Configure d	iging Chan hannels to rece	i nels eive your n	otifications an	d alerts.		
View 🗸	🍄 Create	🥖 Edit	💥 Delete	🛃 Detach		
Name			Туре 🕈	Address		Default

4. Click **View > Columns** and select the columns to display or hide.

You can also click **View > Reorder Columns** to display a dialog to reorder the displayed columns.

Messaging channel names and addresses are retrieved from the underlying identity store, such as Oracle Internet Directory.

2.6.4.3.2 Creating, Editing, and Deleting a Messaging Channel Oracle BPM Worklist uses an underlying identity store, such as Oracle Internet Directory, to manage messaging channels and addresses. Therefore, you cannot directly create, modify, or delete messaging channels using Oracle BPM Worklist.

To perform these actions, contact the system administrator responsible for managing your organization's identity store.

2.6.4.4 Managing Messaging Filters

You can use the **Messaging Filters** tab to define filters that specify the types of notifications you want to receive along with the channels through which to receive these notifications. You can do this through a combination of comparison operators (such as *is equal to, is not equal to*), attributes that describe the notification type, content, or source, and notification actions, which send the notifications to the first available messaging channels, all messaging channels, or to no channels (effectively blocking the notification).

For example, you can create a messaging filter called *Messages from Lise*, that retrieves all messages addressed to you from your boss, Lise. Notifications that match all of the filter conditions might first be directed to your business mobile phone, for instance, and then to your business email if the first messaging channel is unavailable.

2.6.4.4.1 Viewing Messaging Filters You can display your existing messaging filters.

To view your messaging filters:

- 1. Click the Notification tab.
- 2. Click the Messaging Filters tab.

The My Messaging Filters list appears (Figure 2–15) and displays the following information:

- Filter Name: The name of the messaging filter
- Description: An optional description of the messaging filter

Figure 2–15 Messaging Filters

Messagin	g Channels	Mess	aging Filters	5
My Messa Configure ru	ging Filters Jles to filter your	notificati	on and alert me	nessages.
View 👻	Create	🥖 Edit	💥 Delete	Detach
Filter Name			Description	n 💦 .

3. Click View > Columns and select the columns to display or hide.

You can also click **View > Reorder Columns** to display a dialog to reorder the displayed columns.

2.6.4.4.2 Creating Messaging Filters To create a messaging filter:

1. Click Create.

The Messaging Filters page appears, as shown in Figure 2–16.

erences	Messaging Channels Messaging Filters		
Preferences	Your Reference System Time: Monday, January 9, 2012 2:30:08 PM PST		OK Cancel
rtificates	* Filter Name:		
es	Description:		
tification Preferences	Condition		
cessibility	Matching: All of the following conditions		
	Add Filter Condition: To	4	
	Attribute Operator Value V	alue2 (if required)	Delete
	Action Messaging Option: Send No Messages 💌 Add Notification Channel: 💽 🗣		
	Action Messaging Option: Send No Messages Add Nobtification Channel: 💌 💠 Channel Address	Up	Down Delete
	Action Messaging Option: Send No Messages 💌 Add Notification Channel: 💌 🗣 Channel Address	Up	Down Delete
	Action Messaging Option: Send No Messages Add Nobtrication Channel: Address Channel Address	Up	Down Delete

Figure 2–16 Adding a Messaging Filter

2. Specify the following information:

- Filter Name: The name of the messaging filter.
- Description: An optional description for the messaging filter.
- **3.** Define the filter conditions using the lists and fields in the **Condition** section, as follows:
 - Select whether notifications must meet all of the conditions or any of the conditions by selecting either the All of the following conditions or the Any of the following conditions options.
 - **b.** Select the attribute from the list.
 - c. Select the operator, such as isEqual, from the list.
 - **d.** Type the value of the condition in the text box.
 - e. Click Add to add the condition to the list.
 - **f.** Repeat these steps to add more filter conditions. To remove a filter condition, click Delete.
- 4. Select from the following messaging options in the Action section:
 - Send No Messages: Do not send a message to any channel.
 - Send Messages to All Selected Channels: Send a message to all specified channels in the address list.
 - Send to the First Available Channel: Send a message serially to channels in the address list until one successful message is sent. This entails performing a send to the next channel when the current channel returns a failure status.

- **5.** To set the delivery channel, select a channel from the **Add Notification Channel** list and click **Add**. To remove a channel, click **Delete**.
- **6.** Use the up and down arrows to prioritize channels. If available, the top-most channel receives messages meeting the filter criteria if you select *Send to the First Available Channel*.
- 7. Click OK.

The messaging filter appears on the My Messaging Filters page. The My Messaging Filters page enables you to edit or delete the channel. Click **Cancel** to dismiss the dialog without creating the filter.

2.6.4.4.3 Editing a Messaging Filter To edit a messaging filter:

- 1. Select the filter on the My Messaging Filters page.
- 2. Click Edit.
- **3.** Click **OK** to update the messaging filter. Click **Cancel** to dismiss the dialog without modifying the filter.

2.6.4.4.4 Deleting a Messaging Filter To delete a messaging filter:

- 1. Select the filter on the My Messaging Filters page.
- 2. Click Delete. A confirmation dialog appears.
- **3.** Click **OK** to delete the messaging filter. Click **Cancel** to dismiss the dialog without deleting the filter.

2.7 Creating ToDo Tasks in Process Workspace

Use the Create ToDo Task dialog, shown in Figure 2–17, to create a top-level ToDo task for yourself or others. This task is not associated with a business task.

Figure 2–17 The Create To-Do Task Dialog

Create To-Do Task					\boxtimes
* Task Title					
Category					
Priority	1 💌				
Percentage Complete	50 韋				
Start Date		Ċ	6		
Due Date		Ċ	6		
Assignee	weblogic			Q	
				OK	<u>C</u> ancel

To-Do tasks appear in the assignee's **Inbox**.

To-Do tasks can be reassigned, escalated, and so on, and deleted (logical delete) and purged (physical delete). Reassignment, escalation, and so on of the task does not affect the assignment of any child ToDo tasks. The completion percentage of a ToDo task can be reset to less than 100% after it is completed.

Assignment rules (such as vacation rules) are not applied to ToDo tasks. You cannot specify business rules for ToDo tasks.

To create a ToDo task:

1. From the **Actions** list, select **Create To-Do Task**, as shown in the example in Figure 2–18.

Figure 2–18 Creating a To-Do Task

ORACLE' Business Proc	ess Workspace	Tasks Process Tracking Standard
Views Applications Links -	My Tasks(3) 🔻 🕂 🧪 💥 🛛 🖾	Actions V Kale Assignee Me & My Group
	Title	Complete
Inbox	💼 🖻 task 1	Create Sub Task
My Tasks (3)	subtask for task 1	Reassign
Initiated Tasks	🗊 task 2	Escalate
Administrative Tasks		Release
Standard Views		Suspend
Due Soon		Withdraw
High Priority		Purge
Past Day		Delete
Past Week		Create To-Do Task
Past Month		
Past Quarter	task 1	

- Provide details in the Create ToDo Task dialog, shown in Figure 2–17, and click OK.
 - **Task Title**: Enter anything that is meaningful to you.
 - **Category**: Enter anything that is meaningful to you.
 - **Priority**: Select from **1** (highest) to **5** (lowest)
 - **Percentage Complete**: This attribute indicates how much of the task is completed. 100% sets the attribute as completed.
 - **StartDate**: The task start date.
 - **Due Date**: The due date does not trigger an expiration. You can also see overdue tasks. The start date need not be the current date.
 - Assignee: You can assign yourself or someone else.
- 3. Click OK.

2.8 Creating Subtasks in Tasks in Process Workspace

A subtask is a child of a task, either a To-Do task or a business task. Creating a subtask can be helpful, for example, when a purchase order contains several line items and you need a separate approval process for one of them.

2.8.1 What You May Need to Know About Creating Subtasks

Here are some things to keep in mind when creating subtasks:

• A ToDo task can only have ToDo subtasks. When all ToDo subtasks are 100% complete, the ToDo task is also marked as complete. If the ToDo task is completed, then ToDo subtasks are at 100% within the workflow system.

If the task is a business task and that task is completed, then the subtasks of that task are withdrawn. You can only create a business tasks for other business tasks.

• If you explicitly set a ToDo task to 100%, there is no aggregation on the task.

• If you are using a release of Oracle BPM that is prior to 11g Release 1 (11.1.1.7.0), then, when you create a subtask from the **Actions** menu of the task list, that subtask will appear in the task list of the assignee for that subtask. However, to enable subtask creation from the **Actions** menu of the task details page and display of subtask section in the task details page, you must re-generate the form by using 11g Release 1 (11.1.1.7.0).

You may continue to use processes that were deployed in earlier releases even if you do not re-create the task form.

- Sub Tasks created in Process Workspace are not included as composite audit trail in Oracle Enterprise Manager Cloud Control Console.
- Parallel tasks are created when a parallel flow pattern is specified for scenarios such as voting. In this pattern, the parallel tasks have a common task. You can create sub tasks using different pattern lines: Single approver, Sequential approver and Parallel approver.

The sub tasks are included in the Tasks Details page of the task. To view the sub tasks for a task, select the task from the Task List to display the Task Details page. The sub tasks are displayed under the **Sub Tasks** section.

2.8.2 How to Create a SubTask

This section includes the following topics:

- Section 2.8.2.1, "Creating SubTasks from the Actions List"
- Section 2.8.2.2, "Creating SubTasks from an Existing Task"

2.8.2.1 Creating SubTasks from the Actions List

- 1. In the Task List, select the task for which you want to create a subtask.
- 2. From the Actions list, select Create Sub Task as shown in Figure 2–19.

Figure 2–19 Selecting to Create a Sub Task



The Create Subtask dialog box appears as shown in Figure 2–20.

Create Subtask For 200000 : task 1								
* Task Title								
Category								
Priority 1								
Percentage Complete 50 🚔								
Start Date	20							
Due Date	20							
Assignee weblogic	0	\$						
		O <u>K</u> <u>C</u> ancel						

Figure 2–20 Create Subtask Dialog Box - Task List

- **3.** In the Create Subtask dialog box, define the subtask and provide the following details, if required:
 - Task Title: Enter anything that is meaningful to you (required field).
 - **Category**: enter anything that is meaningful to you.
 - **Priority**: Select from 1 (highest) to 5 (lowest).
 - Percentage Complete: This attribute indicates how much of the task is completed. 100% sets the attribute as completed.
 - Start Date: The task start date.
 - **Due Date:** The due date does not trigger an expiration. You can also see overdue tasks. The start date need not be the current date.
 - Assignee: You can assign yourself, someone else or more than one participant (separating each ID with a comma).
- **4.** When you have finished specifying the subtask, in the Create Subtask dialog box, click **OK**. This refreshes the task list. When you select the task, the Task Details page now includes a Subtasks section displaying the details about the subtask you created for that task.

Note:

- If you specified more than one participant for the subtask, then the Subtask region displays a separate item for each participant.
- If a participant completes a subtask, then you must manually refresh the task to show the details for that completed subtask.
- If you create a sub task, the assignee becomes the creator.

2.8.2.2 Creating SubTasks from an Existing Task

- 1. In the Task List, select the task for which you want to create a subtask.
- **2.** From the Task Details page, go to the **Actions** list, select **Create Sub Task** as shown in Figure 2–21.

Figure 2–21 Selecting to Create a Sub Task from a the Task Details Page

Bug14231349			Approve Reject A	ctions -	4	
🖂 🦰 Details			Request Information	n		
Assignees John Steinbeck Creator James Cooper	Expiration Date	Priority 3	Create Subtask			
Created Sep 14, 2012 8:24 AM	Acquired By	State Hanghou	Adhoc Route			
Updated Sep 17, 2012 12:03 AM	Outcome Attachment Added Task Number 200470		Escalate Suspend			

The Create Subtask dialog box appears as shown in Figure 2–22.

Figure 2–22 Create Subtask Dialog Box - Task Details Page

Creat	Create Subtask For 200030 : Folder Edit test								1
	* Title Sub Task for Help Desk - Review								
	Priority	3	1						
	Form Default Form - Standard task form for displaying task details.								
Com	ments:	Plea	ise review						
Rou	ting ty	pe S	ingle Approv	er	~				
All			ystein				Search R	eset	
	ID		First Name	Last Name	Email	Title	Manager	Organization	
	💧 js	tein .	John	Steinbeck	jstein@emailExample.	Manager2	wfaulk	Demo User	
Select	od	8 -	stais 😚						
Select	60	<u>w</u>)	stein 🦇						
								O <u>K</u> <u>C</u> ancel	

- **3.** In the Create Subtask dialog box, define the subtask and provide the following details, if required:
 - **Title**: Enter anything that is meaningful to you (required field).
 - Priority: Select from 1 (highest) to 5 (lowest).
 - **Comments**: Enter anything that is meaningful to you.
 - Routing Type: Select Single Approver, Group Vote or Chain of Single Approvers. For more information on these options, see Section 2.11, "Routing Tasks in Process Workspace".
 - Provide or search for user or group names and select the approvers by clicking the check box next to the user or group name.
- **4.** When you have finished specifying the subtask, in the Create Subtask dialog box, click **OK**. This refreshes the task list. When you select the task, the Task Details page now includes a Subtasks section displaying the details about the subtask you created for that task.

Note:

- If you specified more than one participant for the subtask, then the Subtask region displays a separate item for each participant.
- If a participant completes a subtask, then you must manually refresh the task to show the details for that completed subtask.
- If you create a sub task, the assignee becomes the creator.

2.9 Acting on Tasks that Require Passwords in Process Workspace

If a human task was set up to require a password, then when you act on it (approve or reject the human task), you must provide the password when prompted, as shown in Figure 2–23.

Figure 2–23 Acting on a Task That Requires a Password

acation R	equest for jcooper		
🗄 De	tails		
Contents			
Creator	jcooper		
From Date	Mar 15, 2009 5:00 PM		
To Date	Mar 21, 2009 5:00 PM		
Reason	Yearly vacation		
History	apshot V Future Participants] Full task actions	
# P	articipant		Action
L E	a 🛐 default		
1.1	🔏 jstein		Assigned
		Enter Password	
		* Password	
			OK Cancel

2.10 Acting on a Task that Requires a Digital Signature in Process Workspace

Oracle Business Process Management Workspace supports the signature policy created in the human task:

- No signature required Participants can send and act on tasks without providing a signature.
- Password required Participants must specify their login passwords.
- Digital certificate (signature) required —Participants must possess a digital certificate before being able to send and act on tasks. A digital certificate contains the digital signature of the certificate-issuing authority so that anyone can verify that the certificate is real. A digital certificate establishes the participant's credentials. It is issued by a certification authority (CA). It contains your name, a serial number, expiration dates, a copy of the certificate holder's public key (used for encrypting messages and digital signatures), and the digital signature of the certificate-issuing authority so that a recipient can verify that the certificate is real.

The evidence store service is used for digital signature storage and non-repudiation of digitally-signed human workflows.

For more information about digital signatures and the evidence store service, see Section 2.12, "Evidence Store Service and Digital Signatures".

When you act on a task that has a signature policy, the **Sign** button appears, as shown in Figure 2–24.



Figure 2–24 Digital Signature Task Details

To use a certificate for digitally signing tasks, you must first upload the certificate as described in Section 1.5, "Providing a Digital Certificate in Process Workspace" on page 1-30.

2.11 Routing Tasks in Process Workspace

If the human task was designed to permit ad hoc routing, the task can be routed in an ad-hoc fashion in the worklist. For such tasks, a **Route** button appears on the task details page. From the Route page, you can look up one or more users for routing. When you specify multiple assignees, you can select whether the list of assignees is for simple (group assignment to all users), sequential, or parallel assignment.

If you are a developer and want more information about designing tasks to permit ad hoc routing, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

To route a task:

1. From the Actions list, select Adhoc Route, as shown in Figure 2–25.

Figure 2–25 Ad Hoc Routing

Help Desk reques	t for wfaulk	Ta	sk Actions 👻 🛛 RESOLVED
Details Genteents			Request Information
Location	California		Reassign
Туре	Hardware		Adhoc Route
Problem Description	Unable to reboot the system		
Severity	2		Escalate
Status	Created		Suspend
Requester			Renew
First Name William			
Last Name Faulkne	r		DAVE

2. Select an action and a routing option, as shown in Figure 2–26.

Figure 2–26 Routing a Task

Route Task		×
RESOLVED and route to: Single Approver Group Vote Chain of Single Approvers	Comments:	
Groups V First name, Last name, ID	Search Reset	
Ataliable		ן ר
	Move	
	Move All	
		$\overline{\nabla}$
	Remove	Z
🗆 Details		
Please select an item to see its details		
	OK Can	ncel

- Single Approver: Use this option for a single user to act on a task. If the task is
 assigned to a role or group with multiple users, then one member must claim
 the task and act on it.
- Group Vote: Use this option when multiple users, working in parallel, must act, such as in a hiring situation when multiple users vote to hire or reject an applicant. You specify the voting percentage that is needed for the outcome to take effect, such as a majority vote or a unanimous vote, as shown in Figure 2–27.

Figure 2–27 Providing Consensus Information

Default cutcon e Reject	¥
Consensus Percentage 🛛 🔒	
	-0
Minimium agreement to override J	efaul

- Chain of Single Approvers: Use this option for a sequential list of approvers. The list can comprise any users or groups. (Users are not required to be part of an organization hierarchy.)
- 3. Add optional comments for the next participant on the route.
- **4.** Provide or search for user or group names; then move the names to the **Selected** area.
- 5. Click OK.

2.12 Evidence Store Service and Digital Signatures

The evidence store service is used for digital signature storage and non-repudiation of digitally-signed human workflows. A digital signature is an electronic signature that authenticates the identity of a message sender or document signer. This ensures that the original content of the message or document sent is unchanged. Digital signatures are transportable, cannot be imitated by others, and are automatically time-stamped. The ability to ensure that the original signed message arrived means that the sender cannot repudiate it later. Digital signatures ensure that a human workflow document:

- Is authentic
- Has not been forged by another entity
- Has not been altered
- Cannot be repudiated by the sender

A cryptographically-based digital signature is created when a public key algorithm signs a sender's message with a sender's private key.

During design time, signatures are enabled for the task. During runtime in Process Workspace, when a user approves or rejects the task, the web browser:

- Asks the user to choose the private key to use for signing.
- Generates a digital signature using the private key and task content provided by Oracle BPM Worklist.

Notes:

- The certificate refers to a Personal Information Exchange Syntax Standard (PFX) file that includes a certificate and a private key, and is protected by a simple text password. PFX specifies a portable format for storing or transporting a user's private keys, certificates, miscellaneous secrets, and so on.
- The possession of a private key that corresponds to the public key of a certificate is sufficient to sign the data, because the signature is verifiable through the public key in the certificate. However, no attempt is made to correlate the name of a user of a certificate with the person updating it. For example, user jstein can sign using the private key of user cdickens if jstein has that private key.

The following digital signature features are supported:

- PKCS7 signatures based on X.509 certificates
- Browser-based, digitally-signed content without attachments

You can search the evidence store, as shown in Figure 2–28.

Figure 2–28 The Evidence Store

My Tasks Initia	ted Tasks Administra	ation Tasks Adminis	ration Evidence	e Search			
Coarch Euida	aca Stora						
Search Evide	ice store						
Match 💿 A	I 🕘 Any						
Signer jstein		📎 Signed Afte	er 5/8/2009	12	Signed Before 5/3	31/2009	B 🕑
🗆 Advance	I						
Sign	ature Policy Digital Sig	nature 🗧 Ve	rified Date	20	Task Outcom	e	
	Status Success	😝 Ta	sk Number				
							Validate
	Signature Policy	Signer	Status	Creation Date	Signed Date		

2.13 Creating Worklist Reports

Table 2–3 lists the worklist reports available for task analysis.

Report Name	Description	Input Parameters
Unattended Tasks	Provides an analysis of tasks assigned to users' groups or reportees' groups that have not vet been	 Assignee—This option (required) selects tasks assigned to the user's group (My Group), tasks assigned to the reportee's groups (Reportees), tasks where the user is a creator (Creator), or tasks where the user is an owner (Owner).
	acquired (the "unattended"	Creation Date—An optional date range
	tasks).	 Expiration Date—An optional date range
		 Task State—The state (optional) can be Any, Assigned, Expired, or Information Requested.
		 Priority—The priority (optional) can be Any, Highest, High, Normal, Low, or Lowest.
Tasks Priority	Provides an analysis of the number of tasks assigned to a user, reportees, or their groups, broken down by	 Assignee—Depending on the assignee that you select, this required option includes tasks assigned to the logged-in user (My), tasks assigned to the user and groups that the user belongs to (My & Group), or tasks assigned to groups to which the user's reportees belong (Reportees).
	priority.	 Creation Date—An optional date range
		 Ended Date—An optional date range for the end dates of the tasks to be included in the report
		 Priority—The priority (optional) can by Any, Highest, High, Normal, Low, or Lowest.
Tasks Cycle Time	Provides an analysis of the time taken to complete tasks from assignment to	 Assignee—Depending on the assignee that you select, this required option includes your tasks (My) or tasks assigned to groups to which your reportees belong (Reportees).
	completion based on users' groups or reportees' groups.	 Creation Date—An optional date range
		 Ended Date—An optional date range for the end dates of the tasks to be included in the report
		 Priority—The priority (optional) can by Any, Highest, High, Normal, Low, or Lowest.
Tasks Productivity	Provides an analysis of assigned tasks and completed tasks in a given	 Assignee—Depending on the assignee that the user selects, this required option includes the user's tasks (My & Group) or tasks assigned to groups to which the user's reportees belong (Reportees).
	time period for a user, reportees, or their groups.	 Creation Date (range)—An optional creation date range. The default is one week.
		 Task Type—Use the Search (flashlight) icon to select from a list of task titles. All versions of a task are listed on the Select Workflow Task Type page (optional).
Tasks Time Distribution	Provides the time an assignee takes to perform a task.	 Assignee—Depending on the assignee that the user selects, this required option includes the user's tasks (My & Group) or tasks assigned to groups to which the user's reportees belong (Reportees).
		 Fromto (date range)—An optional creation date range. The default is one week.
		 Task Type—Use the Search (flashlight) icon to select from a list of task titles. All versions of a task are listed on the Select Workflow Task Type page (optional).

Table 2–3 Worklist Report Types

2.13.1 How To Create Reports

Reports are available from the **Reports** link. Report results cannot be saved.

To create a report:

1. From the toolbar at the top center of the Home page, select **More**, then the **Reports** link as shown in Figure 2–29.

Figure 2–29 Creating a Report

ORACLE' Business Pr	ocess Workspace	Tasks Process Tracking Standard Dash	boards More 🗸	62	Home Preferences He	lp Logout John Steinbeck 오
✓ Reports ☑ Unattended Tasks Report ☑ Inastended Tasks Report ☑ Tasks Cycle Time Report ☑ Tasks Productivity Report ☑ Tasks Time Distribution Report			Address Reports			
	•					

2. Click the type of report you want to create.

Figure 2–30 shows the report types available.

Figure 2–30 Oracle BPM Worklist Reports



3. Define the search parameters of the report.

Figure 2–31 shows an example of the Unattended Tasks Report input page. The other reports are similar. See Table 2–3 for information about input parameters for all the report types.

Figure 2–31 Unattended Tasks Report—Input Page for Task Analysis

Unattended Tas Provides an analysis	sks Report of tasks assigned to users' groups or reportees' group	os that need attention because they have not yet been acquired
Assignee My & Gr	oup 💌	Run
Creation Date	🖄 to	20
Expiration Date:	🖄 to	₿ <mark>8</mark>
Task Any State	•	
Priority Any		

4. Click Run.

2.13.2 What Happens When You Create Reports

As shown in Figure 2–32, report results (for all report types) are displayed in both a table format and a bar chart format. The input parameters used to run the report are

displayed under **Report Inputs**, in the lower-left corner (may require scrolling to view).



Figure 2–32 Report Display—Table Format, Bar Chart Format, and Report Inputs

2.13.2.1 Unattended Tasks Report

Figure 2–33 shows an example of an Unattended Tasks report.

Figure 2–33 Unattended Tasks Report



The report shows that the California group has 15 unattended tasks, the Supervisor group has 7 unattended tasks, and the LoanAgentGroup has 11 unattended tasks. The unattended (unclaimed) tasks in this report are all DocumentReview tasks. If multiple types of unattended task exists when a report is run, all task types are included in the report, and the various task types are differentiated by color.

2.13.2.2 Tasks Priority Report

Figure 2–34 shows an example of a Tasks Priority report.

Figure 2–34 Tasks Priority Report



The report shows that the California group, the Supervisor group, and the LoanAgentGroup each have 16 tasks of normal priority. The users rsteven and jcooper have 5 and 22 tasks, respectively, all normal priority. Priorities (highest, high, normal, low, lowest) are distinguished by different colors in the bar chart.

2.13.2.3 Tasks Cycle Time Report

Figure 2–35 shows an example of a Tasks Cycle Time Report.



Figure 2–35 Tasks Cycle Time Report

The report shows that it takes 1 hour and 6 minutes on average to complete DocumentReview tasks, and 1 hour and 28 minutes on average to complete VacationApproval tasks. The bar chart shows the average cycle time in milliseconds.

2.13.2.4 Tasks Productivity Report

Figure 2–36 shows an example of a Tasks Productivity Report.



Figure 2–36 Tasks Productivity Report

The report shows the number of tasks assigned to the California, LoanAgentGroup, and Supervisor groups. For individual users, the report shows that jcooper has 22 assigned tasks. In addition to his assigned tasks, jcooper has completed 2 tasks. The report shows that mtwain and rsteven have completed 6 and 11 tasks respectively. In

the bar chart, the two task states—assigned and completed—are differentiated by color.

Note: The **Me & My Group** and **Reportees** options have been removed from the Productivity Report.

Working on Process Instances in Process Workspace

This chapter, primarily for owners of process instances in Oracle Business Process Management Workspace, describes how to initiate and cancel process instances, search, view details, associate comments and attachments, perform bulk operations, and customize the presentation of process instances.

This chapter contains these topics:

- Section 3.1, "Initiating Process Instances in Process Workspace"
- Section 3.2, "Searching for Process Instances in Process Workspace"
- Section 3.3, "Viewing the Details of a Process Instance in Process Workspace"
- Section 3.4, "Performing a Bulk Operation in Process Workspace"
- Section 3.5, "Canceling a Process Instance in Process Workspace"
- Section 3.6, "Customizing the Presentation of Process Instances in Process Workspace"

For information about troubleshooting human workflow issues, see section "Human Workflow Troubleshooting" of Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite.

3.1 Initiating Process Instances in Process Workspace

To initiate a process instance, you begin with the Applications panel located at the top of the navigation pane.

To initiate a process:

1. In the Process Tracking page, in the Applications panel, click the process you want to initiate. The dialog box for that process instance appears.

Note: If your browser has pop-up blocking enabled, then enable pop-ups for task flow servers; otherwise the application does not launch and the dialog box does not appear.

2. Create any comments or add any attachments.

To add a comment:

a. In the comments section, click **Create**. The Create Comment dialog box appears.

b. Enter your comment and click **OK**. The comment is added at the process level.

To add an attachment:

- **a.** In the Attachments section, click **Add**. The Add Attachment dialog box appears.
- **b.** Select the attachment type. If you select Desktop File, a dialog box enables you to enter the path name of the file or browse to find it. If you select URL, you are prompted for the name of the file and the URL.
- **c.** After you have specified the attachment, click **OK**. The attachment is added at the process level.

The process instance is initiated and is assigned to you as the initiator.

3.2 Searching for Process Instances in Process Workspace

You can perform either a basic search or an advanced search for process instances. This section contains these topics:

- Section 3.2.1, "How to Perform a Basic Search for Process Instances"
- Section 3.2.2, "How to Perform an Advanced Search for Process Instances"

3.2.1 How to Perform a Basic Search for Process Instances

For a basic search, you enter all or part of the name of a process instance in the Search field.

To perform a basic search for a process instance:

Select the Process Tracking page and, in the Search field, enter all or part of the name of the instance for which you are searching. Note that the search results are based on the Description column of the Process Tracking page.

To retrieve a list of all instances, enter a percent sign (%).

Search results are displayed as shown in the example in Figure 3–1.

Figure 3–1 Search Results

Title	Number	Initiator	Process Name		Initiated		Status		
Instance #1 of RequestQuote	1	weblogic	[SalesQuoteProcess]	RequestQuote	Nov 10, 20:	0 10:41 AM	G		
									N
<									5
instance #1 of RequestQu	iote								
∀ Details									
Priority normal			Process [SalesQ	uoteProcess] Request	Quote	Creation t	ime 11/10/2010		
Status Running			Number 1						
♡ Open Activities									
Title					Prior	ity	State	Actions	
Enter Quote							Running	Run	
<									>
> Audit Trail									
√ Comments				∀ Attachments	9				
_				Elenamo	- Descrip	ion			
				rivename	Descrip				

The Details section displays the priority, process name, when the instance is due to complete, status, creation time, and the assignee.

If the process instance has any gateways resulting in multiple threads (paths) that are still running, the Open Activities section display the status of each thread.

The Audit Trail section displays audit information—in either tabular form, tree view, or graphically depending on your selection.

3.2.2 How to Perform an Advanced Search for Process Instances

You can perform a search based on one or more conditions you specify.

To perform an advanced search for a process instance:

- **1.** In the Process Tracking page, click **Advanced**. The Advanced Search dialog box appears.
- **2.** From the Processes list, select each of the processes you want to include in the view.
- **3.** From the Assigned list, select one of the following: Anyone, at Any Role; Me, or Unassigned in my Role(s); Me.
- **4.** From the Status list, select one the following:

Status	Description
All	Displays all instances.
All Open	The work items displayed include all instances that have the following status: In Process, Pending Migration, Suspended, and Errored.
In Process	The work item is active; it has not reached the end of the process.
Completed	The work item reached the end of the process.
Aborted	The work item is no longer active, and did not reach the end of the process.
Errored	The work item has thrown an exception during processing.
Pending Migration	The work item is awaiting migration to a new implementation.
Suspended	The work item is in suspended state. An item is suspended if a new implementation of the same composite revision is deployed, and the previously deployed composite has some components with the newer BPM process definition.

Table 3–1 Add View Status List

5. Click OK.

3.3 Viewing the Details of a Process Instance in Process Workspace

The ability of a user to see process instances depends on the user's roles. Table 3–2 maps the various roles to their corresponding levels of visibility.

Role	Visibility into Process Instance
Administrator—that is, a user with the WorkflowAdmin role or the Administrator role	Can see all information related to all process instances
Process owner	At any given time, can see the instance details of the process instance
User with a swimlane role	Can see any unassigned instance associated with that role
Task assignee	Can see the instance details of that task

Table 3–2Visibility into Process Instances

To view the details of a process instance:

In the Process Tracking page, from the process instance list, select the instance whose details you want to view. The Work Item Detail panel opens in the lower right section of the page as shown in Figure 3–2.

Figure 3–2 Work Item Details Panel

Actio	ns 👻 Views -	🔹 🔞 Search			🥔 🕑 Advanced	1	
Titl	le			Number	Initiator	Process Name	Initia
Net	w title modify D	c		33		[Project3 v1.0] Review	Process Oct :
Ins	tance #50 of O	nboardingProcess		50		[M4HCMDemo v1.0] Or	boarding Oct :
Ins	tance #51 of O	nboardingProcess		51		[M4HCMDemo v1.0] Or	boarding Oct !
Ins	tance #52 of Re	eviewProcess		52		[Project3 v1.0] Review	Process Oct :
<							>
ew t	title modify	DO					Actions 🗸
⊻De	t ails Priority Status	normal Running	Process [Project3 v. ReviewProc Number 33	1.0] :ess	Crea	tion time 10/10/2011	^
≥ Op	en Activities						
Activ	rity	Title	Assigned to		Priority	State	Acl
Very	ExpensiveUserF	Revery Expensive Rev	ieProject3.ProcessOw	ner	-	Running	Ru
Simp	leUserReview	SimpleUserReview	Project3.ProcessOw	ner	-	Running	Ru
1							
_ز							>
							>
⊻ Au	ıdit Trail						>
⊻ Au Li	i dit Trail st View 🔉	 Human Activities; 	Service Activiti 💌 🛽 I	nput Messaç	je Output Messag	e Comment	
⊻ Au Li	idit Trail st View	 Human Activities; 	Service Activiti 💌 II	nput Messaç	re Output Messag	e Comment	
Au Li Activ	i dit Trail st View • ilty VeryExpensiv	 Human Activities; reUserReview 	Service Activitir 💌 In Status Activity mo	nput Messaç ved	je Output Messag	e Comment tart ct 10, 2011 2:44:33 PM	
✓ Au Li	idit Trail st View • ity VeryExpensiv SimpleUserRe	 Human Activities; reUserReview wiew 	Service Activiti 💌 Ir Status Activity mo Activity mo	nput Messag wed wed	re Output Messag 9 0 0	Comment tart et 10, 2011 2:44:33 PM et 10, 2011 2:44:33 PM	
Auria Activ	idit Trail ist View ity VeryExpensiv SimpleUserRe VeryExpensiv	Human Activities; reUserReview view reUserReview	Service Activiti 💌 II Status Activity mo Activity mo Activity cor	nput Messag wed wed mpleted	je Output Messag Si O O O O	Comment tart et 10, 2011 2:44:33 PM et 10, 2011 2:44:33 PM et 10, 2011 2:59:55 PM	
Auth	Idit Trail Ist View VeryExpensiv SimpleUserRe VeryExpensiv FinalApprova	Human Activities; reUserReview view I	Service Activiti 💌 🛽 Status Activity mo Activity mo Activity con	nput Messag ved ved mpleted ved	je Output Messag S O O O O O O O	e Comment tart et 10, 2011 2:44:33 PM et 10, 2011 2:54:53 PM et 10, 2011 2:59:55 PM et 10, 2011 3:02:03 PM	

For information about the Work Item Details panel, see Section 1.3.4.1, "Navigating the Process Workspace Process Tracking Page."

You can also use this panel to create comments and add attachments as described in Section 3.3.2, "Associating Comments and Attachments with a Process Instance in Process Workspace"

3.3.1 Viewing the Audit Trail in Process Workspace

The Audit Trail section of the Work Item Details panel displays audit details in a list view, a tree view, or a graphical view. Select the view from the list in the upper left corner of the Audit Trail section as shown in Figure 3–3.

⊻ Audit Trail	Цŝ	
List View 💌	Human Activities;Service Activiti 💌	Input Message
List View		
A(Treee View	Status	
Graphical View	Activity	moved
ApproveTask	Activity	processing

Figure 3–3 Selecting the Audit Details View

You can select the type of activity, sub-process, gateway, or event you want to audit. To do this, select from the list at the top of the Audit Details section as shown in Figure 3–4.

Figure 3–4 Selecting Activities for Audit Details

⊻ Audit Trail			
List View 💌	Human Activities;Service Activiti	Input Message	Output Message Comr
Activity	All		Start
ApproveTask	Service Activities	/ moved	Oct 7, 201
ApproveTask	Business Rule Activities	/ processing	Oct 7, 201
	Sub-processes		
	Events		
	Gateways		
	Script Activities		
	Other Activities		

If you are an administrator or process owner, and if the selected activity includes an input or output payload, then you can display the input or output message for that activity. To do this, select the activity, then click either **Input Message** or **Output Message**, The message appears in a pop-up as shown in Figure 3–5.

Figure 3–5 Displaying Activity Input and Output Messages



3.3.2 Associating Comments and Attachments with a Process Instance in Process Workspace

To associate a comment with a process instance:

- **1.** From the worklist, select the process instance with which you want to associate a comment.
- 2. In the details panel, in the Comments section, click New.
- 3. Enter your comment and click OK.

To associate an attachment with a process instance:

- **1.** From the worklist, select the process instance with which you want to associate an attachment.
- **2.** In the details panel, in the Attachments section, click **New**. The Add Attachment dialog box appears.
- 3. In the Description field, optionally enter a description for the attachment.
- 4. In the Filename field, enter the path to the file, or click **Browse** to navigate to it.
- 5. Click OK.

Note:

- If you add an attachment to a process instance, and later delete it from the corresponding task, then the attachment is also deleted from the process instance.
- In an environment with servers clustered for high availability purposes, file uploading is not supported in the event of a failover. If the active server shuts down, then the uploading process is not assumed by the other server, and the upload fails.

3.4 Performing a Bulk Operation in Process Workspace

In this release, the only available bulk operation you can perform is Cancel.

To perform a bulk operation:

- 1. In the Process Tracking page, from the process instance list, select several instances. In the Instance Details panel, a message indicates how many instances are currently selected to be processed. Also, the Actions list is updated with the options that can be performed over the group of instances that were selected.
- **2.** Select the operation in the Actions list. The selected operation is performed on the instances you specified.

3.5 Canceling a Process Instance in Process Workspace

Only process owners can cancel a process instance.

To cancel a process instance:

• From the Actions list at the end of the row for the instance you want to cancel, select **Cancel**.

• Select the instance and, from the Actions list on the toolbar of the Process Tracking page, select **Cancel**

3.6 Customizing the Presentation of Process Instances in Process Workspace

A presentation is the column arrangement used when viewing instances in the Work Items panel. You can use an existing presentation or create a new one to be associated with that view.

You can choose the columns that are presented in any view displayed in the Work Items panel.

To customize the presentation of a process instance:

- 1. From the Views panel, select the view whose presentation you want to customize. The view is displayed in the Work Items panel.
- **2.** From the Work Items toolbar, click **Edit Presentation**. This displays the Presentation dialog box in which you can select and sort the columns in your presentation.
- **3.** From the Available Columns list, select the columns you want to include in your view and move them to the Presentation Columns list by clicking the right arrow.
- **4.** To sort the columns:
 - a. Select an individual item in the Presentation Columns list.
 - **b.** Move it into place by using the vertical arrows to the right of the list.
 - c. Repeat for each column you want to sort.
- 5. Click OK.

Note: The Work Items panel displays process instances in two places: in the Process Tracking page and when using the Standard Dashboards page. You can customize the presentation of this panel as it appears in both places. However, if you first change the presentation in the Process Tracking page, and then change the presentation in the Standard Dashboards page, the latter configuration overrides whatever presentation you have configured in the Process Tracking page. When you navigate back to the Process Tracking page, the presentation will be whatever you configured in the Standard Dashboards page.

Working with Activity Guides in Process Workspace

This chapter describes activity guides in Oracle Business Process Management Workspace. They can organize and lead you through the various processes in which you have assigned tasks.

This chapter contains these topics:

- Section 4.1, "Understanding Activity Guides"
- Section 4.2, "Navigating Activity Guides"
- Section 4.3, "Showing or Hiding the Activity Guide Tab Page"

4.1 Understanding Activity Guides

Activity guides organize your tasks into milestones. Each milestone is a specific set of tasks for you to complete, either on your own or in collaboration with others. For each milestone, the activity guide tells you which tasks to complete and in which order. You complete a milestone by completing all the tasks it contains for you. Depending on the nature of the task, you can save an unfinished task and resume it later.

Figure 4–1 shows an example of an activity guide.

Figure 4–1 Sample Activity Guide

Boarding guided process-6 Boarding guided process-5 View the status and		Initiated	Instance #	Process Name	Composite Name
Boarding guided process. View the status and	In Progress	Jun 26, 2011 2:59 PM	6	OnBoarding guided process	OnBoardingProcess!1.0
boarding guided process-5 percent completion (In Progress	Jun 26, 2011 2:59 PM	5	OnBoarding guided process	OnBoardingProcess!1.0
Boarding guided process-4 the overall business	In Progress	Jun 26, 2011 2:59 PM	4	OnBoarding guided process	OnBoardingProcess!1.0
Boarding guided process-3 process in which you	u In Progress	Jun 26, 2011 2:59 PM	3	OnBoarding guided process	OnBoardingProcess!1.0
participate.					
ivity Guide 🛛 🐻 🚽					
tivity Guide Completion	Validate personal info				Actions - Ok No
1000	✓ Contents				
5 100%	Employee Id jcooper				
mpieted lasks (U/1)	First Name John				
sk Filter 🛛 All 🛛 👻	Last Name Cooper				
	User Info - Address				
w •	Street 1156 High Street				
OnBoarding guided process	City Santa Cruz				
	State California				
* Personal Info	Zin Code 95064				
R M Unlighte services al infe	Country United States				
a waldate personal into	country control states				
	✓Comments		\$	✓ Attachments	+ ×
	Manalaka ka akaalayy			No data to display	
- 🦀 Logistics	No data to display				
Logistics Expand a miles	stone to see and act on				
- Cogistics Expand a miles	stone to see and act on sks it includes.				
* Validate personal info	Country United States		•	✓ Attachments No data to display	

Using an activity guide, you can:

- Expand a milestone to see and act on the tasks it includes.
- View the status and percent completion of the overall business process in which you participate.

4.1.1 Why Use Activity Guides

As in the example in Figure 4–1, an activity guide can help organize the tasks involved in an onboarding process. The activity guide moves the user through the correct sequence of steps in the process. In this example, one milestone is *Personal Info*, which includes the task *Validate personal info*.

In an activity guide, users can act on only those tasks presently needing their attention in the current milestone. After they have completed all their tasks for that milestone, they can act on the tasks for the next milestone. They cannot act on tasks in future milestones or on those assigned to other users.

4.2 Navigating Activity Guides

When you select the Activity Guides tab, the tab page displays the Activity Guide List.

The upper section of the Activity Guide List shows a process list displaying all the processes currently moving through the system.

- To get the latest updates to these processes, click the refresh Activity Guide process list icon.
- To filter the data for the Assignee, select from the following:
 - My: Retrieves all activity guide process instances that have active tasks assigned to the logged-in user and those that report to that user.

- Reportees: Retrieves all activity guide process instances that have active tasks assigned to those that report to the logged-in user
- Previous: All activity guides processes instances whose tasks have been completed by the logged-in user
- To filter the data according to the status of the processes, select one of the following:
 - In Progress
 - Completed
 - Errored

Table 4–1 lists and describes the columns in the process list displayed in the Activity Guide List subtab.

Table 4–1 Process List

Column	Description	
Title	Two possible values:	
	Combination of Process Name and Instance Number	
	 Input provided by the end user when instantiating the process 	
Status	Current status of the activity guide process	
Initiated	Date the process is initiated	
Instance #	A number assigned to the activity guide process when the process gets instantiated from the Oracle Fusion Middleware Control	
Process Name	Title provided by the process designer in the BPMN editor while modeling the Activity Guides Process	
Composite Name	Oracle BPM composite name with revision number	
Last Updated	Most recent date the activity guide was updated	

To display a process activity guide in the lower section of the page, select the process from the process list. To display a process activity guide in a separate tab page, right-click the process in the process list and select **Open in separate tab**. This separate tab page refreshes to whatever process you later select from the process list.

The left section of the activity guide displays the guide's navigation panel described in Table 4–2.

ElementDescriptionActivity Guide RefreshRefreshes the navigation panel if any action is performed in the
current activity guide process outside of Process Workspace.Activity Guide Completion
progress barShows the percentage of completion of the current activity guide
process.Completed TasksNumber of tasks the currently logged-in user has completed out
of the total number tasks in the current activity guide process.

 Table 4–2
 Activity Guide Navigation Panel

Element	Description			
Task Filter	For the current activity guide process, specifies the type of task to display in the task details pane. Options:			
	 All 			
	Required			
	 In Progress 			
	Completed			
	 Optional (Tasks that are not mandatory in the process, but that still require a user to explicitly skip) 			

Table 4–2 (Cont.) Activity Guide Navigation Panel

The right section of the activity guide displays the details of the selected node in the activity guide process tree.

For information about the Task Details page, see Section 1.3.2.2, "Using the Process Workspace Task Details Page".

4.3 Showing or Hiding the Activity Guide Tab Page

If you are an administrator, you can specify whether to show or hide the Activity Guide tab page. If you choose to show it, you can make it the default tab when a user logs in.

For instructions on how to show or hide the Activity Guide tab, and to make it the default, see Section 9.2.10, "How to Specify, Show, or Hide the Default Tab Pages in Process Workspace".

Creating and Editing Custom Pages in Process Workspace

This chapter describes how to customize the default layout in Oracle Business Process Management Workspace, and also how process designers and administrators can create new layouts and associate them with different sets of roles. This enables them to define the experience for the various kinds of users of each project.

A layout is a set of pages. It contains information about the pages it contains and about the panels each page contains.

Process Workspace provides its own default layout, but you can customize that layout by adding new pages and panels.

You can choose which pages to hide or show by selecting from the **Hide Tabs** list in the toolbar.

Depending on your roles, you may see several layouts at the same time.

This chapter contains these topics:

- Section 5.1, "Creating Custom Pages in Process Workspace"
- Section 5.2, "Establishing Communication Between Panels in Process Workspace"
- Section 5.3, "Editing Custom Pages and Panels in Process Workspace"

For information about troubleshooting human workflow issues, see Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite, "Human Workflow Troubleshooting".

5.1 Creating Custom Pages in Process Workspace

You can design and add new pages and panels to customize your layout. This section contains these topics:

- Section 5.1.1, "How to Create a Custom Page in Process Workspace"
- Section 5.1.3, "How to Add an Applications Panel to a Customized Page in Process Workspace"
- Section 5.1.4, "How to Add a Dashboards Panel to a Customized Page in Process Workspace"
- Section 5.1.5, "How to Add a Process Instance Panel to a Customized Page in Process Workspace"
- Section 5.1.6, "How to Grant and Manage Access to Customized Pages in Process Workspace"

5.1.1 How to Create a Custom Page in Process Workspace

To create a custom page:

- 1. In the toolbar at the top center of Process Workspace, select More, then select Add Custom Page. The new page appears.
- **2.** In the upper right corner of the new page, click **Add Panel**. The Add Panel dialog box appears with the Applications item selected, as shown in Figure 5–1.

Figure 5–1 Adding an Applications Panel

Applications	Panel Description
Dashboard (Charts)	List of applications, global actions available to
Process Instances	users, that usually start a process
	Preview Image
	×
	Create BPM Test
	Create OOW Demo Instance

You can choose the type of application panel to add: application, dashboard, or process instance.

5.1.2 How to Add an External URL to a Customized Page in Process Workspace

You can set your customized page to open an external web page such as cnn.com or oracle.com. To do this, in the newly created page, select **Show External URL** and add the URL.

Note: Do not enter the external URL of a website that has frame bursting enabled. To include such websites, add them as external applications as explained in Section 9.2.8, "How to Enable Customized Applications and Links in Process Workspace."

5.1.3 How to Add an Applications Panel to a Customized Page in Process Workspace

To add an Applications panel to a customized page:

- 1. In the toolbar at the top center of Process Workspace, select More, then select Add Custom Page. The new page appears.
- **2.** In the upper right corner of the new page, click **Add Panel**. The Add Panel dialog box appears with the Applications item selected.
- **3.** From the Add Panel dialog box, select Applications, then click **Ok**. The Applications panel appears on the page.
- 4. Specify the name of the Applications panel and any configurable properties.
- 5. Click Apply.

5.1.4 How to Add a Dashboards Panel to a Customized Page in Process Workspace

From the Add Panel dialog box, you can also select Dashboards (Charts), as shown in Figure 5–2.
Applications	Panel Description	
Dashboard (Charts)	Graphical display of historical information from Cubes	
Process Instances	Preview Image	
	jstein	
	jcooper	
	wfaulk	
	wchake	

Figure 5–2 Adding a Dashboards (Charts) Panel

This dialog box enables you to include either charts or a chart selector in your new page. Choosing either one provides you with the corresponding preview image.

To add a Dashboards panel to a customized page:

- In the toolbar at the top center of Process Workspace, select More, then select Add Custom Page. The new page appears.
- **2.** In the upper right corner of the new page, click **Add Panel**. The Add Panel dialog box appears with the Applications item selected.
- **3.** From the Add Panel dialog box, select Dashboards (Charts), then click **Ok**. The Dashboards panel appears on the page.
- 4. In the **Panel label** section, specify the name of the Dashboards panel.
- **5.** In the **Panel properties** section, from the **Data Source** list, select a data source for this dashboard panel. The default options are:
 - Workload per Process
 - Performance per Process
 - Workload per Participant
 - Performance per Participant

You can also specify a custom data source. To do this:

- **a.** In the Dashboards panel, in the Panel Properties section, to the right of the **Data Source** field, click the spy-glass icon. The Data Sources dialog box appears.
- **b.** In the navigator pane on the left, click **New**. A Data Source dialog box appears for you to create a new data source.
- **c.** Specify the fields as described in Section 6.2.1, "Specifying Graph Content."
- d. When you are finished specifying the data source, click Ok.
- **6.** Specify the action to be triggered when clicking the graph or widget. Options are:
 - Update another widget: Updates any widget configured to listen to this widget
 - Update a view: Updates any view configured to listen to this widget
- Specify any communication that is to take place with other panels. For information, see Section 5.2, "Establishing Communication Between Panels in Process Workspace" on page 5-6.

5.1.5 How to Add a Process Instance Panel to a Customized Page in Process Workspace

From the Add Panel dialog box, you can select Process Instances, as shown in Figure 5–3.

Figure 5–3 Adding a Process Instances Panel

ld Panel	2
Applications	Panel Description
Dashboard (Charts)	Preview Image
Process Instances	
List of Instances	×
🔵 Instance Details	Actions 🗢 Views 🕶 🚱
🔵 Views (List filters)	Description Due Date
Chart of Instances	Instance #21 of Busii
	Testance #20 of End"

Selecting this option enables you to include one of the following types of process instance panels in your new page: list of instances, instance details, views (list filters), and chart instances. Choosing any one of these displays the corresponding preview image.

To add a Process Instance panel:

- 1. In the toolbar at the top center of Process Workspace, select More, then select Add Custom Page. The new page appears.
- **2.** In the upper right corner of the new page, click **Add Panel**. The Add Panel dialog box appears with the Applications item selected.
- **3.** From the Add Panel dialog box, select Process Instances, select the type of Process Instances panel, then click **Ok**. The type of Process Instances panel you selected appears on the page.
- 4. In the Panel label section, specify the name of the Process Instance panel.
- **5.** If the **Panel properties** section appears for the type of panel you selected, set the properties as required.
- **6.** Specify any communication that is to take place with other panels. For information, see Section 5.2, "Establishing Communication Between Panels in Process Workspace."

5.1.6 How to Grant and Manage Access to Customized Pages in Process Workspace

An administrator can specify the roles that can access a custom page that the administrator has created. Non-administrative users cannot grant access to custom pages.

To specify access to a custom page:

1. In the custom page you are adding, next to the Access field, click Add Access Properties.

Figure 5–4 Specifying Access to a Custom Page

ORACLE' Business Process Workspace	Tasks Process Tracking Standard Dashboards More +	🝓 Home Administration Preferences Help Logout weblogic 🤤
Name* My Custom Page Show External URL Access	User Layout 💌 🔮 🖉 🛞	Add Panel Apply 🗙
⊻My Applications [Applications]		* • + • @/×
Show - Ro Applications available.		

The Add Access Properties dialog appears.

- 2. In the Add Access Properties dialog box:
 - Enter an identifier and a name for this page layout.
 - From the Roles list on the left, select the roles to which you are granting access in this layout. Move them to the list on the right by clicking the arrow icons, then click **OK**.

Note:

- Only an administrator can grant access to custom pages that the administrator has created. Non-administrative users cannot grant access to custom pages.
- To grant access to a role, you yourself must be associated with that role, otherwise it will not appear in the Role list.

The layout you have just created appears in the Access list. That layout now specifies the roles with access to this custom page.

To edit access properties for a custom page:

- 1. In the custom page you are adding, next to the Access field, click **Edit Access Properties**. The Edit Access Properties dialog box appears.
- **2.** Enter your changes, then click **OK**. The layout as you have just edited it appears in the Access list. It now specifies the roles with access to this custom page.

Note: You can add new roles to a layout without removing previously granted roles.

To delete access properties for a custom page:

In the custom page you are adding, next to the Access field, click **Delete Access Properties**.

If the layout is a custom one, then you are prompted to confirm the deletion.

To maintain the custom page for the current user when a custom layout is deleted, the user must first save the page to the default layout. Only then should the user create the custom layout, modify the page, and save it to the new layout. If the user does not first save the page to the default layout, then the page is deleted for all members of the roles defined in the custom layout, including the current user.

If the user had right away created the custom layout and set it on the page before saving the page, then

Any custom pages previously associated with this custom layout remain, but are now associated with the default layout.

If the custom page uses the default layout, then the custom page is deleted. Users previously authorized to see this custom page will no longer see it.

5.2 Establishing Communication Between Panels in Process Workspace

You can establish communication between two panels so that an action in one panel is triggered by an action in the other. For example, you can configure a Process Instance panel to listen for actions that occur in a View List panel. Then, when a user selects a given view in the View List panel, that action triggers the use of that view in the Process Instance panel.

5.2.1 How to Establish Communication Between Panels in Process Workspace

To establish communication between panels, you configure one panel, designed with listening capabilities, to monitor actions in one or more source panels. An action occurring in a source panel then triggers an action in the listening panel.

Note: Communication can be established only between panels on the same page.

To establish communication between panels:

- 1. In the custom page, click Edit Mode. The entire page is available for editing.
- 2. Create a source panel and select the appropriate option from the Data Source list.
- **3.** Create a listening panel. This time, do not select anything from the **Data Source** list.
- **4.** Expand **Communication with other panels**, and, in the **Selected event sources** window, select the source panel created in step 2.
- **5.** In each panel, click **Apply**.

After you have done this, any action in the source panel triggers a corresponding action in the listening panel. For example, when two dashboard panels are communicating, clicking any graphical bars in the source panel triggers the use of that view in the listening panel.

5.3 Editing Custom Pages and Panels in Process Workspace

The custom page has three sections:

- Panel Label, which has a text field for specifying a name for the page
- Panel Properties, which has a field labeled Data Source, in which you can enter all
 or part of the name of a data source. You can also click the adjacent Browse button
 to search for a data source. This section also has a field labeled When clicking on
 the widget. It has an adjacent list from which you select the action you want
 triggered when clicking a widget.
- Communication with Other Panels described in Section 5.2.1, "How to Establish Communication Between Panels in Process Workspace"

This editable display of the panel enables you to define the name and properties for the new page and establish communication with other panels. In addition, the icons in the upper right corner of the new page enable you to:

- Switch to View Mode—that is, the manner in which the page is ordinarily viewed. After you switch to View Mode, you can switch back to Edit Mode.
- Move the panel up, right, left, or down.
- Add another panel to the right, left, above, or below this one.
- Remove this panel.

Using Dashboards in Process Workspace

This chapter describes how users of Oracle Business Process Management Workspace can use and customize standard dashboards—that is, the graphs and drill-downs based on data previously computed in a process cube schema. Dashboards reflect the standard metrics gathered during the execution of a process.

This chapter contains these topics:

- Section 6.1, "Using Standard Dashboards"
- Section 6.2, "Creating Custom Dashboards"
- Section 6.3, "Understanding Data Visibility Rules in Dashboards"

For information about troubleshooting human workflow issues, see Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite, "Human Workflow Troubleshooting".

6.1 Using Standard Dashboards

Standard dashboards are the graphs and drill-downs based on data previously computed in a process cube schema. They reflect the standard metrics gathered during the execution of a process.

In each of the standard dashboards, clicking a bar graph in the left pane causes data to be displayed graphically in the right pane. Clicking an item in the right pane displays certain data in the Work Items panel in the lower pane. This section explains how each type of dashboard operates.

You can customize the presentation of the Work Items panel in the lower pane. For information on how to do this, see Section 3.6, "Customizing the Presentation of Process Instances in Process Workspace".

This section contains these topics:

- Section 6.1.1, "Participant Dashboards"
- Section 6.1.2, "Process Dashboards"

6.1.1 Participant Dashboards

Two types of standard participant dashboards enable you to analyze:

- The workload of participants
- The performance of participants

6.1.1.1 Participant Workload Dashboard

Figure 6–1 shows an example of a participant workload dashboard.



Figure 6–1 Participant Workload Dashboard

The left panel of a participant workload dashboard displays the total number of active instances waiting for completion for each participant across all processes. When you drill down into the chart for a specific participant, the right panel displays the total number of active instances waiting for completion by process for either the selected participant or the selected role. The last drill-down displays the active instances waiting for completion for either the selected participant or the selected role and process.

Note: In this release, participant information in a workload dashboard is for the physical role corresponding to the swimlane for the task. When the swimlane role is not defined, the corresponding participant information is null.

6.1.1.2 Participant Performance Dashboard

Figure 6–2 shows a participant performance dashboard.



Figure 6–2 Participant Performance Dashboard

The left panel of a participant performance dashboard displays the average time taken by participants across all processes. When you drill down into the chart for a specific participant, the right panel displays the average time taken to complete instances by process for the selected participant. The last drill-down displays the list of completed instances for either the selected participant or the selected role and process.

Note: In this release, for processes that have user tasks with multiple users, user information is captured only for the last user that updated the task.

6.1.2 Process Dashboards

Two types of standard process dashboards enable you to analyze:

- The number of process instances waiting for completion
- The average time taken per process

6.1.2.1 Process Workload Dashboard

Figure 6–3 shows a process workload dashboard.



Figure 6–3 Process Workload Dashboard

The left panel in the process workload dashboard displays the total number of active instances waiting for completion per process. When you drill down into a specific process, the right panel displays the total number of active instances waiting for completion for activities in the selected process. The last drill-down displays the active instances waiting for completion for the selected activity and process.

6.1.2.2 Process Performance Dashboard

The left panel in the process performance dashboard displays the average time taken per process. When you drill down into a specific process, the right panel displays the average time taken to complete by activities in the selected process. The last drill-down displays the completed instances for the selected activity and process.

Figure 6–4 shows a process performance dashboard.



Figure 6–4 Process Performance Dashboard

6.2 Creating Custom Dashboards

As with standard dashboards, custom dashboards are the graphs and drill-downs based on data previously computed in a process cube schema. They reflect the standard and user-defined metrics gathered during the execution of the process. They enable you to define new graphs using both standard metrics and user-defined metrics specified by using business indicators.

For instructions about adding a custom dashboard, see Section 5.1.1, "How to Create a Custom Page in Process Workspace" on page 5-2.

6.2.1 Specifying Graph Content

Graph content is specified by defining a data source.

For information about invoking the data source dialog box, see Section 5.1.4, "How to Add a Dashboards Panel to a Customized Page in Process Workspace."

Specify the following data source information:

- Name: Provide a data source identifier
- Title: Provide the title used while displaying the data source
- Graph Type: Select one of the following list of graph types:
 - Area
 - Percent Area
 - Stacked Area
 - Horizontal Bar
 - Horizontal Percent Bar
 - Horizontal Stacked Bar
 - Bar

- Percent Bar
- Stacked Bar
- Line
- Percent Line
- Stacked Line
- Pie
- Pie Bar
- Multiple Pie
- Process: Select the process from the list. All business indicators associated with the selected process become available for the data source definition.
- Data Type: Select one of the following supported data types:
 - Workload
 - Activity and Measurement Sampling
 - Process Level Sampling

Dimensions

Series: Specify the dimension. Series variables categorize the data. The series variable values are displayed in the legend area labels in the chart.

Use Range: Select to use ranges for dimensions with ranges.

• **Group**: Specify the dimension used to group the dimensions used in a series. Group variables also categorize the data. The group variable values are displayed as labels along one of the axis in the chart.

Use Range: Select to use ranges for dimensions with ranges.

Note: Group and Series variables can be used together to perform nested categorization of chart data, in which case, series values are displayed under each group. For example, Total revenue by customer (series) and by product (group).

 Drilldown to Series: Select a drill down series variable which is propagated to a listening dashboard panel. As a result, a drill down dashboard is displayed in the listening dashboard panel.

For more information, see Section 5.2, "Establishing Communication Between Panels in Process Workspace".

Use Range: Select to use ranges for dimensions with ranges.

Table 6–1 summarizes the standard dimensions available based on the Data Type selection. Additionally, all the process specific dimensions are available.

	Workload	Activity and Measurement Sampling	Process Sampling
Dimension: Series	Revision Role Activities Participant	Revision Role Activities Participant Completion Date	Revision Completion Date
Dimension: Group	Participant	Participant Completion Date	Completion Date

 Table 6–1
 Standard Dimensions Available

Measure

Variable: Specify the measurement

Table 6–2 summarizes list of standard measurements available based on the Data Type selection. Additionally, all the process-specific measurements are available.

Table 6–2 Standard Measurements Available

	Workload	Activity and Measurement Sampling	Process Sampling
Variable	Instances Task Avg. Time Task Median Process Avg. Time Process Median Time	Task Time	Task Time

 Operation Type: Specify the aggregation functions to be applied on the measurement variables selected.

Table 6–3 summarizes the operation types that can be used with various Data Type selections.

Table 6–3	Operation	Types Allowed
-----------	-----------	---------------

	Workload	Activity and Measurement Sampling	Process Sampling
Operation Type	Avg Sum	Avg Sum Min Max	Avg Sum Min Max

 Value Type: Select one of the following values: Value, Day(s), Hours, Minutes, Seconds, Percentage

Note: If a new process is deployed while you have a dashboards page open, the new process will not be listed in the Data Sources dialog box until you log out of Process Workspace and then log back in.

Filters

You can add filter conditions based on the measure variables, series, and groups. You can add the filter condition only if a process is selected from the Process list. If "All" is selected in the Process list, then the Filters section is disabled.

Note: If you add a filter for business indicators, that filter filters out the values initially based on the condition given. Aggregation—either the sum or the average—then applies to the resulting set of items.

For example, if the user tries to compute the average of all business indicator values greater than 1000, the user will set a filter for this—for example: BI <Greater or equals> 1000 and set Operation type as "avg". The result is that the business indicator values greater than 1000 are collected first, and then the computation for the average calculation takes place.

6.3 Understanding Data Visibility Rules in Dashboards

This section discusses data visibility rules applicable to dashboards in Process Workspace.

This section includes the following topics:

- Section 6.3.1, "Workload Sampling Data"
- Section 6.3.2, "Activity and Measurement Sampling Data"
- Section 6.3.3, "Process Level Sampling Data"

6.3.1 Workload Sampling Data

The section discusses the restrictions applicable to workload sampling data and the application roles a user must have in order to view certain activities in Process Workspace. The restrictions are:

- The workload for user task activities is seen by the member of the corresponding swimlane application role, process owner application role or bpmprocessadmin application role. Workloads are not tracked based individual user tasks participants.
- The workload for non user task activities is seen by the corresponding process
 owner application role members or bpmprocessadmin application role members.
- If a process is associated with an organization unit, the corresponding workload sampling data is restricted to organization unit members or bpmprocessadmin application role members.

6.3.2 Activity and Measurement Sampling Data

This section discusses the restrictions applicable to activity and measurement sampling data and the application roles a user must have in order to view certain activities in Process Workspace. The restrictions are:

- User task activities are seen by the last participant who acted on the user task or the corresponding process owner application role members or bpmprocessadmin application role members.
- Non user task activities are seen by the corresponding process owner application role members or bpmprocessadmin application role members.
- If a process is associated with an organization unit, the corresponding activity and measurement sampling data is restricted to organization unit members or bpmprocessadmin application role members.

6.3.3 Process Level Sampling Data

This section discusses the restrictions applicable to process level sampling data and the application roles a user must have in order to view certain activities in Process Workspace. The restrictions are:

- If a process is associated with an organization unit, the process data is visible to the organizational unit members or bpmprocessadmin application role members.
- If a process is not associated with an organization unit, it is visible to all users.

7

Managing Workloads Among Users in Process Workspace

This chapter describes how supervisors can use Oracle Business Process Management Workspace to view and update all tasks assigned to the users who report to them.

This chapter contains these topics:

- Section 7.1, "Managing the Availability of Users in Process Workspace"
- Section 7.2, "Distributing Work among Users in a Group"
- Section 7.3, "Setting Rules"
- Section 7.4, "Setting Assignment Rules for Tasks with Multiple Assignees"
- Section 7.5, "Reassigning or Delegating Tasks in Process Workspace"
- Section 7.6, "Using Reports to Manage Workflows in Process Workspace"
- Section 7.7, "Using Dashboards to Manage Workflows in Process Workspace"
- Section 7.8, "Handling Escalations in Process Workspace"

For information about troubleshooting human workflow issues, see Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite, "Human Workflow Troubleshooting".

7.1 Managing the Availability of Users in Process Workspace

If a user is unavailable for a period of time and must delegate some or all assigned tasks to another user, then the user rules can automatically reassign the tasks based on certain conditions. More specifically, if a user is going on vacation, a vacation period can be associated with a rule to create a vacation rule.

For more information about creating user rules, see Section 2.6.1, "How to Create User Rules in Process Workspace."

For more information about setting a vacation rule, see Section 2.6.3, "How to Enable a Vacation Period in Process Workspace.".

7.2 Distributing Work among Users in a Group

To distribute work among users in a group, you create a rule for that group. Creating a group rule is similar to creating a user rule, with the addition of a list of the groups that you, as the logged-in user, manage.

For information about creating user rules, see Section 2.6.1, "How to Create User Rules in Process Workspace."

Examples of group rules include:

- Assigning tasks from a particular customer to a member of the group
- Ensuring an even distribution of task assignments to members of a group by using round-robin assignment
- Ensuring that high-priority tasks are routed to the least busy member of a group

For information about creating a group rule, see Section 2.6.2, "How to Create Group Rules in Process Workspace".

7.3 Setting Rules

Rules act on tasks. These tasks can be either of a specific task type or they can be all of the tasks assigned to a user or group. Figure 7–1 shows where you set rules, including vacation rules (different from the vacation period settings described in Section 2.6.3, "How to Enable a Vacation Period in Process Workspace").

Figure 7–1 Creating a Rule

URACLE Business	Process Workspace	Va Home Administration Preferences Help Logout Weblogic >
Preferences	My Rules Other Rules	
User Preferences Certificates Rules Notification Preferences Accessibility	✓ Rules ✓ △ ⊕ ‰ My Rule ✓ Vacation Period (Disabled) Name * User Rule ✓ My Rules User Rule ● User Rule Use as vacation rule ■ Execute rule only between these dates: Start Date ■ End Date ■ TASKS Tasks matching these conditions	<u>Save</u> <u>Revert</u>
		Match 🔿 All 🕙 Any 🖶
	ACTION Reassign to: Delegate to: Set outcome to: Bake no action	

A rule cannot always apply in all circumstances in which it is used. For example, if a rule applies to multiple task types, it may not be possible to set the outcome for all

rule applies to multiple task types, it may not be possible to set the outcome for all tasks, since different tasks can have different outcomes.

Rules are executed in the order in which they are listed. Rules can be reordered by using the up and down buttons in the header, as shown in Figure 7–1.

If a rule meets its filter conditions, then it is executed and no other rules are evaluated. For your rule to execute, you must be the only user assigned to that task. If the task is assigned to multiple users (including you), the rule does not execute.

You cannot specify business rules for ToDo tasks

7.3.1 How To Create User Rules

Specify the following when creating a user rule:

Rule name

- If the rule is a vacation rule. See Section 2.6.3, "How to Enable a Vacation Period in Process Workspace," for how to set the vacation period that is synchronized with the vacation rule.
- Which task or task type the rule applies to—If unspecified, then the rule applies to all tasks. If a task type is specified, then any attributes mapped for that task type can be used in the rule condition.
- When the rule applies
- Conditions on the rule—These are filters that further define the rule, such as specifying that a rule acts on priority 1 tasks only, or that a rule acts on tasks created by a specific user. The conditions can be based on standard task attributes and any mapped attributes that have been mapped for the specific tasks. See Section 9.4.1, "How to Specify Mapped Attributes," for more information.

User rules do the following actions:

- **Reassign to**—You can reassign tasks to subordinates or groups you manage.
- Delegate to—You can delegate to any user or group. Any access rights or privileges for completing the task are determined according to the original user who delegated the task. (Any subsequent delegations or re-assignments do not change this from the original delegating user.)
- Set outcome to—You can specify an automatic outcome if the workflow task was designed for those outcomes, for example, accepting or rejecting the task. The rule must be for a specific task type. If a rule is for all task types, then this option is not displayed.
- Take no action—Use this action to prevent other more general rules from applying. For example, to reassign all your tasks to another user while you are on vacation, except for loan requests, for which you want no action taken, then create two rules. The first rule specifies that no action is taken for loan requests; the second rule specifies that all tasks are reassigned to another user. The first rule prevents reassignment for loan requests.

To create a user rule:

1. Click the **Preference**s link

The Preferences pane is displayed.

- **2.** In the Preferences pane, click **Rules**. The corresponding pane appears in the middle of the screen.
- 3. Click My Rules and then click Add. The My Rule pane appears on the right.
- 4. In the My Rule pane, do the following and click **Save**:
 - Provide a name for the rule.
 - Select Use as a vacation rule if you are creating a vacation rule. The start and end dates of the rule are automatically synchronized with the vacation period.
 - Browse for task types to which the rule applies.
 - Select Execute rule only between these dates and provide rule execution dates.
 - In the **IF** area, add rule conditions.
 - In the THEN area, select actions to be taken: Reassign to, Delegate to, Set outcome to, or Take no action), as shown in Figure 7–2.

The new rule appears under the **My Rules** node.

7.3.2 How To Create Group Rules

Creating a group rule is similar to creating a user rule, with the addition of a list of the groups that you (as the logged-in user) manage. Examples of group rules include:

- Assigning tasks from a particular customer to a member of the group
- Ensuring an even distribution of task assignments to members of a group by using round-robin assignment
- Ensuring that high-priority tasks are routed to the least busy member of a group

Group rules do the following actions:

- Assign to member via—You can specify a criterion to determine which member of the group gets the assignment. This dynamic assignment criterion can include round-robin assignment, assignment to the least busy group member, or assignment to the most productive group member. You can also add your custom functions for allocating tasks to users in a group.
- Assign to—As with user rules, you can assign tasks to subordinates or groups you directly manage.
- Take no action—As with user rules, you can create a rule with a condition that prevents a more generic rule from being executed.

To create a group rule:

- 1. Click the **Preferences** link. The Preferences pane is displayed.
- **2.** In the Preferences pane, click **Rules**. The corresponding pane appears in the middle of the screen.
- 3. Click the Other Rules tab.
- 4. Select Group from the list.
- 5. Enter a group name and click the **Search** icon, or enter a group name.

The Identity Browser opens for you to find and select a group.

6. Select the group name under the **Group Rules** node and click **Add**, as shown in Figure 7–2.

ORACLE' Business Pr	ocess Workspace	🍓 Home Administration Preferences Help Logout weblogic 드
Preferences	My Rules Other Rules	
User Preferences Certificates Rulies Notification Preferences Accessibility	✓ Rules ✓ △ My Group Mes ✓ ✓ Group Rules ✓ ✓ Group Rules ✓	
Converight @ 2005, 2012, Oracle and/or its	affiliatar. All rights received	T Delegated task access is determined according to rights of original user who delegates.

Figure 7–2 Creating a Group Rule

- 7. Provide group rule information and click **Save**.
 - Provide a name for the rule.
 - Browse for task types to which the rule applies.
 - Provide rule execution dates.
 - In the IF area, add rule conditions.
 - In the THEN area, select the actions to be taken (or none) (Assign to member via, Assign to, or Take no action), as shown in Figure 7–2.

The new rule appears under the **Group Rules** node.

7.4 Setting Assignment Rules for Tasks with Multiple Assignees

If a task has multiple assignees, then assignment rules are not evaluated for the task, and the task is not automatically routed. This is because each of the task's assignees can define assignment rules, which can potentially provide conflicting actions to take on the task. Only tasks that are assigned exclusively to a single user are routed by the assignment rules.

For example, consider the following sequence:

- 1. A rule is created for user cdickens to reassign all assigned requests to user jstein.
- **2.** User jcooper reassigns the allocated tasks to cdickens and cdoyle.
- **3.** Cdickens claims the task, and the task appears in their inbox.

The task is not automatically reassigned to jstein. The task is routed to jstein, following the assignment rule set for cdickens, only if user jcooper explicitly re-assigns the task to cdickens alone. instead of reassigning the task to multiple users (cdickens and cdoyle).

7.5 Reassigning or Delegating Tasks in Process Workspace

A supervisor can reassign or delegate tasks to reportees. A user with BPMWorkflowReassign privileges can delegate a task to anyone.

To reassign or delegate a task:

1. From the Task Actions list, select Reassign, as shown in Figure 7–3.

Figure 7–3 Reassigning a Task



2. Select Reassign or Delegate.

Delegate differs from **Reassign** in that the privileges of the delegate are based on the delegator's privileges. This function can be used by managers' assistants, for example.

3. Provide or browse for a user or group name, as shown in Figure 7–4.

Figure 7–4 Reassigning a Task

assign tasks to one or more is will transfer ownership of t ● Reassign (transfer task to ● Delegate (allow specified u	isers. Le task and remove another user or gro ser to act on my be	it from your worl up) half)	klist.				
💌 j*			Search	Reset			
ID	First Name	Last Name	Email	Title	Manager	Organization	
] 👌 jausten	Jane	Austen	jausten@emailExample.com	Loan Consultant	fkafka	Demo User	
🖸 🐧 jcooper	James	Cooper	jcooper@emailExample.com	Loan Agent 1	jstein	Demo User	
] 🌡 jlondon	Jack	London	jlondon@emailExample.com	Loan Agent 1	sfitzger	Demo User	
] 🍦 jstein	John	Steinbeck	jstein@emailExample.com	Manager2	wfaulk	Demo User	
] 👌 jverne	Jules	Verne	JVerne@emailExample.com	CIO	cdickens	Demo User	
cted 🎖 jverne 🗱	👌 jcooper 🗱 🏼 🐇	🕽 jstein 🗱					

A supervisor can always reassign tasks to any of his reportees.

4. Move names to the Selected area and click OK.

You can reassign to multiple users or groups. One of the assignees must claim the task, as shown in Figure 7–5.

2					🗷 💠 🔿	Complete	4
A Details							
Assignees weblogic John Steinbeck James Cooper	Acquired By		Task Number 20	0020			
Creator weblogic	Category		Priority 1	•			
Created Aug 21, 2012 8:33 AM	Percentage g	0.0	State As	igned			
Expiration Date	Due Date	1					
	Start Date						
	o la coole						
History							
3 weblogic, John Steinbeck, James Coope	er -						
1 Assigned Aug 21, 2012 8:33 AM							
weblogic							
-							
8=							
······							
& weblogic, John Ste							
C neoroger series certi							
<u> </u>							
±							
		<u> </u>	Attachments			4	2
Comments			Lindated By	Date Updated			
data to display			opulated by	Date opdated			
Comments data to display			No data to display	pare oposted			_
Comments data to display			No data to display	bute oposice			
Comments data to display			No data to display	pare opused			_
Comments data to display			No data to display	pore oponeo			

Figure 7–5 Claiming a Task

Note: When task details have been upgraded from an earlier release, you can see a "Request Failed" error when executing the **Reassign** action. Actually, the reassign completes, and when you click OK again, a popup says the task is already assigned.

To eliminate the error message, upgrade your taskflow applications by opening them in Oracle JDeveloper, then redeploy the task form.

7.6 Using Reports to Manage Workflows in Process Workspace

Reports are available on the Tasks page in the Reports panel. Report results cannot be saved.

Figure 7–6 shows the Reports panel displaying the standard report types available.

Figure 7–6 Oracle Business Process Management Workspace Reports



For more information about reports, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite.*

7.7 Using Dashboards to Manage Workflows in Process Workspace

Dashboards provide a variety of analytical information about the workload on users for the different processes being executed. For example, if a specific user has many tasks assigned to her, and she is making little progress, this would indicate that all those corresponding processes are blocked. Having this information enables you to reassign those tasks to someone else and, if necessary some user or group routing rules can be specified to manage the issue in the short term.

For more information about dashboards, see Chapter 6, "Using Dashboards in Process Workspace".

7.8 Handling Escalations in Process Workspace

Escalations are among the system actions you can perform from the Task Details page.

For more information about the Task Details page, see Section 1.3.2.2, "Using the Process Workspace Task Details Page".

If a user escalates a task, it is automatically assigned to whomever is specified in the escalation path.

The escalation path is specified by the administrator. If no escalation path is specified, then the task escalates to the user's supervisor who can reassign it from there.

If a task is assigned to a role, but none of the role members have claimed the task, then the task is escalated as specified in the escalation path for that role. If the task is assigned to a role and a role member has claimed the task, then the escalation goes to the role member's supervisor.

For information about how an administrator specifies the escalation path, see Section 10.5.2, "How to Specify the Escalation Path for a Role".

Working with Adaptive Case Management in Process Workspace

Process Workspace supports the processing of cases implemented using Oracle Business Process Management's Adaptive Case Management. This chapter explains how to work with cases in Process Workspace.

This chapter contains these topics:

- Section 8.1, "About Adaptive Case Management"
- Section 8.2, "Navigating Cases"
- Section 8.3, "Viewing Case Information"
- Section 8.4, "Working on Cases"

8.1 About Adaptive Case Management

Adaptive Case Management supports the modeling and running of very flexible and data-intensive business processes.

A case is a business process that is not rigidly predefined in the way a typical Business Process Management process is. Cases are less structured ad-hoc processes that include information, tasks, rules and services. The tasks you perform while working on a case change as information and activities related to the case are processed. Decisions you make while managing a case can change which processes and tasks are executed.

Some examples of cases that are suitable to be modeled and run using Adaptive Case Management include:

- Insurance claims
- Incident investigations
- Customer complaints

For additional information on Adaptive Case Management, see Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management.

8.2 Navigating Cases

Cases are accessed through the **Case** page in the Process Workspace GUI. To access the Case page, click the Case link at the top of the GUI page.

8.2.1 About the Case Page

The Case page displays information for the currently selected case.

Use the Case page to:

- View a list of cases and select one to view or work on.
- View case details for open or closed cases.
- Add comments or raise events on a case.
- View case activities and their details, create new case activities, and filter the display of case activities by their status.
- View an Audit Trail of case events. Filter the display of case events by event type.
 View and add new documents to a case. View and modify case data.
- View or add documents to the case.
- View case milestones and mark them completed.
- View case stakeholders. Add stakeholders. Add or edit users attached to stakeholders.
- View user information for case details.
- Close cases.
- Reopen cases.

8.2.2 About the Navigation Bar

Use the Navigation bar (shown in Figure 8–1) to identify the current case, to move between cases, to close and reopen cases, and to get information about cases.

Figure 8–1 Navigation Bar

\equiv - \circ [1] Complaint Case for Daniel Duan ()	Add Comment Raise Event	🔞 🗘 다 Close Case

Element	Description
Case Listing	Click the down arrow on the Case Listing control to see a list of current cases. Click on a case to navigate to it.
Status Icon	The Status icon indicates the case's status. Hover over the icon to see a text display of the case's status, for example, ACTIVE.
Case Number and Name	The case number and name are displayed in the Navigation bar.

The Navigation bar provides the following elements:

Element	Description
Information Icon	Click the Information icon to display a pop-up window containing high-level details about the case.
	Fields displayed include:
	■ Title
	 Identification Key
	 Number
	■ Name
	 Priority
	Category
	Creator
	Start Date
	State
	 Document Folder
Add Comment	Click to add a comment to the case. Fields enabling you to add the comment are displayed directly on the Case page, above the Audit Trail panel.
Raise Event	Click to initiate an event for the case. Fields enabling you to add the event are displayed directly on the Case page, above the Audit Trail panel.
	The specific events available to be raised change depending on the composite used to create the case instance.
Refresh	Click the Refresh icon to update the case details displayed in the Case page.
	When you select an action from Actions, a comment window appears. You must enter a comment and click OK to update the task and refresh the task list table. Click Cancel to cancel the action.
Navigation arrows	Click the left and right arrows to navigate to the previous and next case, respectively.
Close Case	Click Close Case to mark the case instance as closed.

8.2.3 Using the Navigation Bar

Use the Navigation bar to move between cases, to close and reopen cases, and to get information about cases. You can also add comments and events to cases.

8.2.3.1 Selecting a Case

Use the controls in the Navigation bar (as well as those in the Case List panel) to select a case to work on. You can select a case by:

- Choosing it from the Case Listing drop-down list
- Choosing it from the Case List panel (visible when the Case Listing is pinned to the Case page)
- Navigating to it using the Previous Case and Next Case buttons (visible when the Case Listing is not pinned to the Case page)

Figure 8–2 Case Listing Drop-down List



See "Viewing Hidden Panels" for more information about pinning and unpinning the Case List panel.

8.2.3.2 Closing a Case

Close cases using the Close Case button on the Navigation bar.

To close a case:

- **1.** Navigate to the case.
- 2. Click Close Case. The Close Case details dialog box is displayed.
- 3. Select an outcome from the Outcome drop-down list.
- 4. Optionally enter a comment.
- **5.** Click **Close Case**. An information dialog box is displayed indicating the results of the close action.
- 6. Click OK.

8.2.3.3 Viewing a Closed Case

You can view closed cases.

To view a closed case:

- 1. If the **Case List** panel is not displayed, pin the Case Listing drop-down list to display it.
- 2. Click Advanced Search. The Advanced Search dialog box is displayed.
- 3. Ensure that Closed is selected in the State section of the search criteria.
- **4.** Click **Search**. The Case List panel is updated with a list of closed cases matching the other search criteria you used in the Advanced Search dialog box.
- **5.** Click the case to view. The Case page is updated with the information for the selected case.

8.2.3.4 Adding a Comment to a Case

Add comments to a case using the Add Comment link in the Navigation bar.

To add a comment to a case:

- 1. Click Add Comment. The comment entry controls appear at the top of the Case page.
- **2.** Enter the text for your comment.

- **3.** Select a permission level from the **Permission** drop-down list. This determines the user permission level needed to be able to view the comment.
- **4.** Click **Post** to add the comment to the case. (Note: to close the comment controls without posting the comment, click **Close**.

Comments can be viewed in the Audit Trail panel. See "Viewing the Audit Trail".

8.2.3.5 Raising an Event on a Case

Raise events on a case using the **Raise Event** link in the Navigation bar. The events that are available to be raised on a case are pre-configured using Oracle Business Process Management Studio JDeveloper.

To raise a comment on a case:

- 1. Click Raise Event. The raise event controls appear at the top of the Case page.
- **2.** Select an event name from the **Event Name** drop-down list to specify the type of event.
- **3.** Enter a comment.
- **4.** Select a permission level from the **Permission** drop-down list. This determines the user permission level needed to be able to view the event.
- **5.** Click **Raise Event** to add the event to the case. (Note: to close the Raise Event controls without posting the event, click **Close**.

Events can be viewed in the Audit Trail panel. See "Viewing the Audit Trail".

8.3 Viewing Case Information

This section describes the panels available on the Case page and the kinds of information displayed on it, how to filter the displayed information, and how to find and view cases.

You interact with cases using the information provided on the Case page, and using the controls and links on the page to perform work related to the case, such as adding comments or events, reviewing and modifying milestones, creating new case activities and so on.

8.3.1 Viewing Case Instances

To view the details for a case, navigate to it using the Navigation bar. Select the case in the Case Listing control, or use the Previous Case and Next Case icons to navigate between cases.

See "Navigating Cases" for more information.

Figure 8–3 shows the Case page, which is divided into the following sections:

- Navigation bar
- Activities
- Details (which can display Audit Trail, Data, or Comments information)
- Milestones
- Stakeholders

Figure 8–3 Case Page

≣ • ○[1] Complair	nt Case for Daniel Duan 🛈	Add Comment Ra	aise Event	₩ 🗢 🔿	Close Case
ctivities All	Audit Trail Data Documents		۱ <u>ال</u>	Milestones	
Available 👘 👻	Press Enter to Search	🔍 All Comments Even	nt 🗸 🦾	Complaint Resolution Completed	Open
Send Email Create Task	Activity started: "Review Complaint Details" Initiating task. 	6 seco	onds ago	Complaint Resolution Started	Open
Review Complaint Details This activity is already running	James Cooper			Initial Assessment Completed	Open
Conduct Initial Assessment	Activity started: "Conduct Initial Assessment" Initial Assessment completed. James Cooper	4 min	nutes ago	Initial Assessment Started Completed 5 minutes ago by workflowsystem Initial assessment started	Completed
Started 4 minutes ago Review Complaint Details Started 6 seconds ago	Milestone completed: "Initial Assessment Started"	" 5 min	nutes ago	Investigation Completed	Open
completed	System			Load More Ite	ems 1-5 of 6 ite
Prepare Complaint Case	Activity completed: "Prepare Complaint Case" System	5 min	nutes ago	Stakeholders	4
	Case started System	4h	nours ago	Analysts	
				Customer Service Representative	
				🏠 Charles Dickens	
				Case Owner	
				🏠 James Cooper	
				🎝 John Steinbeck	
				Investigators	
				🎝 Jane Austen	

8.3.2 Viewing the Case List

You can view (and select) cases in two ways:

- Click the Case Listing drop-down list in the Navigation bar to display a list of cases. (See Figure 8–2.)
- When it is pinned to the Case page, use the Case List panel to view and select cases.

To display the Case List panel, click the Pin icon in the Case Listing window. The Case List panel includes more case details than the Case Listing control. When the Case List panel is pinned to the Case page, the Case Listing control does not appear on the **Navigation** bar.

To hide the Case List panel, click the Pin icon in the upper right-hand corner to un-pin it from the Case page. When the Case List panel is unpinned, the Case Listing control appears on the **Navigation** bar.

8.3.2.1 Viewing Hidden Panels

When the Case List panel is pinned to the page, other panels are hidden. When the Activities panel is hidden, click **Show Activities** to display it. When the Milestones and Stakeholders panels are hidden, click **Show Milestones and Stakeholders** to display them.

8.3.2.2 Case List Panel

Use the Case List panel of the Case page to locate, display and select cases. Cases are displayed in a series of sections in the panel.

Element	Description
Search Criteria	Enter the search text. The text is searched for in the following case elements:
	 Title
	Identification Key
	Case Number
Search	Click the Search icon to search for cases using the Search Criteria .
Advanced Search	Click the Advanced Search control to display the Advanced Search dialog. The controls on the Advanced Search dialog are shown below:
Match	Use the match control to control how the criteria are used in the search.
	Choices include:
	 All: Select to include cases which match all of the search criteria specified in the Advanced Search dialog box.
	 Any: Select to include cases which match any of the search criteria specified in the Advanced Search dialog box.
Search	Click Search to search on the selected criteria. Cases matching the criteria are displayed in the Case List panel.
Title	Case title to search on.
Identificati on Key	Identification Key to search on.
Creator	User that created the case.
Updated by	User(s) that updated the case.
State	Select the case state(s) to search on using the check boxes. Choices include:
	 All: Search on all states listed below.
	Active
	 Closed
	 Expired
	 Aborted
	 Suspended
	 Errored
	■ Stale
Updated	Select an update day to search on. Choices include:
	 None
	 Today
	 Yesterday
	Last Week
Due	Select a due date to search on. Choices include:
	 None
	Overdue
	 In one day
	 In one week

Element	Description
Created	Select a creation date to search on. Choices include:
	■ None
	■ Today
	 Yesterday
	Last Week
Sort By	Sorts the cases displayed in the Case List. Choose from:
	Created Date
	Case Number
	Creator
	Due Date
Load More Items	This link appears when there are more items than will fit in the default panel size. Click to display additional items.

8.3.2.3 Sorting the Case List

Sort the cases displayed in the Case List by selecting a sort criteria from the **Sort By** drop-down list. When you make a selection, the cases are re-displayed in the selected order. Choose from the following sort criteria:

- Created Date
- Case Number
- Creator
- Due Date

8.3.2.4 Searching for Cases

You can search for cases using a simple search, or a more comprehensive advanced search.

To Search For a Case:

- 1. Enter the search criteria in the **Search Criteria** entry box.
- 2. Press Enter or click the Search icon.

To Perform an Advanced Search For a Case:

- 1. Click the Advanced Search arrow. The Advanced Search dialog box appears.
- **2.** Specify your search criteria in the text fields, check boxes, and selection fields in the Advanced Search dialog box as shown in Figure 8–4.
- **3.** Click **Search**. The Case List is updated with the results of your search.

Case List		-#
Desca Falsa la Garada		Constant Data Tar
Press Enter to Search	Sort By	Created Date
Complaint Case for Rosie	Match All	Search
Created 20 minutes ago by Syste		
0		
Complaint Case for John	Title	
Created 21 minutes ago by Syst	Number	
Complaint Case for Danie		
Created 4 hours ago by System	Identification Key	
2	Creator	
	Unders 10	
	Updated By	
	State	
		Active
		Closed
		Expired
		· · · ·
	Updated	Nene
	opulicu	Today
		Yesterday
		Last Week
	Due	None
		Over Due
		In one day
		In one week
	Constant	
	created	None
		Yesterday
		Last Week

Figure 8–4 Case List Panel Advanced Search

8.3.3 Working With Case Activities

Activities for the selected case are displayed in the Activities panel.

When the Activities panel is hidden, click **Show Activities** to display it.

8.3.3.1 Activities Panel

Use the Activities panel of the Case page to view activities and their details, and to initiate new activities. Activities are displayed in a series of sections in the panel.

Element	Description
Activities Filter	Select the type of activities to display. Choices are:
	 All
	 Available
	 Running
	Completed
	 Errored

Element	Description
Activities List	Activities are displayed in sections, grouped by type. If no activities of a particular type exist, the section is not displayed. Section types include:
	Available
	 Running
	Completed
	 Errored
Add Activities	Click to select an activity to add to the case. The choices available in the drop-down list depend on the case configuration, and whether the activity has already been added.
Load More Items	This link appears when there are more items than will fit in the default panel size. Click to display additional items.

8.3.3.2 Filtering the Activities Displayed

Use the Activities drop-down list to select which activities are displayed. Choose from:

- All
- Available
- Running
- Completed
- Errored

Activities for the case are displayed in their own section.

To see the details for an activity, click it.

8.3.3.3 Initiating an Activity

Activities that can be initiated are displayed in the **Available** section of the Activities panel. Activities that are required for a case are marked with an asterisk.

To initiate an activity, click its link in the Activities panel. Provide any required information in the dialog box that is displayed and click **Initiate**.

8.3.3.4 Viewing Activity Details

Activities that can be viewed are displayed in the **Available**, **Running**, **Completed** and **Errored** sections of the Activities panel.

To view details for an activity, hover the mouse over its link in the Activities panel. Click **More Details...** in the pop-up window that is displayed.

8.3.4 Viewing the Audit Trail

The Audit Trail panel is displayed by default when viewing a case. If it is not visible, click the **Audit Trail** link below the **Navigation** bar to display it.

You can view only one of the following panels at a time:

- Audit Trail panel
- Data panel: see "Data Panel"
- Documents panel: see "Documents Panel"

8.3.4.1 Audit Trail Panel

Use the Audit Trail panel to view case events, details on the users who executed the events, and to set permissions for events. Events are displayed in a series of sections in the panel.

Element	Description		
Search Criteria	Enter the text to search for events by title.		
Search	Click the Search icon to search for events using the Search Criteria.		
All	Click to display all events.		
Comments	Click to display only events that are comments.		
Event	Use the event drop-down list to filter which event types are displayed in the Audit Trail panel.		
	Choices include:		
	Activity		
	Lifestyle		
	Document		
	Milestone		
	User Defined		
	Data Upload		
Update Permission	Click to change the user permissions for the currently selected event. If no event is selected, the Update Permission icon is disabled.		
Audit Trail Details	Audit trail events are displayed as a series of sections in the Audit trail panel. Each event includes:		
	Event type icon		
	Event name		
	Event timestamp		
	Event detail summary (optionally)		
	User information		
Load More Items	This link appears when there are more items than will fit in the default panel size. Click to display additional items.		

8.3.4.2 Viewing Events in the Audit Trail

Case events are displayed in the Audit Trail panel. Each event is displayed in a section of the panel and includes the following information:

- Event type icon
- Event name
- Event timestamp
- Event detail summary (optionally)
- User who performed the event

To see additional information about the user associated with an event, hover your mouse over the user name.

8.3.4.3 Updating Event Permissions

You can change the user permissions for an event.

To update user permissions for an event

- 1. Select the event from the Audit Trail panel.
- 2. Click the Update Permissions icon. The Permission dialog box displays.
- **3.** Select the desired permission for the event from the drop-down list.
- 4. Click Apply.

8.3.5 Data Panel

Use the Data panel to view and modify the data associated with a case.

To view the Data panel, click the Data link on the Case page.

You can view only one of the following panels at a time:

- Audit Trail panel: see "Viewing the Audit Trail"
- Data panel
- Documents panel: see "Documents Panel"

8.3.5.1 Working with Data Forms in the Data Panel

The specific forms, tabs, fields and controls that appear in the Data panel are dependent on the Case project design for the case you are working on, and not pre-determined by Process Workspace.

8.3.6 Documents Panel

Use the Documents panel to view and add documents attached to a case.

To view the Documents panel, click the Documents link on the Case page.

You can view only one of the following panels at a time:

- Audit Trail panel: see "Viewing the Audit Trail"
- Data panel: see "Data Panel"
- Documents panel

8.3.6.1 Documents Panel

Use the Documents panel to view documents attached to the case and details on the users who uploaded them. You can also upload additional documents, and set permissions for the documents.

Documents are displayed in a series of sections in the panel.

Element	Description
Upload File	Click to upload a new document to the case. The Upload File dialog box is displayed.
Element	Description
---------------------	---
Document Details	Documents are displayed as a series of sections in the Documents panel. Each document includes:
	Document icon
	Document title
	Document comment
	Creator name
	Timestamp
	Action List control
Action List	Use the Action List to choose an action to perform on the document, such as Delete.
Load More Items	This link appears when there are more items than will fit in the default panel size. Click to display additional items.

8.3.6.2 Viewing Documents in the Documents Panel

Documents attached to the case are displayed in the Documents panel. Each document is displayed in a section of the Documents panel and includes the following:

- Document icon
- Document title
- Document comment
- Creator name
- Timestamp

To download the document, click the Document title. A pop-up window is displayed containing the contents of the document.

8.3.6.3 Performing Actions on a Document

You can perform actions on a document attached to a case. For example, you can delete a document if you have the permissions required to do so.

To perform an action on a document

- 1. Select the document from the Documents panel.
- **2.** Click the **Action List** drop-down list. A list of actions available to your user account displays.
- **3.** Select the desired action for the document from the drop-down list. The action is taken immediately.

8.3.7 Viewing Milestones

Milestones are checkpoints that indicate the progress of the case.

8.3.7.1 Milestones Panel

Use the Milestones panel on the Case page to view case milestones, and to change the status of milestones. Milestones are displayed in a series of sections in the panel.

Element	Description
Milestone	Each milestone section includes:
Details	 Status icon: color-coded icons are used to indicate the status of the milestone. See below for details.
	Milestone name
	 Milestone status. See below for details.
	 Timestamp (optional): includes completion information and the user who completed the milestone
	 Comment (optional): includes comment information provided when the milestone changed.
Status icon	Color-coded icons indicate the status of the milestone:
	Blue: Open.
	Green: Completed
	Amber: Reopened
Milestone Status	Change the status of a milestone by selecting a value from the milestone status drop-down list.
	Milestones with status:
	• Open : can be changed to Mark Complete
	• Closed: can be changed to Reopened
	Reopened: can be changed to Closed
Load More Items	This link appears when there are more items than will fit in the default name size. Click to display additional items

8.3.7.2 Changing a Milestone's Status

You can change the status of a milestone. For example, a milestone with status Completed can be changed to Reopened.

Note that in most cases, milestone status changes occur automatically when certain case activities happen.

To change a milestone's status

- **1.** Select the desired status from the **Milestone Status** drop-down list. The Comment entry dialog box is displayed.
- 2. Enter an appropriate comment for the action you are taking.
- 3. Click OK. The milestone is updated with the new status.

8.3.8 Working with Stakeholders

Stakeholders represent interested parties in the case. Stakeholders have users, groups or roles associated with them that are involved in the processing of a case. Case elements such as activities can be restricted to certain stakeholders, and therefore, to particular users, groups or roles.

8.3.8.1 Stakeholders Panel

Use the Stakeholders panel on the Case page to view and modify case stakeholders. Stakeholders are displayed in a series of sections in the panel.

Element	Description
Add Stakeholder	Click to add a new stakeholder definition to the case. When you click Add Stakeholder, the Add Stakeholder dialog is displayed.
	The new stakeholder specifies at least one associated user, role, or group.
Edit Stakeholder	Click to edit the currently selected stakeholder in order to add a new user, role, or group. When you click Edit Stakeholder, the Edit Stakeholder dialog is displayed.
Stakeholders Details	 Each section in the Stakeholders details panel includes: Stakeholder name Stakeholder member icons (user, role or group) and stakeholder members
Load More Items	This link appears when there are more items than will fit in the default panel size. Click to display additional items.

8.3.8.2 Adding Stakeholders

You can add new stakeholders to a case.

To add a new stakeholder

- **1.** Click the **Add Stakeholder** button (green plus sign). The **Add Stakeholder** dialog displays.
- 2. Enter a name for the new stakeholder, and optionally, a comment.
- **3.** Specify at least one user, role or group.
- 4. Click OK.

8.3.8.3 Editing Stakeholders

You can edit a case's stakeholders (if permissions allow it).

To edit a stakeholder

- 1. Select the stakeholder you wish to edit from the Stakeholders panel.
- 2. Click the Edit Stakeholder button. The Edit Stakeholder dialog displays.
- **3.** Modify the stakeholder values.
- 4. Click OK.

8.4 Working on Cases

When a new case is created in the system, a task is added to your Worklist, visible in the **Tasks** page. When you claim a task associated with a new case, you can then work on the case in the **Case** page.

Tasks are still performed in the Tasks page of Process Workspace. The Case page is used more as a point of reference to see information about a case, add or view documentation, comments and activities, and to move the case process forward.

Part II

Administering Process Workspace

This part contains the following chapters:

- Chapter 9, "Managing Process Workspace"
- Chapter 10, "Managing Your Organization in Process Workspace"
- Chapter 11, "Modifying Running Process Instances in Process Workspace"

Managing Process Workspace

This chapter explains the various tasks an administrator can perform in Process Workspace, including setting the display of the Work Items panel, and administering and configuring task-related information.

This chapter contains these topics:

- Section 9.1, "What You May Need to Know When Managing Process Workspace"
- Section 9.2, "Setting the Display of the Work Items Panel in Process Workspace"
- Section 9.3, "Administering Approval Groups in Process Workspace"
- Section 9.4, "Managing Mapped Attributes (Flex Fields) in Process Workspace"
- Section 9.5, "Using Task Configuration in Process Workspace"
- Section 9.6, "Using Evidence Search in Process Workspace"
- Section 9.7, "Managing Exceptions in Process Workspace"
- Section 9.8, "Using the User Metadata Migration Utility"

9.1 What You May Need to Know When Managing Process Workspace

When an end user launches an application from the Applications panel of the Process Tracking page, the application is launched and a form appears. If the user dismisses that form, then that process instance continues to run, but as a superfluous instance never to be processed or used. Such superfluous instances can be cleaned up by a background instance that can be specified and scheduled by the administrator of the BPMN service engine. If you are the administrator of the BPMN service engine and want more information about specifying this background instance, see *Oracle Fusion Middleware Administrator's Guide for Oracle SOA Suite and Oracle Business Process Management Suite*.

9.2 Setting the Display of the Work Items Panel in Process Workspace

To set the display of the Work Items panel, administrators can specify various application preferences. To do this: From the Process Workspace global toolbar, click **Administration**, then click **Application Preferences**.



ORACLE' Business Process Workspace

🔞 Home | Administration | Preferences | Help | Logout | weblogic 읻

Administration Areas	Aux licetian Desferences
Organization	
Delea	
Clearland	Login page years label LAREL LOCIN DEALM
Calendars	
Organization Units	Resource bundle oracle.bpel.worklistapp.resource.WorklistResourceBundle
Parametric Roles	Use language settings of 🔘 Browser 🔿 Identity Provider
Extended User Properties	User Name format. 🔿 User Id 🕥 User Name
Flex Fields	g,
Public Flex Fields	Branding And Skinning
Protected Flex Fields	Branding Logo /afr/logo-oracle-red.png
Task Administration	Branding THE LAREL WOOKSDACE TITLE
Approval Groups	
Evidence Search	Choose a Skin fusion 🛛 💌
Task Configurations	Application customization class name
Application Preferences	Man Tack actions to an image 🚺 📫
Application Preferences	
	Show Standard Tabs 🗹 💿 Tasks 🗹 🔿 Activity Guide 🗹 🔿 Process Tracking 🗹 🔿 Standard Dashboards
	Flex Field INTEGER Display



This section contains these topics:

- Section 9.2.1, "How to Specify the Login Page Realm Label in Process Workspace"
- Section 9.2.2, "How to Specify the Resource Bundle in Process Workspace"
- Section 9.2.3, "How to Specify the Language Locale Information in Process Workspace"
- Section 9.2.4, "How to Specify the User Name Format"
- Section 9.2.5, "How to Specify a Branding Logo in Process Workspace"
- Section 9.2.6, "How to Specify the Branding Title in Process Workspace"
- Section 9.2.7, "How to Choose a Skin in Process Workspace"
- Section 9.2.8, "How to Enable Customized Applications and Links in Process Workspace"
- Section 9.2.9, "How to Specify an Image for a Task Action"
- Section 9.2.10, "How to Specify, Show, or Hide the Default Tab Pages in Process Workspace"

9.2.1 How to Specify the Login Page Realm Label in Process Workspace

If the identity service is configured with multiple realms, then, when a user logs in to Oracle BPM Worklist, the login page displays a list of realm names. LABEL_LOGIN_ REALM specifies the resource bundle key used to look up the label to display these realms.

You can change the term *realm* to fit the user community—terms such as *country*, *company*, *division*, or *department* may be more appropriate. To change the term *realm*, customize the resource bundle, specify a resource bundle key for this string, and then set the **Login page realm label** parameter to point to that resource bundle key.

For information about customizing the resource bundle, see Appendix B, "Customizing a Resource Bundle".

For information about specifying a resource bundle, see Section 9.2.2, "How to Specify the Resource Bundle in Process Workspace."

9.2.2 How to Specify the Resource Bundle in Process Workspace

The resource bundle provides the strings displayed in the Worklist Application. By default, the class path to the resource bundle is:

As an administrator, you can add or modify strings shown in the application by creating a custom resource bundle. You can then use the **Resource Bundle** field in the Application Preferences page to specify the class path to your custom resource bundle.

For more information about customizing the resource bundle, see Appendix B, "Customizing a Resource Bundle."

9.2.3 How to Specify the Language Locale Information in Process Workspace

From the Application Preferences page, you can specify how the Process Workspace display language is determined. Information about the language locale can be derived from either the user's browser or the identity provider that stores information about Process Workspace users.

You reach the Application Preferences page by clicking **Administration** on the global toolbar at the very top of Process Workspace, then, under Administration Areas, selecting **Application Preferences**. The Application Preferences panel appears in the right pane. To specify the display language, in the **Use language settings of** field, select either **Browser** or **Identity Provider**.

ORACLE' Business F	rocess Workspace 🔬 Home Administration Preferences Help Logout weblogic 🧲
Administration Areas	Application Preferences Save Revert -
Administration Areas Organization Roles Calendars Organization Units Parametric Roles Extended User Properties Flex Fields Public Flex Fields Protected Flex Fields Task Administration Approval Groups Evidence Search Task Configurations Application Preferences Application Preferences	Application Preferences Save Revert ▼ Login page realm label LABEL_LOGIN_REALM Resource bundle Resource bundle Use language settings of O Browser () Identity Provider User I die User I
	Show Standard Tabs V I Tasks V Activity Guide V Process Tracking V OStandard Dashboards Flex Field INTEGER Display V

Figure 9–2 Specifying Language Local Information

9.2.4 How to Specify the User Name Format

From the Application Preferences page, you can specify how the user name format is displayed in the Creator, Assignee or other name columns throughout Process Workspace. You can select to display the user ID (for example, jstein) or the full user name (for example, John Stein).

You reach the Application Preferences page by clicking **Administration** on the global toolbar at the very top of Process Workspace, then, under Administration Areas, selecting **Application Preferences**. The Application Preferences panel appears in the right pane. To specify the display language, in the **User Name Format** field, select **User ID** or **User Name**, as shown in Figure 9–3.

Figure 9–3 User Name Format

	Process Workspace 😽 Home Administration Preferences Help Logout	weblogic
Administration Areas	Application Preferences Save	Revert +
Organization		
Roles		
Calendars		
Organization Units		
Parametric Roles		
Extended User Properties		
Flex Fields		
Public Flex Fields	Login page realm label LABEL_LOGIN_REALM	
Protected Flex Fields	Resource bundle orade. boel.workistapp.resource.WorkistResourceBundle	
Task Administration	The local use retrieve of the Provider of Identity Broader	
Approval Groups		
Evidence Search	User Name romat 🛞 User I a 🕲 User Name	
Task Configurations	Brandina And Skinning	
Application Preferences	Branding Loop / Joff Joop-oracle-red.ong	
Application Preferences		
	Branding Title (LABEL_WORKSPACE_TITLE	
	Choose a Skin fusion 💌	
	Application customization class name	
	Map Task actions to an image 🖾 🔸 📃	
	Show Standard Table 📝 👩 Tasks 📝 🔗 Activity Guide 📝 🔗 Process Tracking 📝 💮 Standard Dashboards	
	Flex Field INTEGER Display	

9.2.5 How to Specify a Branding Logo in Process Workspace

A branding logo is the image displayed in the top left corner of every page of the Worklist Application. The Oracle logo is the default, and you can change it to one of your choosing.

Note: The ideal image size is 120px x 40px (length x width) for proper display. Although images with high resolution and size are compressed to fit the branding logo size, smaller images display better.

To specify the branding logo:

Do one of the following:

- Refer to an external image-hosting website. To do this: In the Branding Logo field, enter the URL of the image.
- Upload an image to a particular location on the server and, in the Branding Logo field, enter its relative path, for example, /afr/my_logo.png.
- Refer to an image from the shared library. To do this: In the Branding Logo field, enter the path of the logo name as found in the shared library, for example, /my_ logo.png

Note: Customizing the branding logo from either the Worklist Application or Process Workspace changes the logo in both applications. For example, if you change the logo from Worklist Application, then the Process Workspace logo is changed automatically.

For more information about deploying images in a shared library, see Appendix D, "Deploying Images and JAR files as Part of a Shared Library JAR file".

9.2.6 How to Specify the Branding Title in Process Workspace

You can specify the title for your site, changing the default title, *BPM Worklist*, to one that you choose.

To specify the branding title:

Do one of the following:

- In the Branding Title field, enter a simple string for your title.
- In the Branding Title field, enter a label that refers to a key-value pair in the Resource Bundle. In this way, you can internationalize your title, for example, LABEL_WORKLIST_TITLE.

9.2.7 How to Choose a Skin in Process Workspace

A skin determines the look and feel of your graphical interface.

To choose a skin:

Do one of the following:

From the Choose a Skin list, select one of the default ADF skins

If you want the Human Task flow application to use the skin selected in Process Workspace, you must verify that the trinidad-config.xml file (found in the Human Task flow application) includes the following:

<skin-family>#{param.sf}</skin-family>

 Upload your own customized skin.css file in a.JAR file and deploy it as a part of shared library. Then, when you restart your application from the console, your custom skin appears in the Choose a Skin list.

To create a customized skin:

1. Create a directory structure similar to the following example:

```
C:\temp\META-INF\adf\oracle\skin\images
META-INF\skins\custom.css
META-INF\trinidad-skins.xml
```

Note that, in this example, you can change the word *custom* to the name of your own customized skin.

2. Make sure the content of trinidad-skins.xml file is as follows:

```
<?xml version="1.0" encoding="ISO-8859-1"?>
<skins xmlns="http://myfaces.apache.org/trinidad/skin">
<skin>
<id>custom.desktop</id>
```

```
<family>custom</family>
<extends>custom.desktop</extends>
<render-kit-id>org.apache.myfaces.trinidad.desktop</render-kit-id>
<style-sheet-name>skins/custom.css</style-sheet-name>
</skin>
</skins>
```

3. Create the .JAR file by issuing the following command from the c:\temp directory:

```
jar -cvf customSkin.jar META-INF/
```

4. Copy this Jar file to the directory /scratch/username/sharedLib.

Note: Refer to the images in your CSS file this way:

```
../adf/oracle/skin/images/example.gif (with the two trailing dots).
```

This allows the search for the META-INF root to start one directory above the META-INF/skin directory in which the .css file is located.

For more information about deploying images in a shared library, see Appendix D, "Deploying Images and JAR files as Part of a Shared Library JAR file".

9.2.8 How to Enable Customized Applications and Links in Process Workspace

In Process Workspace, you can create customized external applications and links that become available in the Favorites list in the Views, Applications, and Favorites pane. Moreover, in both Process Workspace and the Worklist Application you can specify the columns that appear in the inbox.

To enable customized applications:

- **1.** In the Application Preferences page, enter the class name of your custom application in the **Application customization class name** field.
- **2.** Restart the application from the console.

Depending on your customization, you can now see its effects.

If your customization is for Process Workspace and involves either creating an external application or specifying inbox columns in the Task Details pane or both, you see the following:

- Your custom application listed in the Favorites list of the Tasks, Views, and Favorites pane.
- The columns of the inbox adjusted according to your specifications.

For information about customizing applications, see Appendix C, "Customizing Applications in Process Workspace and the Worklist Application".

9.2.9 How to Specify an Image for a Task Action

If you are an administrator, then you can specify whether an action is displayed with a red X icon or with a green check mark icon.

To specify an image for a task action:

1. Select Administration, then Application Preferences.

- **2.** From the lists in the **Map task actions to an image** field, select the task actions you want to map to either the green check mark icon or the red X icon.
- 3. Click Save.

9.2.10 How to Specify, Show, or Hide the Default Tab Pages in Process Workspace

If you are an administrator, then you can specify whether to show or hide the Process Workspace standard tabs, for example, Tasks, Activity Guides, Process Tracking, and Standard Dashboards. Also, from the tabs you decide to show, you can select the one to be the default tab when a user logs in.

To show or hide the standard Process Workspace tabs:

- 1. Select Administration, then Application Preferences.
- **2.** From the **Show Standard Tabs** row, select the check boxes for the tabs you want to show, or clear them for the tabs you want to hide.
- **3.** From the tabs you decide to show, specify which tab is to be the default when a user logs in. Do this by selecting the radio button next to the check box for that tab. If you do not select a default tab, then the standard tabs will not show or hide as you specified.
- 4. Click Save.

Figure 9–4 shows the Application Preferences page with the **Show Standard Tabs** row highlighted. It calls out a check box for selecting a tab to show or hide, and a radio button for selecting a tab to be the default at user login.

Figure 9–4 Show Standard Tabs Selection



9.3 Administering Approval Groups in Process Workspace

An approval group consists of a name and a predefined set of users configured to act on a task in a certain pattern. This pattern is similar to a human workflow routing slip pattern where users can act on tasks in serial or in parallel. An approval group also can contain a nested approval group in the pattern.

The name of an approval group is necessary when specifying the approval group list builder. The pattern configured in the approval group is used by default to order the

users who must act on the task. However, when creating the list builder, the default pattern can be overridden by specifying the voting method.

Process Workspace enables with administrator rights to manage approval groups. This section contains these topics:

- Section 9.3.1, "How to Use Task Forms"
- Section 9.3.2, "How to Create Mapped Attribute Labels"
- Section 9.3.3, "How to View Approval Groups"
- Section 9.3.4, "How to Search for an Approval Group"
- Section 9.3.5, "How to Add a Static Approval Group"
- Section 9.3.6, "How to Add a New Member to a Static Approval Group"
- Section 9.3.7, "How to Delete a Member from an Approval Group"
- Section 9.3.8, "How to Move an Approval Group's Members"
- Section 9.3.9, "How to Nest Approval Groups"
- Section 9.3.10, "How to Rename an Approval Group"
- Section 9.3.11, "Using Dynamic Approval Groups"
- Section 9.3.12, "How to Delete an Approval Group"

9.3.1 How to Use Task Forms

The human workflow service creates tasks for users to interact with the business process. Each task has two parts—the task metadata and the task form. The task form is used to display the contents of the task to the user's worklist.

The task form is designed in Oracle JDeveloper with Oracle Application Development Framework (Oracle ADF). If you are a process designer and want more information, see "Designing Task Display Forms for Human Tasks," in *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

Process Workspace displays all worklist tasks that are assigned to a user or a group. When a worklist user drills down into a specific task, the task display form renders the details of that task. For example, an expense approval task may show a form with line items for various expenses, and a help desk task form may show details such as severity, problem location, and so on.

The sections that follow describe the task form in Process Workspace that users with administrator rights use to manage tasks. Figure 9–5 shows a task form.

task 1					Actions - Complete
Assignees weblogic John Steinbeck James Cooper Creator weblogic Created Aug 17, 2012 5:32 AM Updated Aug 21, 2012 7:47 AM Escivitation Date	Acquired By weblogic Category Percentage Complete Due Data Oct 19, 2012 11:50 PM	8	Task Number 200000 Priority 1 - State Assigned		
	Start Date Aug 17, 2012 11:59 PM	10			
≥ History					
a weblogic, John Steinbeck, James Cooper 1 Assigned Aug 17, 2012 5:32 AM	0				
weblogic weblogic, John Steinbeck, James Cooper- Assigned Aug 17, 2012 5:33 AM	(a)				
Weblogic, John Steinbeck, James Cooper 3 Acquired Aug 21, 2012 7:47 AM weblogic	٥				
🔏 weblogic, John Ste					
A weblogic, John Ste					
A weblogic, John Ste					
✓ Comments		🔮 🗹 Attachme	ents		4×
No data to display			Updated By	Date Updated	

Figure 9–5 Task Form

9.3.1.1 Header View

The **Header** view, shown in Figure 9–6, is created during design time in Oracle JDeveloper using the header drop handler.

Figure 9–6 Header View in Process Workspace

task 1				🗶 🗘 🗘 Complete Claim
🖂 🤷 Details				
Assignees	weblogic John Steinbeck James Cooper	Acquired By	Task Number 200000	
Creator	weblogic	Category	Priority 1 🖛	
Created Updated	Aug 17, 2012 5:32 AM Aug 18, 2012 8:08 AM	Percentage 50.0 Complete	State Assigned	
Expiration Date		Due Date Oct 19, 2012 11:59	рм 🖏	
		Start Date Aug 17, 2012 11:59	РМ 🖾	

By default, the JDeveloper drop handler includes the **Header** fields listed in Table 9–1. However, using JDeveloper, the developer can include or remove any of the fields based on the use case.

Field	Description
Task Number	The number automatically assigned to the task.
State	The current status of the task.
Outcome	The user outcome. For example, if the user clicks Approve , the Outcome field displays "Approve."
Priority	The priority level assigned to the task.
Creator	The user who originated the task.
Created	The task's creation date.
Updated	The date the task was last updated.
Expires	The date the task expires.

Table 9–1 Header Fields

Field	Description
Assignees	The name(s) of the administrator(s) to whom the task has been assigned.
Acquired By	The name of the user who claims the task before any action is taken on it.
Due Date	The date by which the task is due.

 Table 9–1
 (Cont.)
 Header Fields

The **Header** also contains custom and system actions. *Custom actions* are those that depend on task metadata outcomes. For example, if the metadata contains Approve and Reject outcomes, then **Approve** and **Reject** appear in the **Header** as custom actions. If the metadata contains more than two outcomes, then the custom actions appear in the **Header** as a dropdown list instead of separate buttons.

System actions, such as Escalate, Suspend, and Resume, always appear in a dropdown list. The actions that appear depend on what the user is doing. For example, after a task has been initiated it can be withdrawn. Subsequently, if a user logs into the Process Workspace to view the details of an initiated task, then **Withdraw** appears in the list containing the available actions.

Table 9–2 lists all the actions the administrator can perform from the **Header** and their descriptions.

Action	Description
Reject	Enables the administrator to reject the task.
Approve	Enables the user to approve the task.
Delegate	Enables the administrator to assign a task to another person to act on his or her behalf.
Reassign	Enables the administrator to transfer a task to another administrator.
Suspend	Enables the administrator to suspend the task to work on it at a later time.
Resume	Enables the administrator to resume a suspended task.
Withdraw	Enables the administrator to withdraw an initiated task.
Escalate	Enables the administrator to escalate a task to a supervisor.
Delete	Enables the administrator to delete a task. Appears for all "To Do" tasks.
Purge	Enables the administrator to purge a task. Appears for all "To Do" tasks.
Go	Enables the administrator to "go" to the action selected from the dropdown list.
Save	Enables the administrator to save the task.

Table 9–2Header Actions

9.3.1.2 Task Payload View

The **Task Payload** view, shown in Figure 9–7, displays the details of the task parameters and provides the ability to update them. Any parameter that has been designed based on SDOs also can be viewed here. While the values of the general parameters are based on what was passed to the task or update by any user, the values

of the SDO-based parameters reflect the current value of the business object as it was updated in the underlying application.

Figure 9–7 Task Payload

Contents		
Id	2222	
Creator	jstein	
CreatedDate	4/25/2008	
CostCenter	HW	
ReimbCurrency	\$	
Purpose	Educational expenses	
Status	CREATED	
ExpenseId	ReceiptDate	ReceiptAmount
2222	4/23/2008	4000
2222	4/23/2008	80
4))+
		1.

If you are a process designer and want more information, see "Designing Task Display Forms for Human Tasks" in *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

9.3.1.3 Task History View

The **Task History** view, shown in Figure 9–8, provides a graphical and tabular view of events in the task life cycle. In addition, if the Edit Approver Configuration option was selected in the designer, special controls in the tabular view that allow future approvers to be edited are available. Any approver added manually can be deleted and new approvers can be added. Approver changes made in the tabular view are immediately reflected in the graphical view. When the task is saved or a custom action, such as approve or reject, is performed on the task, all approver-related changes also are saved.



Figure 9–8 Task History in Process Workspace

If the **Allow all participants to edit future participants** option is selected while configuring the approval task, the history region displays additional actions that allow a participant to edit the future participants list.

Figure 9–9 shows the addition of the **Apply** and **Reset** buttons.

Figure 9–9 Task History - Additional Actions

History			
+ /	🕅 Task Snapshot 🛛 🗹 Future Participants 🗹 Full task actions 📝 All parent tasks		Apply Reset ~
#	Participant	Action	Action Date
1	🖃 🕼 SupervisoryStage		
1.1	🔏 wfaulk(user)	ASSIGNED	Jul 25, 2008 4:10 PM
1.2	🔏 cdickens(user)		
2	🖃 🕼 Lines		
2.1	🖃 👑 Lines		
2,1.1	🖃 🔐 Lines		
2.1.1.1	🔏 wfaulk(user)	£.	
2.1.1.2	🔏 cdickens(user)		
2.1.2	🗄 👑 Lines		

Table 9–3 describes all additional approval-task actions.

Action	Description
Add	Enabled when the user selects a future participant. When this option is selected, the Add Participant dialog opens and the user can insert ad- hoc participants.
Edit	Enabled when the user selects a participant that has been inserted ad hoc. When this option is selected, the Edit Participant dialog opens and user can move the position of an inserted participant.
Delete	Enabled when the user selects a participant that has been inserted ad hoc. When this option is selected, the corresponding participant is deleted.
Apply	Persists the edits to the future participant list.
Reset	Resets the edits from the future participant to a system-generated list.

 Table 9–3
 Edit Future Participants List Actions

Table 9–4 lists the actions the administrator can perform from the **Task History** view and their descriptions.

Action	Description
Task Snapshot	Displays the task details for the selected version.
Full task actions	If checked, displays the full task-action history. If unchecked, displays only action history. By default, the box is unchecked.
All parent tasks	If checked, displays the parent task in a sub-task view. If unchecked, displays only the sub-task history. By default, the box is checked.
Show future approvers	If checked, displays the future approver with history. By default, the box is unchecked.

Table 9–4 Task History Actions

9.3.1.4 Comments and Attachments View

The **Comments and Attachments** view, shown in Figure 9–10, is created in JDeveloper by using the task data control drop handler. It includes a text-entry field in which to enter comments about the task, and the functionality to attach supporting documents.

Figure 9–10 Comments and Attachments

✓ Comments	9	✓ Attachments			4 ×
No data to display			Updated By	Date Updated	
		No data to display			

9.3.2 How to Create Mapped Attribute Labels

You use mapped attributes to store use-case-specific data, such as data extracted from a task's payload. You can view and create mapped attribute labels on the server by using Process Workspace.

Note: You must have the *workflow.mapping.protectedFlexfield* privilege to create protected flexfield attributes. The default administrative user, *weblogic*, has this privilege.

For more information, see Section 9.4.1, "How to Specify Mapped Attributes."

To view attribute labels:

- 1. Click the Administration in the upper-right corner of the page.
- **2.** In the Administration pane, click the **Protected Flex Fields** link. A page similar to the one shown in Figure 9–11 appears.

Figure 9–11 Flexfield Mapping: Protected

Iministration Areas	Flex Field Mapping : Pr	otected				
rganization						
Roles	View existing flex field mapping	s (i.e. task attribute labels). (Freate additional flex field labels	and map these to task attri	butes.	
Ialendars	The labels will be displayed to d	ne enu users, anu snouiu be i	user-menuity terms for the task a	unbutes.		
rganization Units			Recurso all mannings			
arametric Roles			Edit mappings by ta	sk type:		
Extended User Properties						
2x Fields		-	100		4	
Public Flex Fields		Show: All Attributes	v	~ / %		
rotected Flex Fields		Label	Attribute Format	Description	✓ Details	1
k Administration		industry	ProtectedTextAttribu	Auto-generat	discount (ProtectedNumberAttribute3) is being used in:	
Approval Groups		discount	ProtectedNumberAttr	Auto-generat	'discount' (Auto generated	
udenes Creek		numQuoteEdits	ProtectedNumberAttr ProtectedNumberAttr	Auto-generat	RequestQuote for mapping for process)	
vidence bearch		revenuebimension	Protectedinginberatur	Autorgenerat		
ask Configurations						
plication Preferences						
Application Preferences						
	•	<		2		

The page displays a list of existing attribute labels. You can filter the list by selecting an attribute type from the **Show** list. Clicking a specific label displays the list of mappings the attribute uses in the **Details** panel.

To create an attribute label:

- 1. Click the **Create Label** (+) button. The Create Label dialog displays.
- 2. Select an attribute type and task attribute from the dropdown lists.
- **3.** Enter a unique label name and a description.
- **4.** Click **Create**. The label is created and is made available for mapping in task components.

To delete an attribute label:

To delete an attribute label, first select it from the list of attribute labels, then click the **Delete Label** (-) button.

Note: Attribute labels can be deleted only if they are not used in any mappings.

9.3.2.1 Importing and Exporting Attribute-Label Definitions

If attribute labels have been defined on one server and must be re-created on another, then you can use the user metadata migration utility to export a list of protected attribute labels from the server on which they were defined to an XML file. The utility then can deploy the attribute labels from this file to a new server. This eliminates the necessity to manually re-create the attribute labels manually in Process Workspace. For more information, see Section 9.8, "Using the User Metadata Migration Utility."

9.3.2.2 Internationalizing Attribute Labels

When attribute labels are displayed to end users, for example in the task listing page of the Process Workspace, the label name that is used is the one specified when the label was created. In cases where users of different nationalities may see the label, a translation of the label name appropriate to Process Workspace user's locale can be displayed instead. Translations of attribute labels can be customized using the **WorkflowLabels.properties** resource bundle.

For more information, see Section 9.2.3, "How to Specify the Language Locale Information in Process Workspace." on page 9-3.

9.3.3 How to View Approval Groups

To view approval groups:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the navigation pane of the Approval Groups page, select an approval group. A details page for that approval group appears in the right pane, similar to the one shown in Figure 9–12.

Figure 9–12 Process Workspace: Approval Group Detail

Details		Reset Apply
* Name DisbursementTeam		
Members		+
	Ļ	
	bpalmer	
	a rjames	
	*	

The figure shows that the DisbursementTeam approval group has two users, bpalmer and rjames. The users will act on a task in a specific sequence configuration.

9.3.4 How to Search for an Approval Group

You can search for an approval group either by user name or group name.

To search by user name:

1. From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.

- **2.** Under Task Administration, click **Approval Groups**. The Approval Groups page appears in the right pane.
- 3. Select User from the list.
- **4.** Enter the full user name for the user in the field. (You also can perform a wildcard search (*) with a partial user name.)
- **5.** Click the action (>) button.

A list of all approval groups to which the user belongs displays in the left pane, as shown in Figure 9–13.

Figure 9–13 User Name Search Results

lm* 🤇	
m	

Clicking the approval group name refreshes the details pane on the right with the structure of that group.

To search by group name:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the left pane of the Groups page, enter the full group name in the field. (You also can perform a wildcard search (*) with a partial group name.) and click **Search**. A list of all matching approval groups displays in the left pane.

Clicking the approval group name refreshes the Details pane on the right with the structure of that group.

9.3.5 How to Add a Static Approval Group

To add a static approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the middle pane.
- **3.** Click the **Add** (+) button and select **Create Static Approval Group** from the list, as shown in Figure 9–14.

Figure 9–14	Create Approval	Group: Select	Static Group
•			

ORACLE' Business	Process Workspa	ace		1	Home Administration Prefer	ences Help Logout	weblogic Ϙ
Administration Areas	🔟 Groups	📲 👻 🗶 Details				Save	Revert
Organization	Search	Create Static Approval G Create Dynamic Approva	iroup al Group ynamic appr group				
Roles	🖏 new static appr	group Memb	ers			÷/%	
Calendars	🚳 new dynamic	appr group				• • • •	
Organization Units				Ē			
Parametric Roles				<u> </u>			
Extended User Properties				*			
Flex Fields				🖏 <u>new static ap</u>	ipr gr		
Public Flex Fields				÷			
Protected Flex Fields				9 achviet			
Task Administration				Cal dictilitat			
Approval Groups				. Á			
Evidence Search							
Task Configurations							
Application Preferences							
Application Preferences							
	•	•					
	Π	F					

- **4.** Enter a new name for the group.
- 5. Click Apply.

You can add members to the new approval group.

9.3.6 How to Add a New Member to a Static Approval Group

Members of a static approval group can be either users or other approval groups.

To add a new user member to an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the middle pane.
- **3.** In the Groups pane, select an approval group.
- 4. From the Details page shown in Figure 9–15, click the Add (+) icon.

The other icons enable you to edit, delete, and reorder members in the approval sequence.

The Add to Group dialog box appears.

- 5. Select User.
- **6.** Do one of the following:
 - Enter a full user name and click OK.

The dialog box closes and the new member appears in the **Members** section of the **Details** pane.

Click the magnifying glass to search for a user.

If you click the magnifying glass, an Identity Browser pop-up dialog appears.

- 7. Select Users from the list.
- **8.** Enter a full name in the text-entry field and click **Search**. (You also can perform a wildcard search (*) with a partial user name.)

The Identity Browser dialog refreshes and the search results appear.

9. Choose a user from the list.

The details for that user appear in the Details section of the dialog.

- **10.** Click **OK**.
- 11. Click **OK** again to close the Add to Group dialog.

A node representing the selected user appears in the approval group structure in the **Members** section of the **Details** pane.

You can add more members to the approval group by repeating the steps above. The resulting approval group structure will look similar to the one shown in Figure 9–15.

Figure 9–15 Approval Group Structure: Multiple Members



9.3.7 How to Delete a Member from an Approval Group

To delete a member from an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the middle pane.
- **3.** In the Groups pane, select an approval group. The corresponding Details pane appears to the right.
- 4. Choose the appropriate member node from the approval group structure.
- 5. Click the **Delete** icon.

The approval group structure refreshes and the member node has been deleted.

9.3.8 How to Move an Approval Group's Members

To change the sequence order of an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the Groups pane, select an approval group. The corresponding Details pane appears to the right.
- 4. Choose a member node to move.
- **5.** Use the Push Member Up (^) and Push Member Down (v) icons to move the member to the desired location.

9.3.9 How to Nest Approval Groups

Nesting an approval group means making it part of another approval group.

To nest an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the Groups pane, select an approval group. The corresponding Details pane appears to the right.
- 4. Click the Add icon. The Add to Group dialog box appears.
- 5. Select Approval Group.
- 6. From the Member list, choose the approval group you want to add.

Its structure appears in the right pane.

7. Click **OK** again to close the Add to Group dialog box.

The new approval group appears in the approval group's structure.

9.3.10 How to Rename an Approval Group

To rename an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the Groups pane, select an approval group. The corresponding Details pane appears to the right.
- 4. Enter the new name of the approval group in the **Name** field.
- 5. Click Apply.

The name change is reflected in other approvals groups in which this approval group is nested.

9.3.11 Using Dynamic Approval Groups

}

Dynamic Approval Groups provide a way to create approval groups through a custom Java class at runtime. This requires the following:

- Writing a custom dynamic approval group class for the custom implementation by the developer
- Registering the custom dynamic approval group using the worklist apps UI by the IT department
- Making the class file available in a globally well-known directory that is part of the Service Integration class path

9.3.11.1 How to Write a Custom Dynamic Approval Group Class

To define a dynamic approval group, the user must define an implementation class using the interface file IDynamicApprovalGroup.java, defined by AMX for dynamic approval groups in the package oracle.bpel.services.workflow.task. This package contains only one public method that gets the approval group members. The Task object will be the only input parameter. The primary key list can be obtained from the task task/systemAttributes/collectionTarget.

Example 9–1 Implementation Class

public interface IDynamicApprovalGroup {

```
/**
      * Get members of this dynamic approval group
      * @param task Property bag containing information required to generate
          the approver list
      * @return list of IApprovalListMember including sequence, member, member_
          type; null for empty group
      * The primary key list can be obtained from task:
          task/systemAttributes/collectionTarget
      */
  public List getMembers(Task task )
     throws WorkflowException;
```

Figure 9–16 shows a code snippet for a sample dynamic approval group class.

Figure 9–16 Code for Dynamic Approval Group Class

```
package oracle.bpel.services.workflow.repos.test;
import java.util.ArrayList;
import java.util.List;
import oracle.bpel.services.workflow.WorkflowException;
import oracle.bpel.services.workflow.task.IDynamicApprovalGroup;
import oracle.bpel.services.workflow.runtimeconfig.impl.RuntimeConfigUtil;
import oracle.bpel.services.workflow.runtimeconfig.model.ApprovalGroupMember;
import oracle.bpel.services.workflow.task.model.Task;
import oracle.bpel.services.workflow.IWorkflowConstants;
public class SampleDynAprGrp1 implements IDynamicApprovalGroup {
   public List getMembers(Task task)
       throws WorkflowException
      {
        // Add logic later to do conditional return of different group mebers b
ased on some task attribute
        List approversList=new ArrayList();
         ApprovalGroupMember taskAssignee1 = RuntimeConfigUtil.getFactory().crea
teApprovalGroupMember();
        taskAssignee1.setMember("jlondon");
         taskAssignee1.setType(IWorkflowConstants.IDENTITY_TYPE_USER);
        taskAssignee1.setSequence(1);
         approversList.add(taskAssignee1);
        ApprovalGroupMember taskAssignee2 = RuntimeConfigUtil.getFactory().crea
teApprovalGroupMember();
        taskAssignee2.setMember("wfaulk");
         taskAssignee2.setType(IWorkflowConstants.IDENTITY_TYPE_USER);
        taskAssignee2.setSequence(2);
         approversList.add(taskAssignee2);
         ApprovalGroupMember taskAssignee3 = RuntimeConfigUtil.getFactory().crea
teApprovalGroupMember();
         taskAssignee3.setMember("jstein");
         taskAssignee3.setType(IWorkflowConstants.IDENTITY_TYPE_USER);
        taskAssignee3.setSequence(3);
         approversList.add(taskAssignee3);
        ApprovalGroupMember taskAssignee4 = RuntimeConfigUtil.getFactory().crea
teApprovalGroupMember();
        taskAssignee4.setMember("achrist");
        taskAssignee4.setType(IWorkflowConstants.IDENTITY_TYPE_USER);
        taskAssignee4.setSequence(4);
         approversList.add(taskAssignee4);
           turn approversList;
     }
```

9.3.11.2 How to Register the Custom Dynamic Approval Group Class

If you are a process designer and want more information, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*

9.3.11.3 How to Make the Custom Dynamic Approval Group Class Available

To make the class file available in a globally well-known directory that is part of the Service Integration class path, put your class files in the following Oracle WebLogic Server directory:

\$BEAHOME/AS11gR1SOA/soa/modules/oracle.soa.ext_11.1.1/classes

For example, for the Java class *oracle.apps.DynamicAG*, the path would be \$BEAHOME/AS11gR1SOA/soa/modules/oracle.soa.ext_ 11.1.1/classes/oracle/apps/DynamicAG.class. You **must** restart WebLogic Server after you put your class files there.

9.3.11.4 How to Add a Dynamic Approval Group

The following procedure explains how to add a dynamic approval group.

1. From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.

- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- **3.** In the left pane, click the **Add** (+) button and select **Create Dynamic Approval Group** from the list
- **4.** Enter a name and a class for the group.
- 5. Click Save.

9.3.12 How to Delete an Approval Group

To delete an approval group:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** Under Task Administration, click **Approval Groups**. The Groups page appears in the right pane.
- 3. In the Groups page, choose the approval group you want to delete.
- 4. Click the **Delete** (-) button.

A confirmation dialog box appears.

5. Click OK.

The approval group is deleted.

Note: If the approval group you deleted is nested in other approval groups, it also is deleted from those parent groups.

9.4 Managing Mapped Attributes (Flex Fields) in Process Workspace

Human workflow mapped attributes (formerly referred to as flex fields) store and query use case-specific custom attributes. These custom attributes typically come from the task payload values. Storing custom attributes in mapped attributes provides the following benefits:

- They can be displayed as a column in the task listing.
- They can filter tasks in custom views and advanced searches.
- They can be used for a keyword-based search.

For example the Requester, PurchaseOrderID, and Amount fields in a purchase order request payload of a task can be stored in the mapped attributes. An approver logging into Process Workspace can see these fields as column values in the task list and decide which task to access. The user can define views that filter tasks based on the mapped attributes. For example, a user can create views for purchase order approvals based on different amount ranges. If the user must also retrieve tasks related to a specific requester or a purchase order ID, the user can specify this in the keyword field and perform a search to retrieve the relevant tasks.

For the mapped attributes to be populated, an administrator must create mapped attribute mappings, as follows:

- 1. Specify a label for the mapped attribute to be populated.
- **2.** Map the payload attribute that contains the data to the label.

These mappings are valid for a certain task type. Therefore, each task type can have different mapped attribute mappings. After the mapping is complete and a new task is initiated, the value of the payload is transferred to the mapped attribute that has just been mapped. Tasks initiated before the mapping do not contain the value in the mapped attribute. Only top-level simple type attributes in the payload can be transferred to a mapped attribute. Complex attributes or simple types nested inside a complex attribute cannot be promoted. It is important to define the payload for a task in the Human Task Editor, keeping in mind which attributes from the payload may must promoted to a mapped attribute. All text and number mapped attributes are automatically included in the keyword-based search.

Essentially, the Human Task Editor is used only when defining the payload for a task. All other operations are performed at runtime.

Directory naming is not available concomitantly with the flex file naming convention.

Note: Mapped attributes must be defined before instances of the business process are generated. Only instances generated after mapped attributes are created reflect the correct mapped attributes. Older instances of the business process do not reflect subsequent mapped attribute changes.

For more information about how to specify mapped attributes, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

9.4.1 How to Specify Mapped Attributes

An administrator, or users with the necessary privileges, can use mapped attributes, shown in Figure 9–17, to transfer data from the payload to inline mapped attributes. By promoting data to mapped attributes, the data becomes searchable and can be displayed as columns on the task list.

Administrators and users with the appropriate privileges can map both public and protected mapped attributes. They see both a **Public Flex Fields** node and a **Protected Flex Fields** node in the **Administration** panel as shown in Figure 9–17.

ORACLE' Business P	Process Workspace 💀 Home Administration Preferences Hob Logout weblogic \
Administration Areas	Flex Field Mapping : Public
Organization Roles Calendars Organization Units Parametric Roles Extended User Properties	Vew existing flex field mappings (i.e. task attribute labels). Create additional flex field labels and map these to task attributes. The labels will be displayed to the end users, and should be user-friendly terms for the task attributes. © Browse all mappings © Edit mappings by task type:
Flex Fields Public Flex Fields Protected Flex Fields Task Administration Approval Groups Endone Search Task Configurations Application Preferences Application Preferences	Show: All Attributes

Figure 9–17 Specifying a Mapped Attribute

If you are a process designer and want more information about public and protected mapped attributes, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*.

To create labels: To create a mapped attribute mapping, an administrator first defines a semantic label, which provides a more meaningful display name for the mapped attribute. Click **Add** to use the Create Label dialog box shown in Figure 9–18.

Figure 9–18 Creating a Label

Create Label		×
Attribute Type:	Text Attributes	
Task Attribute:	TextAttribute1	
* Label Name:		
Description		
		_
	Create <u>C</u> ance	

As the figure shows, **labelName** is mapped to the task attribute **TextAttribute1**. The payload attribute is also mapped to the label. In this example, the **Text** attribute type is associated with **labelName**. The result is that the value of the **Text** attribute is stored in the **TextAttribute3** column, and **labelName** is the column label displayed in the user's task list. Labels can be reused for different task types. You can delete a label only if it is not used in any mappings.

A mapped payload attribute can also be displayed as a column in a custom view and used as a filter condition in both custom views and workflow rules. The display name of the payload attribute is the attribute label that is selected when doing the mapping.

Note the following restrictions:

- Only simple type payload attributes can be mapped.
- A mapped attribute (and thus a label) can be used only once per task type.
- Data type conversion is not supported for the number or date data types. For example, you cannot map a payload attribute that is assigned the string datatype to a label that is assigned the datatype number.

To browse all mappings:

- 1. Click Browse all mappings.
- **2.** Select a row in the label table to display all the payload attributes mapped to a particular label.

To edit mappings by task type:

- 1. Click Edit mappings by task type, optionally provide a task type, and click Search.
- 2. Select a task type and click OK.
- 3. With the task type displayed in the Edit mappings by task type field, click Go.
- 4. Select a mapping label and click Select.

9.5 Using Task Configuration in Process Workspace

Task Configuration is a web-based application in Process Workspace that enables business users and administrators to review and modify rules that were predefined by the workflow designer. These predefined rules can be changed for a specific customer based on the customer's applicable corporate policies. For example, suppose that a corporate policy requires two levels of approvals for expense amounts greater than 1000. Suppose further that this policy is changed to require three levels. You can use Task Configuration to change the rule rather than having your IT department modify the rule in the underlying process and then deploy it again. Any change to the rule is applied starting with the next instance, and instances already in progress use the current rule definitions.

This section contains these topics:

- Section 9.5.1, "About Task Configuration in Process Workspace"
- Section 9.5.2, "How to Edit Workflow Task Settings"
- Section 9.5.3, "How to Specify Routing Settings"
- Section 9.5.4, "How to Specify an Expiration Policy"
- Section 9.5.6, "How to Specify an Escalation Policy"
- Section 9.5.5, "How to Extend an Expiration Policy Period"
- Section 9.5.7, "How to Specify Participant Notification Preferences"
- Section 9.5.8, "How to Enable Task Access"
- Section 9.5.9, "How to Specify Actions for Acting Upon Tasks"
- Section 9.5.10, "How to View Settings for Rules"
- Section 9.5.11, "How to Add a Rule to a Ruleset in Process Workspace"
- Section 9.5.12, "How to Add and Customize Decision Tables"
- Section 9.5.13, "How to Add a Variable to a Rule"

9.5.1 About Task Configuration in Process Workspace

Task Configuration enables you to edit the event driven and data-driven rules associated with an approval flow at run time—that is, when the workflow has already been deployed.

To access the Task Configuration page:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- 2. In the Administration Panel, under Task Administration, click **Task Configurations**. The Task Configuration page appears in the right pane. To edit the task, click **Edit task**. The task becomes editable, as illustrated in Figure 9–19.

Figure 9–19 Task Configuration: Editable Task Page

ORACLE' Busi	ness Process Workspace		Gi Ho
Administration Areas	Section 2 Tasks to be configured ✓	Task Rules	
Organization	Search *	SimpleApprovalTask : Event Driven Configuration	
Roles	👔 SimpleApprovalTask (1.0)	Task Assessment Name	
Calendars			
Organization Units			
Parametric Roles		Allow all participants to invite other participants	
Extended User Properties		Allow participants to edit future participants	
Flex Fields		V Allow initiator to add participants	
Public Flex Fields		Assignment and Routing Policy	
Protected Flex Fields		Complete task when participant chooses	
Task Administration		Complete name tasks of early completion subtasks	
Approval Groups		Somprises parality administry comprising administra	
Evidence Search		✓ Expiration and Escalation Policy	
Task Configurations		Never Expire	
Application Preferences			
Application Preferences		v Natification Cattings	
		May May	
		up dow Task Status Recipient Notification Hi	
		Assign Assignees	
		△ ▼ Complete Initiator Make notifications secure (exclude details)	
		▲ ♥ Error ♥ Owner ♥ Ø Make notification actionable	
		Don't send multiple notifications for the same hum	an task event

The **Tasks to be configured** section in the middle pane lists all workflow tasks configured to use approval-flow rules. It also provides a search capability. When you click a link to one of the tasks, the corresponding data appears in the right pane in two tab pages: Task, shown in Figure 9–19, which displays the default configuration for the task, and Rules, shown in Figure 9–20, which you can use to override the approval-flow list builder configuration. The rule configurations are displayed graphically based on the stages defined in the approval flow.

Figure 9–20 Task Configuration: Rules Tab Page

s Search * 🕑	Configure Task Approval Rules	Global RulesSelect 💉 Add variable
ndars (F Humantaski (1.0) initation Units (F Humantaski (1.0) metric Roles (F Humantaski (1.0) metric Roles (F Humantaski (1.0) initation (1.0) ields (F Humantaski (1.0) Administration (F Humantaski (1.0) sector Flock (F Humantaski (1.0) sector Flock (1.0) sector Flock (1.0) sector Flock (1.0) islaministration (F Humantaski (1.0)	Click on the participant to edit rules	Stage1

9.5.2 How to Edit Workflow Task Settings

This section contains information about workflow task settings (task metadata).

To edit a task setting:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** To search for a particular task, enter all or art of the task name, or enter an asterisk (*) as a wildcard, then click Search Task type. The search results appear in the **Tasks to be configured** pane.

From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.

- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- 5. Make your changes and click either **Commit task** to apply the changes or **Save**.

Note: An improper or incomplete rules definition in a list-creation rule set can cause runtime errors. Errors can be caused by the following:

- No rule was defined in the rule set.
- None of the conditions defined in the rule was met.

Ensure that rules are properly defined to handle all conditions.

9.5.3 How to Specify Routing Settings

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.

The Task tab page contains a limited set of the routing options.

Approval aggregation requirements can be any of the following:

- None
- Once per task
- Once per stage

9.5.4 How to Specify an Expiration Policy

1. From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.

- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the Task tab page, in the Expiration and Escalation Policy section, select one of the following:
 - Never Expire
 - Expire After. Select **Fixed Duration** to specify the amount of days, hours, and minutes for completing the task. Alternatively, select **By Expression** to dynamically enter a value as an XPath expression.

9.5.5 How to Extend an Expiration Policy Period

You can extend the expiration period when the user does not respond within the allotted time. You do this by specifying the number of times the task can be renewed upon expiration (for example, renew it an additional three times) and the duration of each renewal (for example, three days for each renewal period).

To extend an expiration policy period:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click **Edit task**. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the Task tab page, in the Expiration and Escalation Policy section, select **Renew Once**.
- 6. Specify the maximum number of times to continue renewing this task.

9.5.6 How to Specify an Escalation Policy

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click **Edit task**. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the Task tab page, in the Expiration and Escalation Policy section, select **Escalate Once**.
- **6.** Specify the following additional values. When both are set, the escalation policy is more restrictive.

Maximum Escalation Levels

Number of management levels to which to escalate the task. This field is required.

Highest Approver Title

The title of the highest approver (for example, self, manager, director, or CEO). These titles are compared against the title of the task assignee in the corresponding user repository. This field is optional.

The escalation policy specifies the number of times the task can be escalated on expiration and the renewal duration.

9.5.7 How to Specify Participant Notification Preferences

Notifications indicate when a user or group is assigned a task or informed that the status of the task has changed. Notifications can be sent through email, voice message, instant message, or SMS. Notifications are sent to different types of participants for different actions. Notifications are configured by default with default messages. For example, a notification message is sent to indicate that a task has completed and closed. You can create your own or modify existing configurations.

Note: Embedded LDAP does not support group email addresses. Therefore, when a task is assigned to a group ID, emails are sent to all of its members instead of to the group email address.

9.5.7.1 How to Make Email Messages Actionable

To make email messages actionable:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the **Notification Settings** section, select **Make notification actionable**. This action enables you to perform task actions through email.

Note: FYI tasks are not actionable and cannot be acknowledged from email messages.

9.5.7.2 How to Secure Notifications to Exclude Details

To secure notifications, make messages actionable, and send attachments:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.

- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In Task tab page, in the **Notification** section, select **Make notifications secure** (exclude details).

If selected, a default notification message is used. There are no HTML worklist task details, attachments, or actionable links in the email. Only the task number is in the message.

9.5.7.3 How to Set Up Reminders

You can send task reminders, which can be based on the time the task was assigned to a user or the expiration time of a task. The number of reminders and the interval between the reminders can also be configured.

To set up reminders:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- 5. In the Notification Settings section, select the number of reminders to send.
- **6.** If you selected to remind the assignee one, two, or three times, select the interval between reminders, and whether to send the reminder before or after the assignment.

9.5.7.4 How to Notify Recipients of Changes to Task Status

Three default status types display in the **Task Status** column: **Assign**, **Complete**, and **Error**. You can select other status types for which to receive notification messages.

Notifications can be sent to users involved in the task in various capacities. When the task is assigned to a group, and there is no notification endpoint available for the group, then each user in the group is sent a notification.

To notify recipients of changes to task status:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- 5. In the Task tab page, navigate to the Notification Settings area.
- 6. In the Task Status column, select a type to display the complete list of task types:
 - Assign

When the task is assigned to users or a group. This captures the following actions:

- Task is assigned to a user
- Task is assigned to a new user in a serial workflow
- Task is renewed
- Task is delegated
- Task is reassigned
- Task is escalated
- Information for a task is submitted
- Complete
- Error
- Expire
- Request Info
- Resume
- Suspend
- Update
 - Task payload is updated
 - Task is updated
 - Comments are added
 - Attachments are added and updated
- Update Outcome
- Withdraw
- All Other Actions
 - Any action not covered in the above task types. This includes acquiring a task.
- **7.** In the **Recipient** column, click an entry to display a list of possible recipients for the notification message:
 - Assignees

The users or groups to whom the task is currently assigned.

Initiator

The user who created the task.

Approvers

The users who have acted on the task up to this point. This applies in a serial participant type in which multiple users have approved the task and a notification must be sent to all of them.

Owner

The task owner

Reviewer

The user who can add comments and attachments to a task.

9.5.7.5 How to Edit the Notification Message

A default notification message is available for delivery to the selected recipient. You can modify the default message text.

To edit the notification message:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- 5. In the Task tab page, navigate to the Notification Settings area.
- **6.** In the **Notification Header** column, click the **Edit** icon to display the default notification message.

This message applies to all the supported notification channels: email, voice, instant messaging, and SMS. Email messages can also include the worklist task detail defined in this message. The channel by which the message is delivered is based upon the notification preferences you specify.

- 7. Modify the message wording as necessary.
- 8. Click OK.

9.5.7.6 How to Prevent Multiple Email Notifications for a Human Task Event

To prevent multiple email notifications for a human task event:

- 1. From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- 2. In the Administration Panel, under Task Administration, click Task Configurations.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the **Notification Settings** section, select **Don't send multiple notifications for the same human task event**. This action enables you to perform task actions through email.

9.5.8 How to Enable Task Access

Access-rule settings can be set to control the actions a user can perform. Content and action permissions can be specified based on the logical role of a user, such as creator (inititator), owner, assignee, and reviewers.

To enable task access:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the Task tab page, expand the Task Access section at the bottom of the page, then select the Task Content tab.
- Assign privileges (read, write, or no access) to users to act upon task content. A user cannot be assigned a privilege above their highest level. For example, an ADMIN user cannot be assigned write access on the PAYLOAD task content. Table 9–5 shows the maximum privilege each user has on task content.

Task Content	Individual with Read Access	Individual with Write Access
Assignees	Admin, Approvers, Assignees, Creator, Owner, Reviewers	
Attachments	Admin, Approvers	Assignees, Creator, Owner, Reviewers
Comments	Admin, Approvers	Assignees, Creator, Owner, Reviewers
Dates	Admin, Approvers, Assignees, Creator, Owner, Reviewers	
Flexfields	Admin, Approvers, Reviewers	Assignees, Creator, Owner
History	Admin, Approvers, Assignees, Creator, Owner, Reviewers	
Payload	Admin, Approvers, Reviewers	Assignees, Creator, Owner
Reviewers	Admin, Approvers, Assignees, Creator, Owner, Reviewers	
Payload elements	Inherited from payload	Inherited from payload

Table 9–5 Highest Privilege Levels for Users of Task Content

Note: Access rules are always applied on top of what the system permits, depending on who is performing the action and the current state of the task.

9.5.9 How to Specify Actions for Acting Upon Tasks

You can specify the actions that specific users can perform when acting on the task content that you specified in the Configure Task Content Access dialog in Section 9.5.8, "How to Enable Task Access."

To specify actions for acting upon tasks:

1. From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.

- **2.** In the Administration Panel, under Task Administration, click **Task Configurations**.
- **3.** From the **Tasks to be configured** pane, select a task. The corresponding Task tab page appears in the right pane.
- **4.** In the **Tasks to be configured** pane, click the **Edit task** icon. The Task tab page becomes editable as shown in Figure 9–19.
- **5.** In the Task tab page, expand Task Access at the bottom of the page, and select the Task Actions tab.
- 6. Select the task action for which to specify users.
- 7. Specify whether participants can or cannot perform the selected actions.

9.5.10 How to View Settings for Rules

To view a rule set for a rule-based participant:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configuration**. The Tasks to Be Configured page appears in the middle pane.
- **3.** In the Tasks to Be Configured page, select a rule whose settings you want to modify. The corresponding details appear in the Tasks tab page in the right pane.
- 4. Select Edit task. The right pane refreshes in edit mode.
- **5.** In the right pane, select the Rules tab. The Rules tab page displays the Configure Task Approval Rules page, similar to what is shown in the example in Figure 9–21.
- **6.** Click the **Participant** link to navigate to the Rules page where rules can be added, modified, or deleted.

Administration Areas	Tasks to be configured	/ 🥱 🗟 😒	Task Rules	
Organization	Search *		🗟 Configure Task Approval Rules	Global RulesSelect 🗸 Add variable
Roles	PoCompositeTask1 (1.0)		Click on the participant to edit rules	
Calendars	SimpleApprovalTask (1.0)			â
Organization Units				
Parametric Roles				
Extended User Properties				Hander
lex Fields				l
Public Flex Fields				•
Protected Flex Fields				888 Participant1
Task Administration				Final
Approval Groups				
Evidence Search				
Task Configurations				age Participant1
Application Preferences				
Application Preferences				±

Figure 9–21 Editing a Task Rule

The Configure Task Approval Rules page graphically shows the approval flow for the task. The icon in each participant node indicates the type of participant: single, parallel, serial, or FYI.

To see a tool tip displaying the name of the ruleset for a participant in this task, hover over the link for that participant.

Note that value- based participants are shown but are disabled.

9.5.11 How to Add a Rule to a Ruleset in Process Workspace

You can add a rule to a ruleset, specifying not only conditions, but also a description, the effective date, and priority.

To add rules for a rule-based participant:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configuration**. The Tasks to Be Configured pane appears as the middle pane.
- **3.** In the Tasks to Be Configured pane, select a rule whose settings you want to modify. The corresponding details appear in the Tasks tab page in the right pane.
- 4. Select Edit task. The right pane refreshes in edit mode.
- **5.** In the right pane, select the Rules tab. The Rules tab page displays the Configure Task Approval Rules page showing the approval flow for the task. The icon in each participant node indicates the type of participant: single, parallel, serial, or FYI.
- **6.** To edit the rules, click the link for the participant. The rules editor page appears with the participant node (ruleset) selected. Here you can either add or edit rules defined for the rule set. (You can also add or edit decision tables as explained in Section 9.5.12, "How to Add and Customize Decision Tables".)
- 7. Click Add Rule as shown in Figure 9–22.

Tasks to be configured	/ 🔊 🖯 🖾	Task Rules		
Search *		Back to the Participant Tree		
Search * () Humantaski (1.0) SingleApprovallaski (1.0) ruleTaskAppStask (4.0)		Back to the Participant Tree (X) Globals	¥ rule1 View [F/THEN Rules ♥ ♣ ♥ ♥ ※ ۞ 1-1 of 1 ♥ ۞ ● Rule1 ● ₩ ○ ♥ № □ ▲dd Rule]Ignore this participant
			1	

Figure 9–22 Adding a Rule

8. To specify the rule, click the **Expand** icon to the left of the rule field. This enables you to specify the conditions for the rule as described in Section 9.5.11.1, "How to Define Rule Conditions". To provide a description and specify the effective date and priority, click the **Show Advanced Settings** icon to the left of the rule field.

9. When you have finished making your changes, in the Tasks to Be Configured pane, click **Save**. The changes are saved to the rule definitions in the rules dictionary.

To commit the changes to MDS, click **Commit**. The task is refreshed and no longer editable.

For more information about adding rules, see the chapter on "Working with Rulesets and Rules" in *Oracle Fusion Middleware User's Guide for Oracle Business Rules*.

9.5.11.1 How to Define Rule Conditions

You can set the left and right sides of a condition by selecting operands from condition browsers. Clicking the magnifying glass icon displays the browsers.

The operator for comparing the operands of the condition change based on the type of operand selected for the left side of the condition.

You also can define more complex conditions using the Expression Builder.

For more information, see the section "Creating ADF Data Binding EL Expressions" in *Oracle Fusion Middleware Fusion Developer's Guide for Oracle Application Development Framework*. See also the section "Creating EL Expressions" in Oracle Fusion Middleware *Web User Interface Developer's Guide for Oracle Application Development Framework*.

9.5.11.2 How to Add Actions

You can add actions to a rule. To add an action:

- 1. From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- 2. In the Administration Panel, under Task Administration, click **Task Configuration**. The Tasks to Be Configured pane appears as the middle pane.
- **3.** In the Tasks to Be Configured pane, select a rule. The corresponding details appear in the Tasks tab page in the right pane.
- 4. Select Edit task. The right pane refreshes in edit mode.
- **5.** In the right pane, select the Rules tab. The Rules tab page displays the Configure Task Approval Rules page showing the approval flow for the task. The icon in each participant node indicates the type of participant: single, parallel, serial, or FYI.
- **6.** To add an action, click the link for the participant. The rules editor page appears with the participant node (ruleset) selected.
- 7. From the View drop-down field, select IF/THEN Rules.
- **8.** If there are no rules defined, add a rule as described in Section 9.5.11, "How to Add a Rule to a Ruleset in Process Workspace"

If there are rules defined for the ruleset, go to step 9

9. Expand the rule as shown in Figure 9–23.

Figure 9–23 Adding Actions to a Rule

🕑 Tasks to be configured 🦉 🗐 🔄 🔯	Task Rules	
Search *	Back to the Participant Tree	
SampleMultilistbuldersTask (1.0)		Japore this participant
Ssoch * (2) (2) SamplethuthistbuldersTask (1.0) (2) SmplekoprovaTask (1.0)	Back to the Participant Tree (A) Globals Bucksteets Bucksteets Consistent Participant Bucksteets SubstitutionRules SubstitutionRules SubstitutionRules SubstitutionRules Johniset4	Ignore this participant Very Prhetistudes ■ ♦ • X 10 f ■ Very Prhetistudes ■ ♦ • X 10 f ■ Very Prhetistudes ■ ♦ • X 10 f ■ Very Prhetistudes ■ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
		4

10. Under the THEN pane, click the add icon to add an action.

Select the action you want to add from the displayed list as shown in Figure 9–24.

Figure 9–24 Available Actions for Rules



You can add multiple actions for any rule by repeating step 10.

- **11.** After adding a rule, you can perform the following commands:
 - Delete
 - Copy
 - Cut
 - Paste
 - Up and Down (to change the order of the listed actions)

To perform commands on an action:

1. Select the action by clicking the check box next to the action as shown in Figure 9–25.

Figure 9–25 Delete, Up, Down, Copy, Cut and Paste Commands

THEN					 		 	 	 	-			
🗣 💌 🛠 🖓 👶 😹 🛅 🛍	THEN												
Z call a Calendar () 10	4	- 🗙 🕆	Ð	ж	C	1							
						_							

- **2.** Select any of the possible actions such as **Delete**, **Cut**, **Copy** or **Paste** by clicking the corresponding icons.
- **3.** If you have multiple actions, you can change the order of the actions by using the up and down arrows.

9.5.12 How to Add and Customize Decision Tables

You can customize existing decision tables or add new ones.

9.5.12.1 How to Customize an Existing Decision Table in Process Workspace

You can specify the actions for a ruleset that were specified in a previously created decision table.

To customize an existing decision table in a ruleset in Process Workspace:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configuration**. The Tasks to Be Configured pane appears as the middle pane.
- **3.** In the Tasks to Be Configured pane, select a rule whose settings you want to modify. The corresponding details appear in the Tasks tab page in the right pane.
- 4. Select Edit task. The right pane refreshes in edit mode.
- **5.** In the right pane, select the Rules tab. The Rules tab page displays the Configure Task Approval Rules page showing the approval flow for the task. The icon to the left of each participant in the flow indicates the type of participant: single, parallel, serial, or FYI.
- **6.** To edit the rules, click the link for the participant. This displays the ruleset associated with that participant.
- **7.** From the View list, select the decision table you want to customize. The selected decision table appears in the lower half of the page.
- **8.** Edit any of the existing rows in the decision table.
- **9.** To add an action, in the decision table, from the **Add** list, select the action you want to add as shown in the example in Figure 9–26.

Figure 9–26 Adding Actions to a Decision Table

🗹 Tasks to be configured 🥒 🖗 🗔 民	Task Rules				
Search *	Back to the Participant Tree				
Aask to be configured A configu	Task Rules Back to the Participant Tree (x) Globals ② Bucketsets ③ Decision Functions ③ Translations Rulesets ③ SubstitutionRules ③ Johntiset4	dellaneSHSet View Decision DecisionTable 1	ontrable I	Double dick to edit or R2 2000	els
		Resource Position	¥		=
		Users Resource	"fkafka"	"ficafica"	
		Groups Supervisory	nul	nul	
		Application Role	null	nul	
		Response Type	ResponseType.REQUIRED	ResponseType.REQUIRED	
		Rule Name	"rip"	"rip"	
		Lists	Lists		
		Supervisory	¥		
		Number of levels	4		
					•
ovright @ 2005, 2013, Oracle and/or its affiliates, All rights n	eserved.				

The corresponding actions appear in the table for you to specify.

9.5.12.2 How to Edit an Action (Multiple Parameters: All Rules at a Time)

In the Rules tab, you can edit multiple parameters of an action from the Decision Table. To edit multiple parameters:

1. From the View list, select the decision table that contains the action you want to edit. The selected decision table appears in the lower half of the page as shown in Figure 9–27.

Figure 9–27 Decision Table with Editable Actions

✓ Tasks to be configured / ♥ C La	Task Rules					
Search *	Back to the Participant Tree					
記 Sample/HultilistbuidersTask (1.0) 記 SimpleApprovalTask (1.0) 記 Tashfame (1.0)	(x) Globals 좋? Bucketsets @ Links Q Decision Functions Caranalations	 ruleBaseSrlS DecisionTa 	iet View Dec	sonTable1 💌 🔌		Ignore this participant
	Rulesets					
	SubstitutionRules					
	vileBaseSrlSet					
	🚱 Johiset	Conditions	Image:	Construction	Double dick to edit of R2 2000	els
			Groups	nul	nul	
			Application Role	null		
			Response Type	response rype.kc.Quikco	Response Type. REQUIRED	
			Liete	rip	rp	
			Supervisory			
			Number of levels	4		

2. Select the action you want to edit and click Edit Action as shown in Figure 9–28.

₽ - ∥	× # # & Ø	0 Local List of Ranges	s 🚽 🖉 🛛 Double click to edit o	ells
ſ	Edit Action	R1	R2	
Conditions	Task.payload.process.salary	1000	2000	
Conflict Re	solution			
Actions	Resource	✓		
	Users	"fkafka"	"fkafka"	
	Groups	null	null	

Figure 9–28 Editing an Action from the Decision Table

3. An Action Editor dialog box appears as shown in Figure 9–29. You can modify the values available for the action you selected. You can choose not to specify values for all the parameters in the Action Editor dialog box.

Figure 9–29 Action Editor

Action Editor		×
Users	"fkafka"	Q
Groups	null	Q
Application Role	null	Q
Response Type	💿 Required 💿 FYI	
Rule Name	"rip"	
Lists		
	OK Can	cel

4. Click OK.

The values appear in the Decision Table for the selected action for all rules. For more information on Decision Tables, see Section 9.5.12, "How to Add and Customize Decision Tables".

9.5.12.3 How to Edit an Action (Single Cell Editing)

You have to option to edit one of the parameters of an action in the Decision Table. To edit a single cell of any action, you must perform the following steps:

- 1. From the View list, select the decision table that contains the action you want to edit. The selected decision table appears in the lower half of the page as shown in Figure 9–27.
- **2.** In the Decision Table, click the cell you want to modify. This makes the cell editable as shown in Figure 9–30.

+ - /	🛛 🛛 🖬 🛤 🛛 🕖	•	Local List of Ranges	-	Double click to edit c	ells
		R1		R2		
Conditions	Task.payload.process.salary		1000		2000	
Conflict Res	solution					
Actions						
Actions	Resource		V			
	Users	"fkafka"	null	<u> </u>	'fkafka"	
	Groups	null			null	
	Application Role		null		null	

Figure 9–30 Single Cell Editing

3. Update the parameter with the specific value/option.

9.5.12.4 How to Add a New Decision Table to a Ruleset in Process Workspace

You can add a table to a ruleset and specify the actions it contains.

To add a new decision table to a ruleset in Process Workspace:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Panel appears in the left pane.
- **2.** In the Administration Panel, under Task Administration, click **Task Configuration**. The Tasks to Be Configured pane appears as the middle pane.
- **3.** In the Tasks to Be Configured pane, select a rule whose settings you want to modify. The corresponding details appear in the Tasks tab page in the right pane.
- 4. Select Edit task. The right pane refreshes in edit mode.
- **5.** In the right pane, select the Rules tab. The Rules tab page displays the Configure Task Approval Rules page showing the approval flow for the task. The icon to the left of each participant in the flow indicates the type of participant: single, parallel, serial, or FYI.
- **6.** To edit the rules, click the link for the participant. This displays the ruleset associated with that participant.
- **7.** In the ruleset, from the **View** list, select Add Decision Table as shown in Figure 9–31.

Figure 9–31 Adding a Decision Table

🗹 Tasks to be configured 🛛 🖉 🕞 🗔	Task Rules	
Search *	Back to the Participant Tree	
 Humantaski (1.0) SimpleApprovalTask (1.0) ruleTaskApp3task (4.0) 	□tgrow Globals Bucketsets Decision Functions Rulesets ModificationRules SubstitutionRules SubstitutionRules	re this participant

The new decision table appears in the lower half of the page.

8. To add an action, in the decision table, from the **Add** list, select the action you want to add as shown in the example in Figure 9–26. The corresponding actions appear in the table for you to specify.

9.5.13 How to Add a Variable to a Rule

To add a variable to a rule:

- 1. In the Rules page, click Add variable. The Add Variable window appears.
- **2.** Enter a name for the variable and, optionally, a description.

3. From the Type list, select a variable type.

The types displayed in the list correspond to those that are available in the rule dictionary (built-in by default as well as others that have been registered).

- 4. Enter a value.
- 5. Click OK.

The variable can now be used to define conditions.

9.6 Using Evidence Search in Process Workspace

The evidence store service is used for digital signature storage and nonrepudiation of digitally signed human tasks.

To search the evidence store:

- **1.** In the Process Workspace toolbar in the upper right corner, click **Administration**. The Administration Areas panel appears in the left pane.
- **2.** In the Administration Areas panel, under Task Administration, select **Evidence Search**. The Evidence Search page appears in the right pane, as shown in Figure 9–32.

ORACLE Business Process Workspace Home | Administration | Preferences | Help | Logout | weblogic 으 Administration Areas Search Evidence Store Organization Match 💿 All 🔘 Any Roles 🖄 Signed Before 🖏 🕟 Signer 🔍 Signed After 🛛 Calendars ⊳Advanced Organization Units Parametric Roles Validate Extended User Properties Creation Date Signed Date Verified Date Signature Policy Signer Status Flex Fields Public Flex Fields Protected Flex Fields Task Administration Approval Groups Evidence Search Task Configurations Application Preferences Application Preferences Evidence Details Please select an evidence to view the details Copyright © 2005, 2012, Oracle and/or its affiliates. All rights res

Figure 9–32 Searching the Evidence Store

3. Fill in the search fields, then click Search Evidence Store.

For information about the evidence store for digital signatures, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

9.7 Managing Exceptions in Process Workspace

Under certain circumstances—for example, an employee who has left the company being assigned to a task—an exception can occur. When an exception occurs, the task gets put into an alerted state. The task then gets assigned to the error assignee or, if one is not specified, to the administrator.

For information about configuring the error assignee, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

There are other scenarios in which an exception can happen.

9.8 Using the User Metadata Migration Utility

The user metadata migration utility, hwfMigrator, is a tool that automates the process of migrating workflow user-configurable data from one Service Integration server to another by executing a shell script. The tool also includes a property file that contains key-value pairs and all the input parameters required to perform the migration operation. You customize the property file and then perform the migration by executing the shell script.

If you are a process designer and want more information about using the User Metadata Migration Utility, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*.

For information about moving Oracle BPM from a test site to a production site, see *Oracle Application Server Administrator's Guide*.

Managing Your Organization in Process Workspace

This chapter describes how administrators of Oracle Business Process Management Suite can model an organization by defining organization units, business calendars, business holidays, roles, and other user properties.

This chapter contains these topics:

- Section 10.1, "Understanding Deployment of Organization Entities in Process Workspace"
- Section 10.2, "Managing Holiday Rules in Process Workspace"
- Section 10.3, "Managing Calendar Rules in Process Workspace"
- Section 10.4, "Managing Other Users' or Groups' Rules (as an Administrator) in Process Workspace"
- Section 10.5, "Managing Roles in Process Workspace"
- Section 10.6, "Managing Parametric Roles in Process Workspace"
- Section 10.7, "Managing Extended User Properties in Process Workspace"
- Section 10.8, "Managing Organization Units in Process Workspace"

10.1 Understanding Deployment of Organization Entities in Process Workspace

For information about organization units, see Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management.

When a process is deployed, the various organization entities—for example, the organization unit, calendar, and so on—are created. Those entities can then be modified only through Process Workspace. Later, when the process is redeployed, data in Oracle BPM Studio does not overwrite any changes made during run time.

10.2 Managing Holiday Rules in Process Workspace

Figure 10–1 Holiday Rules

Create holiday sets, which can be used in calendar	5			
✓ Holidays List My Holiday Rule	✓ Details : My Holiday Rule Y 2 X			Apply Reset
	Name	Type	Date	
	US Independence Day	Same Day Every Year	July 4	
	US Thanksgiving	Nth Week Day of Month	Fourth Thursday of November	
	New Years Day	Same Day Every Y	January 1	
	Memorial Day	Nth Week Day of Month	Fourth Monday of May	
	Labor Day	Nth Week Day of Month	First Monday of September	

Business holidays are collections of holidays that can be applied to calendar rules. Then, when computing the duration of a process, the calendar takes into account the specified holidays.

You can create as many holiday rules as needed for different calendar rules. For example, if you apply holiday rules for India, the United States, and China to a given calendar, then those various national holidays are taken into account when the duration of a process is computed.

When you create a business holiday rule, you specify the holiday name, date, and holiday type. Table 10–1 lists and describes the holiday types.

Holiday Type	Description			
Same Day Every Year	On the same day each year			
Current Year	In the current year only			
Nth Week Day of Month	In a specified week, on a specified day, in a specified month eac year—for example, in the United States, Thanksgiving Day is celebrated each year on the fourth Thursday of November			
Nth Day of Reference Holiday	On a day in relation to another specified holiday—for example, on the day after Thanksgiving Day			
Closest week day	The closest day to a given date—for example, in the United States, the Independence Day holiday may be celebrated on the weekday closest to the Fourth of July.			

Table 10–1 Holiday Types

To create a holiday rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- 2. In the Organization panel, click Calendars. The Calendars page appears.
- **3.** In the Holidays List section, click **Add a Holidays List**. The Add a Holidays List dialog prompts you for the name of the new holiday list.
- **4.** Enter the name of the new holidays list and click **OK**. The new holidays list appears in the holidays list.
- 5. Select the new holidays list. The details for the new list appear in the right pane.
- **6.** In the holiday details section, click **Add a Holiday**. A row appears for you to specify the attributes of the new holiday.
- **7.** Specify the name of the holiday, and the day on which the holiday occurs each year. Options for specifying holiday occurrence are described in Table 10–1.

8. Click Apply. The new holiday rule is configured according to your specifications.

To edit a holiday rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Calendars**. The Calendars panel appears in the right pane.
- **3.** In the Holidays List section, select the holiday rule you want to edit. The details page for the rule appears in the right pane.
- **4.** Change the settings, and click **Apply**. The holiday rule is now modified according to your specifications.

To delete a holiday rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Calendars**. The Calendars panel appears in the right pane.
- **3.** In the Holidays List section, select the holiday rule you want to delete. The details for the rule appears in the right pane.
- 4. Click Delete Holiday Rule. The holiday rule is now deleted.

10.3 Managing Calendar Rules in Process Workspace

Figure 10–2 Calendar Rules

⊻ Calendars 🗳 🗙 My Calendar	⊻ Details : Hy Calendar	Apply Reset
	Time Zone (GMT-8:00) Pacific Standard Time (PST8PDT) Holdav List Mv Holdav Sula 💌 🚳	V
	Starting Time Finish Time Starting Time Finish Time	
	✔ Monday 8 ⊕ 0 ⊕ 12 ⊕ 0 ⊕ ✓ 13 ⊕ 0 ⊕ 17 ⊕ 0 ⊕ ✔ Tuesday 8 ⊕ 0 ⊕ 12 ⊕ 0 ⊕ ✓ 13 ⊕ 0 ⊕ 17 ⊕ 0 ⊕ Ø Workeeday 8 ⊕ 0 ⊕ 12 ⊕ 0 ⊕ ✓ 13 ⊕ 0 ⊕ 17 ⊕ 0 ⊕	
	V Thursday 8 ± 0 ± 12 ± 0 ± V 13 ± 0 ± 17 ± 0 ± V Friday 8 ± 0 ± 12 ± 0 ± V 13 ± 0 ± 17 ± 0 ±	
	Saturday Sim Oim 12 Oim 12 Oim 17 Oim Oim Sunday 8 0 12 0 0 13 0 17 0 0 Sunday 8 0 12 0 0 13 0 17 0 0	

A business calendar defines a work pattern for each day and a work week structure for each week. In addition, a calendar also defines non-working days by using business holiday rules—that is, the business calendar and holiday rule together define the work period.

When you create a business calendar, you define the following:

- Calendar name
- Time zone
- Working days in week
- Start time and end time for each day
- Optional holiday rule

You can create as many calendars as necessary.

You can specify a business calendar for a role and organization unit association. For example, you can specify that the InsuranceAgents role in the US-California organization unit follows the US-California business calendar.

To create a calendar rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Calendars**. The Calendars page appears in the right pane.
- **3.** In the Calendars section, click **Add a Calendar Rule**. The Add a Calendar Rule dialog prompts you for the name of the new calendar rule.
- **4.** Enter the name of the new calendar rule, and click **OK**. The new calendar rule appears in the Calendars list.
- **5.** Select the new calendar rule. The details for the new rule appears in the right pane.
- **6.** Specify the following:
 - The time zone
 - The holiday rule, if any, that applies
 - The appropriate starting and finishing times
- **7.** Click **Apply**. The new calendar rule is now configured according to your specifications.

To edit a calendar rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Calendars**. The Calendars page appears in the right pane.
- **3.** Select the calendar rule you want to edit. The details page for the new rule appears in the right pane.
- **4.** Change the settings, and click **Apply**. The calendar rule is now modified according to your specifications.

To associate a calendar rule with a role and an organization unit:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Roles**. The Roles panel displays a list of the roles you are authorized to administer.
- **3.** In the Details panel, in the Calendars section, click **Associate a new calendar rule to an organization unit for the application role**. Calendar and Organization Unit lists appear in the Calendar section.
- **4.** Using Calendar and Organization Unit lists, specify the calendar and organization unit to which you want to associate this role.
- 5. Click Apply.

To associate a calendar with an organization unit:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** Under Organization, click **Organization Units**. The Organization Units panel appears.
- **3.** Select the organization unit with which you want to associate a calendar rule. The details for that organization unit appear in the right pane.
- **4.** In the Details panel, from the Calendars list, select the calendar to associate with this organization unit.
- 5. Click Apply.

To delete a calendar rule:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Calendars**. The Calendars panel appears in the right pane.
- **3.** In the Calendars section, select the calendar rule you want to delete. The details for the rule appears in the right pane.
- 4. Click Delete Calendar. The calendar is now deleted.

10.4 Managing Other Users' or Groups' Rules (as an Administrator) in Process Workspace

Administrators can make changes to rules. This can be useful for correcting a problem with a rule. For example, for a user who no longer works for the company, you can set up a rule automatically assigning that user's tasks to another user or group.

To create a rule for another user or group:

1. From the Process Workspace toolbar, click **Preferences**. The Preferences pane appears on the left.

In the User Preferences pane, click Rules.

- **2.** Click the **Other Rules** tab.
- **3.** Search for the user or group for whom rules are to be created, as shown in Figure 10–3.

Figure 10–3 Creating Rules for Another User or Group

My Ri	ules Other Rules
Rule	s マ △ + -
Use	er 🔽 jstein 📎 Go
jst	tein's Rules In's Group Rules CoanAnalyticGroup WesternRegion Supervisor Supervisor StaternRegion

4. Click a user rules node, or click a group name (for a group rule).

- 5. Click the Add icon to create a rule.
- 6. Provide rule information, as shown in Figure 10–4, and click Save.

Figure 10–4 Defining Rules for Another User or Group

ORACLE' Business F	rocess Workspace		🝓 Home Administration Preferences Help Logout weblogic 으
Preferences	My Rules Other Rules		
User Preferences	✓ Rules	▽△ᅷ※	Group Rules Save Revert
Certificates Rules Notification Preferences	My Group 💌 *	Show Rules	Name * Group Rule
Accessibility	⊽ 🖏 * � Group Rule	R.	Start Date 6
			TASKS Tasks matching these conditions 💌
			Match 🔿 All 🕘 Any 💠
			ACTION
			Assign to member via: Least Busy O Assign to: User O Take no action
		1	Reassigned task access is determined according to new assignee rights. Delegated task access is determined according to rights of original user who delegates.

10.5 Managing Roles in Process Workspace

Roles are created as application roles under the application OracleBPMProcessRolesApp. There are two types:

 Swimlane roles: Each swimlane role in a BPM process is created during design time in Oracle BPM Studio. It is then mapped to an application role that is created during deployment. This mapping cannot be changed after deployment.

Members for these application roles can be defined and updated before deployment by using Oracle BPM Studio or after deployment by using Process Workspace.

In this release, swimlanes define the default task assignee and how to initiate a process if the initiator task pattern is used. During runtime, only members of those roles can perform actions such as viewing and acting on tasks and initiating processes.

 Application roles: These represent any roles in the organization, and they can be created in addition to the swimlane roles defined during design time. An application role can be used as a task assignee or as a grantee of another application role. They can be created by using either Oracle BPM Studio or Process Workspace.

When you create a role, you define both the role name and the grantees of the role. The grantees can be users, groups, or other application roles.

If users are logged in when you grant them a new role, then they must log out and log back in to exercise the newly granted role.

10.5.1 How to Add New Application Roles

To add a new application role:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Roles**. The Roles panel displays a list of the roles you are authorized to administer.
- **3.** In the upper-right corner of the Roles panel, click **Add a new role**. The Add Role dialog box appears.
- **4.** Specify a name, and, optionally, a description for the new role. Click **OK**. The new role is now listed in the Roles panel.

Note: It is recommended that you do not include spaces in role names.

- **5.** To grant this role to users, groups or application roles, follow the steps in Section 10.5.3, "How to Grant and Revoke Roles".
- **6.** To associate a calendar and its corresponding organization unit with this new role, in the Calendar section:
 - **a.** Select Associate a new calendar rule to an organization unit for the application role.
 - **b.** From the Calendar list, select the calendar you want to associate with this new role.
 - **c.** From the Organization Unit list, select the unit that you want to associate with the calendar you just specified.

Note: When a calendar is associated with an organization unit for a role, this calendar overrides the calendar associated with the calendar page when the particular role is used.

10.5.2 How to Specify the Escalation Path for a Role

An administrator can specify the path an escalated task is to follow.

For more information about the escalation path, see Section 7.8, "Handling Escalations in Process Workspace".

To specify an escalation path:

- 1. From the toolbar at the top of the Process Workspace, click **Administration**. The Administration Areas page appears in the navigation pane, **Roles** is highlighted, and the corresponding information appears in the right pane.
- **2.** If you want to set the escalation path for an existing role, select that role from the list. The corresponding information appears in the lower half of the right pane.

If you are creating a new role, click **Add a new role** and follow the instructions in Section 10.5.1, "How to Add New Application Roles" on page 10-7. The new role appears in the list, and its corresponding information appears in the lower half of the right pane.

Figure 10–5 shows the Roles window with the Escalation Path fields highlighted.

Figure 10–5 Setting the Escalation Path

rganization Name Members Description Context Roles Edenders BPMProcessAdmin Administrators BPM application admin role, has full privilege for performing any operations including security relat OracleBPMProcessAdesApp Calenders Organization Units Description Save Parametric Roles Extended User Properties Escalation Path User Save Public Flax Fields Tack assigned to this role are escalated to the specified user/group/role Calenders Save Public Flax Fields Name Type Calenders Calenders Approval Groups Name Type Calenders Save Application Preferences Application Preferences Group Organization Unit Find Administrators	inistration Areas	Roles Search	۷		
Roles Calendars Organization Units Parametric Roles Extended User Properties ex Fields Protected Flex Fields Protected Flex Fields Reviser CoracleDMM***********************************	anization	Name BPMProcessAdmin	Members Administrators	Description BPM anglication admin role, has full privilege for performing any operations inc	Context
Arganization Units arametric Roles arametric Roles stended User Properties x Fields Ubic Fields Fields sold control Fields Sate Statistication provid Groups vidence Search Sate Configurations pplication Preferences Image: Signal Definition of the preferences	les lendars	Dirit rocostiant	Prominisci acoro		and second reader in recessions pp
Administrators Iber Name Type Administrators Group Calendar Organization Unit	ganization Units rametric Roles	Details : BPMProcess	Admin		Save Re
Calendars Mame Type Administration Approval Groups addence Search ak Configurations ak Configurations ak Configurations ak Configurations ak Configurations	tended User Properties	Escalation Path User	Tasks assigned to this ro	le are escalated to the specified user/group/role	
Members Members Calendars Calendars Name Type Administration Administrators Group oproval Groups Administrators Group addace Search Search Search alkaction Preferences Search Search	rields blic Flex Fields	Context OracleBPMProc	essRolesApp		
k Administration ipped proval Groups Administrators idence Search idence Search ik Configurations idence Search jikation Preferences idence Search	tected Flex Fields	Members	N E	Calendars 🔐 💥	
proval Groups idence Search id	< Administration	Administrators	Group	Calendar Organization Onic	
idence Search sk Configurations lication Preferences lication Preferences	proval Groups				
incation Preferences	idence Search				
plication Preferences	lication Preferences				
	plication Preferences				

- **3.** From the list, select either User, Group, or Role. You may also enter or browse for a specific name.
- 4. When you have specified the escalation path, click **Apply**.

10.5.3 How to Grant and Revoke Roles

You can grant roles to, and revoke them from, users, groups, or application roles

To grant a role to a user, group, or application role:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Roles**. The Roles panel displays a list of the roles you are authorized to administer.
- **3.** Select a role. The Details panel displays the details for the selected role.
- 4. In the Members section, click Grant the role to user, group, or role.
- **5.** Specify the parameters of your search, click **Search**, then select the user, group, or application role to whom you want to grant this role. The Details section for your selection appears.
- **6.** Click **OK**. The user or group you specified now appears in the list in the Members section.
- 7. In the Details panel, click **Save**.

To revoke a role from a user, group, or application role:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Roles**. The Roles panel displays a list of the roles you are authorized to administer.
- 3. Select a role. The Details panel displays the details for the selected role.
- 4. In the Members section, click **Revoke the role from user, group, or role.**

- 5. Click OK. The user or group you specified no longer appears in the Members box.
- 6. In the Details panel, click **Save**.

Note: If, during design time, you update a process by removing a member from a role and then re-deploy the process, the member you removed is still listed as a member of that role. This is because permission to remove members from roles is limited to administrators during runtime.

10.5.4 How to Create and Manage Role-Specific Views and Layouts

For information about creating a role-specific layout, see Section 5.1, "Creating Custom Pages in Process Workspace"

10.6 Managing Parametric Roles in Process Workspace

Parametric roles are logical roles created by the administrator. Members of parametric roles are dynamically evaluated at runtime by specifying them as a parameterized query.

Parametric roles can be based only on process roles that are created by process deployment or created by using the Administration panel in Process Workspace. More specifically, they can be based on any role defined in the application context OracleBPMProcessRolesApp that is already seeded and displayed in the Roles section of the Administration panel.

Parametric roles define either members of an organization unit or the assignees of a task.

10.6.1 Parametric Roles Used for Querying Members of an Organization Unit

These parametric roles are groups of users specified by using a query—for example, all grantees of role InsuranceAgent with expertise in HomeInsurance where expertise is an extended user property.

The query can contain one or more of the following parts:

- Group membership
- Role membership
- Extended user properties from which users select from a list of pre-defined values

For information about extended user properties, see Section 10.7, "Managing Extended User Properties in Process Workspace".

Note: Deleting an extended user property already in use by an parametric role causes an error when administering that role.

10.6.2 Parametric Roles Used in Assigning Tasks

Parametric roles have one or more defined parameters that specify the type of person to whom a task can be assigned—for example, a task can be assigned as a parallel task to all users belonging to the Western region and whose skillsets include JAVA. The assignees that are returned by the dynamic parametric role can be used for any type of list builder—for example, sequential, parallel, or management chain.

The parameter values are defined at design time either as plain values or XPath expressions.

10.6.3 How to Create a Parametric Role

To create a parametric role:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- 2. Select Parametric Role. The Parametric Role panel appears in the right pane.
- **3.** In the Parametric Role panel, click **Create Parametric Role**. This opens an editable Details panel.
- 4. Enter a name for this parametric role.
- **5.** If you do not want this parametric role to be attached to a task assignee, then skip to Step c.

If you want this parametric role to be attached to a task assignee, then you must define the parameters to specify the assignee.

To define parameters for a parametric role used in assigning a task:

a. Expand **Parameters** and click **Add Parameter**. An editable table enables you to specify the name and type of the parameter as in the example in Figure 10–6.

Define the parameter by specifying the name and type. To add another parameter, click **Add Parameter** again and repeat.

ORACLE' Busines	s Process Workspace					වෙ	Home	Administration	Preferences	Help L	.ogout	weblogi
Administration Areas	Parametric Roles	🔮 🗶 De	tails : Untitled								Save	Reve
Organization	Untitled	* N	Jame New Parametric Role									
Roles												
Calendars			arameters									
Organization Units			Name Type									
Parametric Roles			1 param1 String	×								
Extended User Properties			2 param2 String	×								
Flex Fields			3 param3 String	×								
Public Flex Fields			4 param4 String									
Protected Flex Fields		Inc	lude people that meet these con	ditions :								
Task Administration		Inc	idde people that meet these tor	adons .								
Approval Groups				Add condition Description	💌 🕂						_	
Evidence Search								"Please use '\$	symbol to use	paramete	rs"	
Task Configurations		G	rantees Application Role 💌	Select	~			Q X				
Application Preferences			Name equals 🗙					×				
Application Preferences		Des	cription equals					*				
	•	•										
	Π	F					_			_		_
opyright © 2005, 2012, Oracle and/	or its affiliates. All rights reserved											

Figure 10–6 Defining Parameters

Note: You are not required to use any or all of the parameters that you define.

After a parameter is defined, the role is treated as a parametric role regardless of whether the defined parameters are used in a query.

6. From the Grantees list, select either Group or Application Roles.

If you select **Group** specify the name of the group in either of the following ways:

- Enter the name of the group in the text field
- Click Select Group to begin a search. The Select Group dialog box appears.
 Specify your search, or use an asterisk (*) as a wildcard, then click OK.

The group you specified is listed as the grantee of this parametric role.

If you select **Application Role**:

- a. From the Application Role list, select OracleBPMProcessRolesApp.
- **b.** Specify a parameter by entering a dollar sign (\$), then selecting from the list that appears next to the field.
- **c.** Configure the conditions for the inclusion of the grantees. To do this, from the **Add Condition** list, select a condition. The listed conditions include the standard LDAP attribute conditions of LDAP repository. There are 2 condition types listed:
 - LDAP attribute condition: The LDAP attribute condition appears in title case in the condition list.
 - Extended User Properties: The extended user properties condition appears in upper case in the condition list.
- **d.** Click **Add Criteria Row** and specify the values for each condition. The values can be any of the following: (1) The application role or group to which the users must belong, (2) The extended user properties the users must match. You specified the extended user properties in Section 10.7, "Managing Extended User Properties in Process Workspace", and (3) Any parameters you may have defined for a parametric role used in assigning a task. When you begin entering a value, precede it with a dollar sign (\$). The list of defined parameters appears next to the field as in the example in Figure 10–7.

Administration Areas	Parametric Roles 🛛 🗳 🗙	Details : New Parametric Role	Save Revert
Organization	New Parametric Role	* Name New Parametric Role	
Roles			
Calendars		V Parameters	
Organization Units		Name Type	
Parametric Roles		1 param1 String	
Extended User Properties		2 param2 String	
Flex Fields		3 param3 String	
Public Flex Fields		4 param4 String	
Protected Flex Fields		Include people that must these conditions :	
Task Administration			
Approval Groups		Add condition Description 💌 💠	
Evidence Search			"Please use '\$' symbol to use parameters"
Task Configurations		Grantees Application Role V OracleBPMProcessRolesApp V \$param2	~ ~ ×
Application Preferences		Name equais Sisten	\$
Application Preferences		\$paran1 \$paran2 \$paran3 \$paran4	^

Figure 10–7 Selecting Defined Parameters as Conditions of Users with Parametric Roles

- **e.** After you have selected one or more conditions, the conditions you specified appear in the list of users you want to include. You can then specify the filter for that condition.
- 7. Click Save.
- **8.** In addition to the standard LDAP attribute condition which is listed by default in the condition list, adding a condition based on custom LDAP attributes is supported. For custom LDAP attributes condition additional configuration is to be done from Oracle Enterprise Manager Fusion Middleware Control. To add the custom LDAP attribute condition:
 - **a.** Log in to the Oracle Enterprise Manager Fusion Middleware Control (http://<server name>:<port number>/em/)
 - b. Expand Farm_soainfra, SOA, soa-infra (AdminServer)
 - **c.** Navigate to **SOA Infrastructure**, **Administration**, **System Mbean Browser** as shown in Figure 10–8.

ORACLE Enterprise Manager 11g Fusion Mic	ddleware Control			
📲 Farm 👻 🚑 Topology				
 ▼ Farm_soainfra ▷ Application Deployments ▽ SOA ▽ SOA ▽ SoA ▷ default ▷ WebLogic Domain ▷ Metadata Repositories 	Soa-infra () SoA Infrastructure Home Monitoring Logs SOA Deployment	•	Applica > Show Attribu	tion Defined MBeans: B v MBean Information rtes Operations Not
User Messaging Service	Manage Partitions Service Engines Bindings Services and References Business Events SOA Administration		Nam Aud Aud Aud Aud Aud Aud Aud	e iitDetaiThreshold iitKeyExtents
	Administration General Information	MDS Config TopLink Ses	uration sions	pdateBatchSize
3		System MBe	an Browser 7 Cub 8 Cub 9 Cub	esspan MBean beInstanceExpiration beTimerMaxErrorCount beTimerMaxSkipOnErrorCount

Figure 10–8 Customizing LDAP Attributes in Enterprise Manager

 Expand Application Defined MBeans, oracle.as.soainfra.config, Server: Admin Server, WorkflowIdentityConfig, human-workflow, WorkflowIdentityConfig.ConfigurationType, jazn.com, WorkflowIdentityConfig.ConfigurationType.ProviderType as shown in Figure 10–9.

Figure 10–9 Viewing Attributes Tab for JpsProvider

6 7 3	_	Applic	ation w MB	Defined M ean Informat	1Beans: tion	Work
B G CEPConfig B G CEPConfig B G ENCOnfig B G HWEMalexConfig	*	Attribu	utes me	Operations	Notificati	ons Des
B ☐ MediatorConfig B ☐ SoaInfraConfig B ☐ SoaInfraConfig		2 0	Config! Connec	MBean tion		If t Nar
WorkflowConfig workflow workflow WorkflowConfig.DynamicAssignmentFunction WorkflowConfig.DynamicAssignmentFunction		4	eventP eventT	rovider ypes		If t JSR All
WorkflowConfig-LocaleList WorkflowConfig.TaskAutoReleaseConfiguration WorkflowIdentityConfig		6	Name objecti	lame		The
buman-workflow WorkflowIdentityConfig.ConfigurationType		9	Provide ReadO	erType nly		The
 jazn.com WorkflowIdentityConfig.ConfigurationType.Providentity SignsProvider 	eT	11 12	Restar RoleCo	tNeeded ntrols		Na Na
Comparing Control Contro Control Control Control Control Control Control Control Control		13	Search	ableUserAttribu	uteMap	ke
E 📄 orade.as.util E 🔄 orade.dfw		14 15 16	Service System UserCo	MBean Introls		The If t
æ orade.dms æ orade.dms vevent.config ₩ orade.dms.event.config		16	UserCo	ntrois		Na

e. Click JpsProvider and the Attributes tab appears on the right pane

SearchableUserAttribtueMap contains the custom attribute mapping a user can add.

f. Navigate to the Operations tab as shown in Figure 10–10.

Figure 10–10 Viewing Operations Tab for JpsProvider

Attrib	utes Operations	Notifications									
1	Name		Description	Parameters	Return Type						
1	addSearchableUserAt	tributeMap	ProviderMXBean.SearchableUserAttributMapAd	2	boolean						
2	createConnection		Create a new connection.	0	javax.management.ObjectName						
3	createProperty		Create a new property.	0	javax.management.ObjectName						
4	createRoleControls		Create a new Role Control.	0	javax.management.ObjectName						
5	createUserControls		Create a new User Control.	0	javax.management.ObjectName						
6	removeConnection		removeConnection		removeConnection		removeConnection		Remove a connection.	0	boolean
7	removeProperty		Remove a new property.	1	boolean						
8	removeRoleControls		Remove Role Control.	0	boolean						
9	9 removeSearchableUserAttributeMap		Remove a Attribute from Map.In this map the key is the attributeName used by Identity service and value is the attribute name supported by underlying provider	1	boolean						
10	removeUserControls		Remove User Control.	0	boolean						
11	setClassName		The class name of the provider.	1	boolean						
12	2 setProviderType		viderType The provider type for this provider configuration.		boolean						
13	setService		The associated service.	1	boolean						

Application Defined MBeans: WorkflowIdentityConfig.....ProviderType:JpsProvider

- **g.** Click the *addSearchableUserAttributeMap* operation to populate the needed values as shown in Figure 10–11:
 - The first parameter is the user attribute
 - The second parameter is the provider specific attribute mapped to in LDAP

Figure 10–11 Parameters for addSearchableUserAttributeMap

Operation: addS	earchableUserAttribute	еМар	Invoke	Revert	Return						
MBean Name orade.as.soainfra.config:WorkflowIdentityConfig.ConfigurationType=jazn.com,Location=soa_server1, WorkflowIdentityConfig=human-workflow,name=JpsProvider,type=WorkflowIdentityConfig.ConfigurationType.ProviderType, Application=soa-infra											
Operation Name	addSearchableUserAttributeMa	idSearchableUserAttributeMap									
Description	ProviderMXBean.SearchableUse	viderMXBean.SearchableUserAttributMapAdd									
Return Type	boolean	lean									
Parameters											
Name	Туре	Value									
p1	java.lang.String										
p2	java.lang.String										

Return Value

h. Click Invoke to save the changes.

The values are displayed in the Attributes tab.

10.6.4 How to Modify a Parametric Role

To modify a parametric role:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- 2. Select Parametric Role. The Parametric Role panel appears in the right pane.

- **3.** From the Parametric Role panel, select the role you want to modify. The Details panel for that role appears to the right.
- **4.** Enter the new values for the role just as you did when you created the role. See "To create a parametric role:" on page 10-10.
- 5. Click Apply.

10.6.5 How to Delete a Parametric Role

To delete a parametric role:

- **1.** From the Process Workspace toolbar, click **Administration**. The Administration Areas panel appears in the left pane.
- 2. Select Parametric Role. The Parametric Role panel appears in the right pane.
- **3.** From the Parametric Role panel, select the role you want to delete. The Details panel for that role appears to the right.
- 4. Click Delete Parametric Role and, when prompted, confirm the deletion.

10.7 Managing Extended User Properties in Process Workspace

Extended user properties are used in defining parametric roles, which are described in Section 10.6, "Managing Parametric Roles in Process Workspace.".

Typically, users have some properties specified in Oracle Internet Directory or some other LDAP directory. Often, however, additional properties specific to their organization and roles are necessary. At times, these properties are added on demand when newer business processes are created. At that time, it might not be possible to extend the company's global LDAP directory. In these cases, extended user properties are useful. For example, you can specify that a given user with the sales representative role is located in California. Although, from a functional point of view, there is only one role, the individual user is associated with a property, and that property has a value assigned for that user.

Figure 10–12 shows an example of the Extended User Properties page.



Figure 10–12 Sample Extended User Properties Page

Extended user properties can be assigned to users, groups, or roles.

When you create extended user properties, you define the following:

- Property name
- Property type, which can be of the following data types: string, number, date and free form text. String data types are enumerated values. Free form strings can have any value.
- Enumeration in case of a string data type. Note that only string-typed properties can be assigned values at the time of definition and only those values can be assigned to users when the property is associated with a user.

After properties are defined, they can be associated with any user. During association, values must be assigned to the property for that user

Only users who have Administrator privileges can define new properties.

To add extended user properties:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Extended User Properties**. The Extended User Properties page appears in the right pane.
- **3.** In the **Properties** pane, click **Add Property**. Property fields appear in the table.

An Extended User Property has the following fields:

- Name—The name can be any user defined string that is meaningful to the user or the company. Case does not matter. The system will convert all names to uppercase.
- Type—The type field is a drop-down list from which the user must select the data type of the values that the property is expected to hold when it is associated with various users in the organization.
- Value—The value field is meaningful only when the property is assigned the data type string because only string typed properties can be assigned enumerations of values. This value can be a comma-delimited list.

For all other data types (number, date, and freeform text) no values can be assigned to the property.

For number, date, and freeform text-typed properties, a value is assigned to the property only when the property is associated to a user and the value assigned can be different for each user.

4. After you specify the property, click **Apply**. A column for this property appears in the Map Properties pane.

To modify extended user properties:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Extended User Properties**. The Extended User Properties page appears in the right pane.
- **3.** In the **Properties** pane, select the property you want to edit, then click **Edit Property**. The row for that property becomes editable.
- 4. Enter your changes, then click Apply.

To delete extended user properties:

Note: Deleting an extended user property already in use by an parametric role causes an error when administering that role.

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** In the Organization panel, click **Extended User Properties**. The Extended User Properties page appears in the right pane.
- **3.** In the Properties pane, select the property you want to delete, then click **Delete Property**. A Confirm Delete dialog box advises you that deleting a property will dissociate this property from and associated user, if any. If you are sure you want to proceed, click **Yes**.
- **4.** In the Properties pane, click **Apply**.

To associate extended user properties to users:

- 1. To specify the user with whom this property is to be associated, in the Map Properties pane, click **Add User**. The User field appears.
- **2.** Specify the user. You can do this either by entering the user identifier in the field or by clicking **Select User**, and, in the Identity Browser window that appears, performing a search for the user and clicking **OK**.
- **3.** If the property is a string type, then, in the Map Properties pane, from the column for the property, select the property to assign to this user. If the property is a number, enter the appropriate value in the text field.

Note: The **All** check box means that you wish to assign all the values defined on the string property to a particular user. It does not mean that you are assigning all the properties defined in the system to the user. The purpose of the **All** check box must be understood in the context that a user can be assigned multiple values for a string and a freeform text-typed property. For string-typed properties the values that you can assign to a user are enumerated at the time of property definition itself and the **All** check box enables you to assign all of them to a user at once. Freeform text-typed properties do not have any values defined on them. To assign multiple values to a freeform text-typed property at the time of associating it to a user, enter the multiple values and separate each value by a comma.

4. After you map the property to the user, in the Map Properties pane, click Apply.

10.8 Managing Organization Units in Process Workspace

An organization unit represents departments or divisions within an organization. Organization units can contain child organization units creating a hierarchy that corresponds to your organization.

When you create an organization unit, you define the following:

- Organization unit name
- Time zone
- Members of the organization unit, which can include users, groups, application roles, or parametric roles
- Optional parent organization unit
- Optional business calendar

When a process is associated with an organization unit, only members of that organization unit and its children can see that process and the tasks initiated by it.

Note: A manager is not a member of an organization unit by default. The manager must be added to the organization unit in order to see the processes and the tasks within it.

If a process is associated with an organization unit, then the tasks initiated by the process can use the business calendar associated with the organization unit. For more information , see Section 10.3, "Managing Calendar Rules in Process Workspace".

To create an organization unit:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** Under Organization, click **Organization Units**. The Organization Units panel appears.
- **3.** From the **Add Organization Unit** list, select either **Root Organization Unit** or **Child Organization Unit**. The Create New Org Unit dialog box prompts you for the name of the organization unit.

4. Enter the name of the organization unit, and click **OK**. The new organization unit appears in the Organization Units list as shown in Figure 10–13.

Figure 10–13 Adding an Organization Unit

ORACLE' Business	Process Workspace					🔁 Home	Administration	Preferences Help	Logout	weblogic S
Administration Areas	Search	•	⊻ Details	: Sales					Save	Revert
Organization	Organization Units	🔮 👻 🗙	* Name	Sales						
Roles	▷ Sales		Description							
Calendars										
Organization Units										
Parametric Roles										
Extended User Properties			Calendar	S. 62						
Flex Fields				Add Users/Groups/App Role + 😽	Add Parametric Role	+ ×				
Public Flex Fields				jcooper						
Protected Flex Fields										
Task Administration			Members							
Approval Groups										
Evidence Search										
Task Configurations										
Application Preferences				* X						
Application Preferences										
			Manager							
			manayers							
	1									
Security @ 2005, 2012, Overlage de	its affiliatos. All sights recorded		_						_	

- **5.** In the right pane, in the Details section for the new organization unit, do the following:
 - **a.** Optionally provide a description of the newly created organization unit.
 - **b.** From the Calendar list, select a calendar rule to associate with this organization unit.

For information about adding a calendar rule, see Section 10.3, "Managing Calendar Rules in Process Workspace".

- **c.** In the Members window, click **Add new member**. The Select Member dialog box prompts you to search for a user or group to add to this organization unit.
- **d.** Enter the name of the user or group you want to add, then click **Search**. From the Available column, select one or more members you want to add, and use the arrow button to move them to the Selected column. Click **OK**. The members you added appear in the Members window.
- e. In the Add Parametric Role, select Add new member. The Select Member dialog prompts you to select a parametric role. Click OK.

Only the parametric roles that do not have any defined parameters appear in the Parametric Role window.

- f. Similarly, in the Managers window, select Add new member. The Select Manager dialog box prompts you to search for a user or group to add to this organization unit.
- **g.** Enter the name of the manager you want to add, then click **Search**. From the Available column, select one or more managers you want to add, and use the arrow button to move them to the Selected column. Click **OK**. The managers you added appear in the Members window.
- 6. Click Apply.

To edit an organization unit:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** Under Organization, click **Organization Units**. The Organization Units panel appears.
- **3.** In the Organization Units panel, select the organization unit you want to edit.
- **4.** In the right pane, in the Details section for the new organization unit, make the changes you want, then click **Apply**.

To delete an organization unit:

- **1.** From the Process Workspace toolbar, select **Administration**. The Administration Areas panel appears.
- **2.** Under Organization, click **Organization Units**. The Organization Units panel appears.
- **3.** In the Organization Units panel, select the organization unit you want to delete, and click **Delete Organization Unit**.

Modifying Running Process Instances in Process Workspace

This chapter describes how process owners and Process Workspace administrators can alter the flow of running process instances in Process Workspace. It also describes how they can migrate instances in Process Workspace.

This chapter contains these topics:

- Section 11.1, "Altering the Flow of Running Process Instances in Process Workspace"
- Section 11.2, "Migrating Process Instances in Process Workspace"
- Section 11.3, "What You May Need to Know About Redeploying a Process"
- Section 11.4, "Moving Running Composite Instances from One Revision to Another"

11.1 Altering the Flow of Running Process Instances in Process Workspace

You can move the token of a running process instance from a current activity to another, but only within the following structural constraints:

- You can move an instance only to an activity defined at the same process level.
- You cannot move an instance to a boundary event.
- You cannot move an instance into a boundary event flow unless the boundary event is interrupting and is defined at the same level.
- You cannot move an instance inside or outside an event subprocess.
- You cannot move an instance outside a gateway branch. You can, however, move it to the merging gateway to merge that path.

Note: When using an exclusive gateway, you can move the instance outside the exclusive gateway branch.

11.1.1 How to Alter the Flow of Running Process Instances in Process Workspace

To alter the flow of a running process instance:

1. From the Process Tracking tab page, select a process instance. The details of that instance appear.

2. From the Actions list, select Alter Flow and Suspend.

If you select a single process instance, then the Alter Flow dialog box appears as shown in the example in Figure 11–1.

Jpen Activities						Comments	
Current Activity	New Act	tivity Lo	cation				
UserTask	Select Acti	ivity 💉 🛛 Co	prrelationProcess				.:
lation key Data Objects							
ation key Data Objects ✓ Show correl Name	ation keys 🛛 🔽	Show instanc	e attributes	Details	t correlatior	ı key	
Data Objects	ation keys 🔽 nKey1	Show instance Scope Correlat	e attributes	Details	t correlatior	key	
Data Objects	ation keys 🔽 Key1 Jerty	Show instance Scope Correlat Correlat	e attributes ionProcess ionProcess	Details Rese intProperty stringProp	t correlation	ı key	
Data Objects	ation keys 🔽 nKey1 nerty roperty	Show instance Scope Correlat Correlat Correlat	e attributes ionProcess ionProcess ionProcess	Details Rese intProperty stringProp	t correlation / 1 erty test	i key	
Data Objects ✓ Show correl Name ✓ correlation intProp stringP ↔ intDO ↔ stringDO	ation keys 🔽 Key1 berty roperty	Show instance Scope Correlat Correlat Correlat Correlat	e attributes ionProcess ionProcess ionProcess ionProcess ionProcess	Details Rese intProperty stringProp	t correlation / 1 erty test	key	
Lation key Data Objects ✓ Show correl Name ✓ correlation intProp stringP ↔ intDO ↔ stringDO ₩20 composite	ation keys 🔽 Key1 berty roperty Name	Show instance Scope Correlat Correlat Correlat Correlat Correlat	e attributes ionProcess ionProcess ionProcess ionProcess ionProcess	Details Rese intProperty stringProp	t correlation / 1 erty test	key	-
Lation key Data Objects ✓ Show correl Name ✓ correlation intProp stringP (*) intDO (*) stringDO ************************************	ation keys IKey1 rerty roperty Name Label	Show instance Scope Correlat Correlat Correlat Correlat Correlat Correlat	e attributes ionProcess ionProcess ionProcess ionProcess ionProcess ionProcess	Details Rese intProperty stringProp	t correlation / 1 erty test	key	
Image: Second secon	ation keys key1 key1 roperty Name Label s	Show instance Scope Correlat Correlat Correlat Correlat Correlat Correlat Correlat	e attributes ionProcess ionProcess ionProcess ionProcess ionProcess ionProcess ionProcess	Details Rese intPropert stringProp	t correlation / 1 erty test	ı key	 -

Figure 11–1 Altering the Flow of a Process Instance

By default, the Alter Flow dialog box shows only the Process and Project data attributes. When you select one of the open activities, it shows the data objects within the context of that open activity. For example, if the open activity is a subprocess, then the Alter Flow dialog box displays any subprocess level data objects.

- 3. In the Alter Flow dialog box, you can do any or all of the following:
 - View the list of current activities in a running process instance in the Open Activities table. You can select the new activity where you want to move the token. Note that the New Activity list includes only those locations to which a token can be moved validly.
 - Include the correlation keys in the list of data objects as shown in Figure 11–1. The correlation keys will be visible only if the process is using correlation. To do this, select Show correlation key. If you select a correlation key, then the details section shows the correlation property that comprises the key. To reset the correlation key, select Reset correlation key.

For information about correlation keys, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*.

- Include process instance attributes in the list of data objects as shown in Figure 11–1. To do this, select Show instance attributes. To display the details of a particular instance attribute, select the attribute. Note that only those instance attributes that can be modified appear as editable. Others are read-only.
- Add any comments in the Comments field.
- **4.** Click one of the following:
 - Resume to move the token to another activity and return the process to a running state.
 - Save to suspend the process instance. You can return to it later to alter its flow and resume it.
 - Cancel to return the process instance to its last saved state. The process continues in the running state.

Note:

- If you have opened the Alter Flow dialog box and the session times out before you have taken any action, then the process instance is suspended.
- If an activity has more than one running tokens associated with it, you cannot alter the flow to select and move just one of the tokens to another valid target activity. If you alter the flow for that open activity, then all the tokens move.

11.2 Migrating Process Instances in Process Workspace

If a new implementation of the same composite revision is deployed, and the previously deployed composite has some components with the newer BPM process definition, then the affected components go into suspended state. To resume a suspended component, either migrate or suspend process instances that are in the pending migration state.

Unless you resume the component—either explicitly or automatically when all pending migration instance are resumed—the new task instances are not created as a result of the token having been moved.

Oracle BPM provides a way to retain the existing running instances for a process when that process is redeployed. To do this, while redeploying the process in Fusion Middleware Control Console, select **Keep running instances**. If the redeployment succeeds, then the instance goes into pending migration state and the process component is suspended.

Note: If the composite being redeployed has more than 100 existing instances, only 100 are migrated during the redeployment and the component is suspended. This limit is used to minimize the deployment time.

To migrate the rest of the instances, use Process Management APIs.

This section details how to migrate a pending migration instance and how to resume a suspended component. It contains these topics:

Note: If you are running task instances created prior to 11g Release 1 (11.1.1.5.0 Feature Pack), and then create a new instance by using a later release, do not migrate to the new instance until all previously running task instances have completed.

- Section 11.2.1, "How to Migrate a Process Instance in Process Workspace"
- Section 11.2.2, "How to Resume a Suspended Component"

11.2.1 How to Migrate a Process Instance in Process Workspace

You can migrate a process instance in either of two ways:

- Migrate as is: Migrate the instance without moving the token or updating the payload
- Make changes and migrate: Move the token to the new activity you specify, and update the payload as you specify, then migrate the instance

To migrate process instances:

1. Select one or more process instances. From the Actions list, select **Migrate**. The Migrate dialog box appears.

The elements in this dialog box depend on your selection. If you select **Migrate as is**, then only the Comments field is available. If you select a single instance and then select **Make changes and migrate**, then the Migrate dialog box appears as shown in Figure 11–2.

Figure 11–2 Migrating a Process Instance

Current Activity New Activity Location interBankAccountDel Select Activity ▼ OnboardingProcess fanualNewHireTrainir Select Activity ▼ OnboardingProcess Data Objects Show correlation keys Show instance attributes Details Name Scope	Current Activity New Activity Location nterBankAccountDel Select Activity V OnboardingProcess IanualNewHireTrainir Select Activity V OnboardingProcess Data Objects	Current Activity New Activity Location nterBankAccountDe Select Activity V OnboardingProcess IanualNewHireTrainin Select Activity V OnboardingProcess Data Objects
Interfactor influctual interfactor influctual interfactors IanualNewHire Trainin	TanualNewHireTrainir Select Activity V OnboardingProcess	Data Objects
Data Objects Details Show correlation keys Show instance attributes Name Scope Value Scope Value OnboardingProcess	Data Objects	
Data Objects Details Show correlation keys Show instance attributes Details Name Scope Vame Scope training OnboardingProcess cx) equipment OnboardingProcess cx) equipment OnboardingProcess cx) enployee OnboardingProcess	Data Objects	Data Objects
Data Objects Details Show correlation keys Show instance attributes Details Name Scope training OnboardingProcess training OnboardingProcess training OnboardingProcess training>	Data Objects	Data Objects
Data Objects Details Show correlation keys Show instance attributes Name Scope XX training OnboardingProcess <training xmins="http://www.example.org/hcm"> XX equipment OnboardingProcess <training> XX equipment OnboardingProcess <training></training></training></training>	Data Objects	Data Objects
Data Objects Details Show correlation keys Show instance attributes Details Name Scope training OnboardingProcess training OnboardingProcess ctraining > OnboardingProcess ctraining >	Data Objects	Data Objects
Data Objects Details Show correlation keys Show instance attributes Details Name Scope training OnboardingProcess training OnboardingProcess ctraining xmlns="http://www.example.org/hcm"> training OnboardingProcess ctraining xmlns="http://www.example.org/hcm"> training OnboardingProcess ctraining xmlns="http://www.example.org/hcm"> ctraining xmlns="http://www.example.org/hcm"> ctraining xmlns="http://www.example.org/hcm"> ctraining xmlns="http://www.example.org/hcm"> ctraining ctraining xmlns="http://www.example.org/hcm"> ctraining xmlns="http://www.example.org/hcm">	Data Objects	Data Objects
Data Objects Show correlation keys Show instance attributes Name Scope Name Scope training OnboardingProcess training OnboardingProcess ct) equipment OnboardingProcess ct) equipment OnboardingProcess	Data Objects	Data Objects
Name Scope (x) training xmlns="http://www.example.org/hcm"> (x) equipment OnboardingProcess (x) equipment OnboardingProcess (x) enployee OnboardingProcess	Show correlation keys Show instance attributes Details	Show correlation keys Show instance attributes Details
(x) training OnboardingProcess <\Name><\Name> (x) equipment OnboardingProcess <\Type> (x) employee OnboardingProcess		
(x) equipment OnboardingProcess (x) employee OnboardingProcess	Name Scope Name	Name Scope
employee OnboardingProcess	Name Scope Kit Scope Kit Chaining OnboardingProcess Chaining OnboardingProcess Chaine>	Name Scope xtraining ChoordingProcess xtraining OnboardingProcess
	Name Scope Xi training OnboardingProcess CN equipment OnboardingProcess Classical Classical Classical Classical Classical Classical Control Classical Control Classical Clastre	Name Scope Xi training OnboardingProcess CX equipment
bankAccount OnboardingProcess	Name Scope Xi rraining OnboardingProcess Quipment Quipment Quipment Quipment Quipment Quipment Quipment Quipment Quipment <th<<tr> <th<<tr> <th<<tr></th<<tr></th<<tr></th<<tr>	Name Scope XX training OnboardingProcess XX equipment OnboardingProcess XX employee OnboardingProcess
	Name Scope Value Scope Value Scope Value OnboardingProcess OnboardingProcess OnboardingProcess DankAccount OnboardingProcess	Name Scope training OnboardingProcess (x) equipment OnboardingProcess (x) equipment OnboardingProcess (x) enployee OnboardingProcess (x) bankAccount OnboardingProcess
	Name Scope Vitraining OnboardingProcess vitraining OnboardingProcess vitraining OnboardingProcess vitraining OnboardingProcess	Name Scope Vitraining OnboardingProcess Vitraining OnboardingProcess Vitraining OnboardingProcess Vitraining OnboardingProcess Vitraining OnboardingProcess Vitraining OnboardingProcess Vitraining Vitraining Vitraining Vitraining Vitraining Vitraining <th< th=""> <th< th=""></th<></th<>
bankAccount OnboardingProcess	Name Scope Varianing OnboardingProcess Ct) eruipment OnboardingProcess Ct) employee OnboardingProcess	Name Scope training OnboardingProcess <training xmlns="http://www.example.org/hcm"> training OnboardingProcess <name> equipment OnboardingProcess <type> meployee OnboardingProcess</type></name></training>
bankAccount OnboardingProcess	Name Scope Viame Scope Viaming OnboardingProcess Character OnboardingProcess Character OnboardingProcess Character OnboardingProcess Character OnboardingProcess Character OnboardingProcess	Name Scope (X) training OnboardingProcess (X) equipment OnboardingProcess (X) equipment OnboardingProcess (X) employee OnboardingProcess
Dankaccount OnboardingProcess	Name Scope (x) training OnboardingProcess (x) equipment OnboardingProcess (x) employee OnboardingProcess (x) employee OnboardingProcess	Name Scope (x) training OnboardingProcess (x) equipment OnboardingProcess (x) employee OnboardingProcess (x) employee OnboardingProcess
	Name Scope Value Scope training OnboardingProcess cx) equipment OnboardingProcess cx) enployee OnboardingProcess cx) bankAccount OnboardingProcess	Name Scope training OnboardingProcess cx) equipment OnboardingProcess cx) employee OnboardingProcess cx) bankAccount OnboardingProcess
	Name Scope Varianing OnboardingProcess CM equipment OnboardingProcess CM enployee OnboardingProcess CM enployee OnboardingProcess CM enployee OnboardingProcess	Name Scope training OnboardingProcess cx equipment OnboardingProcess cx equipment OnboardingProcess cx bankAccount OnboardingProcess
	Name Scope training OnboardingProcess	Name Scope (x) training OnboardingProcess <name> <name> <name> <name></name></name></name></name>
	Name Scope	Name Scope (X) training OnboardingProcess (X) equipment OnboardingProcess (X) equipment OnboardingProcess (X) employee OnboardingProcess (X) bankAccount (X) bankAccount bankAccount
bankAccount OnboardingProcess	Name Scope training OnboardingProcess equipment OnboardingProcess employee OnboardingProcess child Intervention	Name Scope training OnboardingProcess equipment OnboardingProcess employee OnboardingProcess chaining
	Name Scope Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"> Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"> Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"> Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"> Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"> Itraining OnboardingProcess <itraining xmins="http://www.example.org/hcm"></itraining></itraining></itraining></itraining></itraining></itraining>	Name Scope training xmlns="http://www.example.org/hcm"> training OnboardingProcess equipment OnboardingProcess OnboardingProcess
employee OnboardingProcess	Name Scope <training xmlns="http://www.example.org/hcm"></training>	Name Scope training xmins="http://www.example.org/hcm"> training OphoardingProcess <name></name>
(x) equipment OnboardingProcess (x) employee OnboardingProcess	Name Scope http://www.example.org/hcm">	Name Scope <a> <a> <a> <a> <a> <a> <a> <a> <a> <a> <br <="" th=""/></br></br></br></br></br></br>
cx) equipment OnboardingProcess <type> </type> cx) equipment OnboardingProcess cx) employee OnboardingProcess	Name Scope <training xmlns="http://www.example.org/hcm"></training>	Name Scope VariationScopetraining xmlns="http://www.example.org/hcm">
training OnboardingProcess <type></type>	Manage Canage Canage Strategy and Strategy Strat	Blance Connection
(x) training OnboardingProcess <\Name><\Name> (x) equipment OnboardingProcess <\Type> (x) employee OnboardingProcess		
(x) training OnboardingProcess <name> (x) equipment OnboardingProcess <type></type><th></th><th></th></name>		
(x) training OnboardingProcess (x) equipment OnboardingProcess (x) employee OnboardingProcess		
(x) training OnboardingProcess (x) equipment OnboardingProcess (x) employee OnboardingProcess		
(x) training OnboardingProcess (x) equipment OnboardingProcess (x) employee OnboardingProcess		
(x) training OnboardingProcess <name></name> (x) equipment OnboardingProcess <type></type> (x) employee OnboardingProcess		
(x) training OnboardingProcess <name></name> (x) equipment OnboardingProcess <type></type> (x) employee OnboardingProcess		
(x) training OnboardingProcess <name> (x) equipment OnboardingProcess <type> </type> (x) enployee OnboardingProcess</name>		
premie jscope control (x) training OnboardingProcess (x) equipment OnboardingProcess (x) equipment OnboardingProcess (x) enployee OnboardingProcess		onew contraction keys bitow inscaling autoputes becomes

If you select Make changes and migrate, you can do any or all of the following:

 In the Open Activities table, you can view the activities where the token is currently running, and you can select the new activity where you want to move the token. Note that the New Activity list includes only those locations to which a token can be moved validly.

Note:

- For a pending migration instance to resume, all removed activity tokens need to be moved to new activities.
- If you select more than one activity from the Open Activities table, then the Data Objects tab does not appear, and you are not able to modify the payloads.
- Include the correlation keys in the list of data objects. To do this, select Show correlation keys, then, in the Data Objects tab, select Show as tree and expand the data object. If you select a correlation key, then the details section shows the correlation property that comprises the key. To reset the correlation key, select Reset.

For information about correlation keys, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*.

- Include process instance attributes in the list of data objects. To do this, select Show instance attributes, then, in the Data Objects tab, select Show as tree, then expand the data object, then expand Instance Attributes. To display the details of a particular instance attribute, select the attribute. Note that only those instance attributes that can be modified appear as editable. Others are read-only.
- Add any comments in the Comments field.
- **2.** Click one of the following:
 - **Save** to keep the process in the pending migration state.
 - Migrate to migrate the process and put it in the running state.
 - **Cancel** to return the process instance to its last saved state.

If for some reason the migration fails, then the Migration Confirmation dialog box tells you why, as shown in Figure 11–3.

Figure 11–3 Migration Confirmation: Failure Message

Mig	ration Confirmatio	n		
`	/our request failed			
	Instance	Exception	Solution	
[200340	Flow element EnterBankAccountDetails not found in process		
				Close

11.2.2 How to Resume a Suspended Component

All suspended components are listed in the Pending Components panel, as shown in the example in Figure 11–4.

Applications	Actions - Views - 🔞 Search		🥔 🕑 Advance	d 😭			
🕑 Show 🗸 😽	Title	Number	Initiator	Process Name	Initiated	Status	
Application	Instance #1 of ReviewProcess	1		[Project3 v1.0] ReviewProcess	Dec 5, 2011 10:40 AM	<u>(-)</u>	
o applications available.	Instance #2 of ReviewProcess	2		[Project3 v1.0] ReviewProcess	Dec 5, 2011 10:40 AM	•	
	Instance #3 of ReviewProcess	3		[Project3 v1.0] ReviewProcess	Dec 5, 2011 10:40 AM	<u>(</u>	
	Instance #4 of ReviewProcess	4		[Project3 v1.0] ReviewProcess	Dec 5, 2011 10:40 AM	()	
⊻ Views (List filters)	Instance #1 of ReviewProcess	5					Actions
E Inbox	× Details						
7 🆐 History	Dectails						
🗄 Last day	Priority normal	Process [Project3 v1.0]	Cre	eation time 12/5/2011			
🗉 Last week	Status Pending Migration	Number 1					
🗉 Last month		Number 1					
Three months ago	✓Open Activities						10
	Activity Title	Α	ssianed to	Priority	State	Actions	
	ReviewProcess#activ			-	Pending Migration		
	SimpleUserReview			-	Pending Migration		
Pending Components							
ReviewProcess							
	Neudit Tanil						
	≤Comment		✓ Attachments				
			Filename	Description			
	1						

Figure 11–4 Pending Components Panel in the Process Tracking Tab Page

To resume a suspended component:

 From the Pending Components panel, select the component you want to resume. The Resume Component dialog box appears as shown in the example in Figure 11–5. It tells you how many instances are pending migration for that component.

Figure 11–5 Resume Component Dialog Box



- **2.** In the Resume Component dialog box, choose whether you want the pending migration process instances to be:
 - Migrated as is
 - Suspended

If there are instances that cannot be migrated-as-is, and you want to resume the component, then select **Leave non-migrated instances as suspended**. If you do not select this option, and there are instances that cannot be migrated as-is, then the component can not be resumed.

3. Click **Resume**. The component is resumed and it no longer appears in the Pending Components panel. A Resume Component Confirmation dialog box tells you the result and how many instances are suspended. A new process instance can now be initiated and its requests will be processed.

Note: You can initiate a process instance even if a component is suspended. In this case, however, the requests are queued and are not processed until the component is resumed.

11.3 What You May Need to Know About Redeploying a Process

If you are redeploying a process, and, during design time, the **Keep running instance** option is selected, then redeployment fails in the following scenarios:

- Parallel or inclusive gateway pair is removed
- Inclusive-complex gateway pair is changed to inclusive-inclusive gateway pair
- Subprocess is removed or its loop characteristic is changed
- User task is moved to another branch in gateway pair or outside the gateway pair
- Activity level is changed., for example, the activity is moved inside a subprocess or gateway structure
- Subprocess or event subprocess or a call activity is added
- Event subprocess is changed from interrupting to non-interrupting
- Boundary event is changed from interrupting to no- interrupting
- Boundary event is added
- User task implementation is changed to use another human task

If you do not want redeployment to fail in these scenarios, then you must set the *force.deploy* flag manually by editing the composite.xml file in Oracle JDeveloper. To set *force.deploy* at the component level, enter the following:

```
<component name="T4">
  cproperty name="oracle.bpm.bpmn.force.deploy">true</property>
<implementation.bpmn src="processes/T4.bpmn"/>
</component>
```

To set the *force.deploy* at the composite level, you can use the same .xml element as a child of the composite element as shown in the example in Figure 11–6.

<pre>location="orands:/soa/shared/workflow/TaskServiceInterface.wsdl" importType="wsdl"/> </pre> <pre>callbacksoration="orands:/soa/shared/workflow/TaskServiceInterface.wsdl" importType="wsdl"/> </pre> <pre>callbackInterface="http://xmlns.oracle.com/BPMBpel/BPMBpel/BPMBpel/ dinding.ws port="http://xmlns.oracle.com/BPMBpel/BPELProcess2#wsd" callback></pre>	
<pre>import ype="wsd"> import ype="wsd"/> import ymport ymport</pre>	1
<pre> </pre>	
<pre> <interface.wsdl bpelprocess2#wsd"<="" bpmbpel="" http:="" interface="http://xmlns.oracle.com/BPMBpel/BPMBpel/BPELProc</td><td></td></tr><tr><td><pre>dignding.ws port=" td="" xmlns.oracle.com=""><td>ess2#wsdl.inter BPELProcess2#ws</td></interface.wsdl></pre>	ess2#wsdl.inter BPELProcess2#ws
<pre></pre>	.endpoint(bpelp
<pre><td>dl.endpoint(bpe</td></pre>	dl.endpoint(bpe
<pre> </pre> <	
<pre>callbackInterface="http://xmlns.oracle.com/opmnProcess/Process callbackInterface="http://xmlns.oracle.com/opmnProcess/ProcessBPMBpel# <callbacks< td=""><td>DDNDsol Hugell in</td></callbacks<></pre>	DDNDsol Hugell in
<pre></pre>	/ProcessBPMBpe1
<pre> <callback> </callback></pre>	sd1.endpoint(Pr
<pre></pre>	
<pre> <td>#wsal.enapoint(</td></pre>	#wsal.enapoint(
<pre></pre> <pre><</pre>	
<pre>component name="Humantask1"></pre>	avel
<pre><implementation.bpmn src="processes/ProcessBPMBpel.bpmn"></implementation.bpmn></pre>	
<pre><pre><pre><pre><pre><pre><pre><pre></pre></pre></pre></pre></pre></pre></pre></pre>	
<pre> <</pre>	tlevel
<pre><component name="BPELProcess2" version="1.1"> <implementation.bpel src="BPELProcess2.bpel"></implementation.bpel> </component> <implementation.workflow src="Humantask1.task"></implementation.workflow> <td></td></pre>	
<pre><implementation.bpel src="BPELProcess2.bpel"></implementation.bpel> <implementation.workflow src="Humantask1.task"></implementation.workflow> <td></td></pre>	
<pre> <td></td></pre>	
<pre> <component name="Humantask1"> <implementation.workflow src="Humantask1.task"></implementation.workflow> </component> <component name="Humantask0"> <component name="Humantask0"> <implementation.workflow src="Humantask0.task"></implementation.workflow> </component></component></pre>	
<pre><implementation.workflow src="Humantask1.task"></implementation.workflow> <implementation.workflow src="Humantask0.task"></implementation.workflow> </pre>	
<pre> <td></td></pre>	
<pre><component name="Humantask0"> <implementation.workflow src="Humantask0.task"></implementation.workflow></component></pre>	
<pre><implementation.workflow src="Humantask0.task"></implementation.workflow></pre>	
<pre><component name="HumantaskBPEL"> </component></pre>	
<pre><implementation.workflow src="HumantaskBPEL.task"></implementation.workflow></pre>	

Figure 11–6 Setting the force.deploy Property in Oracle JDeveloper

11.4 Moving Running Composite Instances from One Revision to Another

This section discusses Instance Migration. This section includes the following topics:

- Section 11.4.1, "Overview"
- Section 11.4.2, "Important Considerations"
- Section 11.4.3, "How to Perform a Migration"
- Section 11.4.4, "Restrictions for Components"
- Section 11.4.5, "Ant Tasks Elements"
- Section 11.4.6, "Migration Elements"

11.4.1 Overview

Instance Migration allows you to migrate selected composite instances between existing compatible composite revisions or from an existing revision to a new revision. Compatibility is driven by the component compatibility within a composite. After migration is complete, the instance audit trail is visible under the new composite revision.

In contrast to Instance Patching, migration is not a part of deployment. It is a separate activity accomplished through Ant tasks and XML files that provide declarative specification of what instances migrate and what actions should be taken (for example, data and flow updates). You can select which instances to migrate.

11.4.2 Important Considerations

These are important considerations you must take into account before performing a migration:

- The same constraints apply when migrating as when performing instance patching. For example, you cannot move tokens to a non-interrupting boundary event.
- During migration, composite revision of a source instance is suspended so instances of that revision do not process incoming requests.
- This feature is implemented by using Ant commands.

These commands require the following artifact that is part of your installation:

/AS11gR1SOA/bin/ant-composite-instance-migration.xml.

Note that this artifact requires wlfullclient.jar in the following directory: \$FMW_ HOME/wlserver_10.3/server/lib.

To generate the Oracle WebLogic Server client library:

cd \$fmw.home/wlserver_10.3/server/lib java -jar wljarbuilder.jar

11.4.3 How to Perform a Migration

This section provides an overview of the steps you must perform to migrate an instance. The steps involved in migrating an instance are:

- 1. Identify the scope of instances to be migrated and create the composite filter to specify them.
- **2.** Run the migration feasibility report for the specified instances and review the output.

The output report includes which instances can be automatically migrated, which require a plan to accomplish the migration (for example, an active activity was removed), and which instance cannot be migrated.

- **3.** Develop the migration plan.
- **4.** Run the migration and review the output.

11.4.4 Restrictions for Components

The composite compatibility depends on the components defined inside the composite. If the change to any component is not compatible, then the composite is not eligible for instance migration. Only compatible changes in BPEL, BPMN, Human Workflow and Business Rules are accepted. Changes in any other components make the composite ineligible for instance migration.

This section describes the restrictions you must consider for each component. The section includes the following topics:

- Section 11.4.4.1, "BPEL Components"
- Section 11.4.4.2, "BPMN Components"
- Section 11.4.4.3, "Human Task Components"
- Section 11.4.4.4, "Business Rules Components"
- Section 11.4.4.5, "Mediator Components"

11.4.4.1 BPEL Components

The only eligible BPEL component for migration is sync process. The sync process does not include a durable activity like HumanTask, Wait or Receive.

For sync process instances, BPEL changes the revision for all the completed instances that are part of a running composite instance.

Any other BPEL process is not compatible for instance migration.

11.4.4.2 BPMN Components

The following changes to BPMN components are compatible with automatic instance migration:

- Add Scrip task
- Change data associations
- Add new data object
- Add any new activities except boundary events

The following changes to BPMN components are compatible with manual instance migration using migration plans:

Removal of an activity except for sub process, boundary event and merging gateways

The following changes are not compatible with instance migration:

- Delete Gateway (except Exclusive)
- Delete Subprocess
- Update Human Task implementation
- Update Subprocess Loop Characteristics
- Add a Boundary Event
- Update Boundary Event interrupting to non-interrupting
- Add an Event Subprocess
- Update Event Subprocess interrupting to non-interrupting
- Update Activity level (Move an activity inside a Subprocess or Gateway structure)
- Contains Call Activity

11.4.4.3 Human Task Components

Human workflow supports moving instances for all running and completed human tasks.

A human task stores the routing slip after initiating the task, so the task metadata change does not affect the running task.

11.4.4.4 Business Rules Components

There are no restrictions for Business Rules Components because the migration is executed synchronously.

11.4.4.5 Mediator Components

The following restrictions apply to Mediator components:

- Routing Rules: the older revision of Mediator must have only sequential routing rules.
- Mediator Instance State: the Mediator instance must be completed or faulted.

Message Patterns: the supported message patterns are request only or request response.

11.4.5 Ant Tasks Elements

The elements of the Ant Tasks are (see Figure 11–7):

- CompositeInstanceFilter: specifies the criteria to migrate the instance and defines the scope of the migration. This scope is determined by filtering composite instances based on:
 - compositeDN
 - domainName
 - compositeName
 - revision
 - label
 - ecid
 - conversationId
 - compositeInstanceId
 - tenantId
 - title
 - maxCreationDate
 - minCreationDate
 - maxModifyDate
 - minModifyDate
 - like
 - pageStart
 - pageSize
- Task to run feasibility report: The results are Not Possible, Automatic or Manual. You can review the report and change the criteria to reduce or remove some of the instances from the migration by modifying the CompositeInstanceFilter.
- Task to run migration

Figure 11–7 Example of Ant Task



11.4.6 Migration Elements

The migration elements are:

- **Migration Report**: describes the feasibility of migrating the proposed composite instances. The migration report includes:
 - Composite Instance Details
 - * Name: Component instance name
 - * Id: Component instance Id
 - * State: Composite instance state
 - * Migrate-ability: Automatic, Migration Plan required (and the required change plan, where applicable)
 - * Component Instance Data: Associated component instance details
 - Summary Data:
 - * The target revision
 - * Total number of matching composite instances
 - * Number of composite instances that can be automatically migrated
 - * Number of composite instances that cannot be automatically migrated
- Migration Result: describes the outcome of migrating composite instances:
- Migration Plan: provides the XML description of what to do with non-trivial composite or component changes. The migration plan is based on the MigrationPlan XML schema.

Migration Plan objects are created using MigrationPlanFactory.

Note: For more information on instance migration, navigate to the BPM Suite 11g Samples web page

[http://java.net/projects/oraclebpmsuitellg/pages/Samples] and review the demo samples listed. For a detailed description of the samples, download the PDFs available for each sample.

Part III

Using Oracle Business Process Management Process Spaces

This part contains the following chapters:

- Chapter 12, "Getting Started with Process Spaces"
- Chapter 13, "Using and Customizing a Process Workspace in Process Spaces"
- Chapter 14, "Using and Customizing a Process Modeling Space in Process Spaces"
- Chapter 15, "Using and Customizing a Process Instance Space in Process Spaces"
- Chapter 16, "Creating Custom Dashboards in Process Spaces"
- Chapter 17, "Performing Administration in Process Spaces"

Getting Started with Process Spaces

This chapter describes Oracle Business Process Management Process Spaces, how to start and log in to it, and how to navigate its interface, including the modeling space and the process instance space.

This chapter contains these topics:

- Section 12.1, "Understanding Process Spaces"
- Section 12.2, "Starting and Logging In to Process Spaces"
- Section 12.3, "Navigating Process Spaces"

12.1 Understanding Process Spaces

Process Spaces is a workspace built on top of Oracle WebCenter Portal: Spaces. It enables greater collaboration by combining the content from the panels in Process Workspace with an additional workspace in which you can collaborate in the development of process models.

WebCenter Portal provides group spaces that support discrete communities of any size that are organized around an area of interest or a common goal. WebCenter Portal also provides various collaborative services enabling document sharing, discussions, wikis, lists, links, tagging, and real-time collaboration services such as presence and instant messaging.

As a combination of WebCenter Portal and Process Workspace functionalities, Process Spaces provides WebCenter group spaces that are designed specifically for modeling and executing business processes. In addition to these, Process Spaces also uses the collaborative services of WebCenter Portal.

For information about Oracle WebCenter Portal, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

12.1.1 Oracle WebCenter Portal Group Spaces for Business Process Management

Process Spaces provides these WebCenter Portal group spaces specifically for business process management:

- A process workspace—For process participants to collaborate on relevant processes and instances based on their roles. This enables, for example, a group of sales representatives within a given region to share documents and information about various customers, the products they are buying, their requirements and history.
- A process modeling space—For domain analysts to collaborate on various aspects of modeling business processes. This enables, for example, experts from various

departments to jointly develop processes that reflect the needs of their various departments.

 A process instance space—Created on an ad hoc basis for long-running instances, it displays specific instances that require a high level of collaboration over time.

You can customize these group spaces based on functional teams, granting access only to a relevant set of users with the appropriate roles. After you specify these user sets, only the relevant processes are included in the space.

12.2 Starting and Logging In to Process Spaces

To start Process Spaces, you must have installed Oracle SOA Suite and the Service Integration server must be running. See *Oracle Fusion Middleware Installation Guide for Oracle SOA Suite and Oracle Business Process Management Suite* for more information.

To start and log in to Process Spaces:

1. Go to

http://host_name:port_number/webcenter/

The host_name is the name of the host computer on which Oracle WebCenter Portal is installed.

The port_number is the one used at installation.

2. Enter the user name and password.

You can use the preseeded user to log in as an administrator. If you have loaded the demonstration user community in the identity store, then you can use other users such as jstein or jcooper.

The user name and password must exist in the user community provided to JAZN.

3. Click Login.

12.3 Navigating Process Spaces

Tabs in Process Spaces are customizable depending on your roles—for example, an administrator can add or remove panels in any given page.

This section contains these topics:

- Section 12.3.1, "About the Process Spaces Process Workspace"
- Section 12.3.2, "The Modeling Space in Process Spaces"
- Section 12.3.3, "The Process Instance Space in Process Spaces"

12.3.1 About the Process Spaces Process Workspace

The Process Spaces Process Workspace includes a Home page and a Process Tracking page.

12.3.1.1 The Process Workspace Home Page

After a user logs in, depending on the user's roles and privileges, the Process Spaces Home page displays the user's tasks and related panels, as shown in Figure 12–1.

ipaces	Process Workspace - Mod	eling Space Ir	nstanceG	51	× I		
						Page Acti	ons + 🖉 Tags
	⊻ Task List						4.5.4
69	My Tasks Initiated Tasks	Administration 1	Tasks				
	Actions - View - (a) Assign	ee Me & Group 💙	Status	Assigned	✓ Search	1	Advanced
	Title	Number	Priority	Assignees	State	Created	Expires
	Vacation Request BPM	200001	3	VacationRegu	Assigned	Feb 22, 2010	
	E sequence task	200003	3	weblogic (U)	Assigned	Feb 22, 2010	
	sequence_task	200005	3	weblogic (U)	Assigned	Feb 22, 2010	
	E sequence_task	200007	3	weblogic (U)	Assigned	Feb 22, 2010	
	1						~
	• <						>
~		-0	VacationR	equestBPM21			
673			Process sequence_ Process	_task]			
	2						2
	V Recent Topics	6	9	Recent Documer	nts		_
	No forums were found.		-				68
		Process Workspace - Hod Task List Hy Task S This The Sequence_task Sequence_task Sequence_task C C Contrs Workload per process C C C C C C C C C C C C C C C C C C	Process Workspace	Process Workspace • Modeling Space Instance Task List Hy Tasks Instanced Tasks Actions • View • Actions • Actions • Actions • View • Actions • Actions • Actions • View • Actions • View • Actions • View • Actions	Process Workspace Process Workspace Process Workspace Process Process Workspace Process Proces	Process Workspace Process Workspace Process Workspace Process Workspace Process Workspace Process Process Process Process Process Process Process Process Process Proces Proces Process Process Process Process Proces Proces Proces	Process Workspace Vodeling Space InstanceGS1 Page Acto Page Acto Task List Hy Tasks Instanced Tasks Actors Very Acsignee Ves Gescript Vacation Request State Created Vacation Request State Vacation Vac

Figure 12–1 Process Spaces Home Page

Included in the various panels of the Home page is the Available Processes panel, which displays the processes the user is authorized to execute. The Home page also displays the task inbox and dashboards (charts), and several standard panels of WebCenter Portal.

12.3.1.2 The Process Workspace Process Tracking Page

Selecting the Process Tracking page provides process-related information, as shown in Figure 12–2. Depending on the user's roles and privileges, this page displays the process instance list and various other panels including available processes and dashboards. It also includes a number of standard WebCenter Portal panels.

Figure 12–2 Process Spaces Process Tracking Page

Home Process Tracking - Settings						Page	Actions + 🧭 T	ags @	ÐL
✓ Available Processes		Process Instance Li	ist						
	62	Title Insta	nce # Initiator	Process Name	Activities				^
[HelpDeskRequestBPM] Process v 1.0		Instance #2 of Pro 2		[sequence_task] P	UserTask	Actions -			
[VacationRequestBPM2] Process v1.0		Instance #3 of Pro 3		[sequence_task] P	UserTask	Actions -			
		Instance #4 of Pro 4		[sequence_task] P	UserTask	Actions -			
		Instance #1 of Pro 1	weblogic	[VacationRequestB.	EnterVacationReq	Actions +			
⊻ Members ♣ Add Send	Mai 🛛 🛪	< ✓ Charts						>	
Conan Doyle James Cooper Detricoper	<	Widget Workload per Pro	r process						<
Recent Activity Show Last 30 days Information No activity has been reported within the selected ti period.	me			Process (Sequence)	equest8PM2) task]			>	~
		Description of the second second			Description of the second second			_	

12.3.2 The Modeling Space in Process Spaces

The Process Spaces Modeling Space page enables users to collaborate on process definition. In this page users can, for example, on potential bottlenecks of a deployed process and discuss improvements. Figure 12–3 shows a Process Spaces Modeling Space page.



Figure 12–3 Process Spaces Modeling Space Page

This page includes the Process Catalog, which graphically displays the various processes in the project. Each process is represented by a square. Navigation arrows within a given square enable you to find more information about the process it represents. The page also includes a number of panels that are standard to WebCenter Portal.

12.3.3 The Process Instance Space in Process Spaces

The Process Instance Space allows users to collaborate on a process instance. Usually the process instance space is used for a long-running process instance that involves many ad hoc activities and collaboration.

Figure 12–4 shows a process instance page showing the stakeholders of the process instance, audit trail of the instance, and the process instance calendar.

ersonal Space	weblogic	My Group Spaces	Process Workspace	Modeling Space	Instance	G51 🗸			1		
me 🗸 🛛 Settings									D	Page Actions	🔹 🧭 Tags 🍘
Stake holders	VacationReques	stB	Audit Trial Instance #1 of Process Details Diath council							F-1 22 (22 h)	
			Status Running		Process P Number 1	rocess	questorm2]		reation time	: Peb 22 (23 nd	our(s) ago)
			Title						Drinriby	C+a	te.
			EnterVacationRequest						=	Run	nning
Recent Documer	its		Comments			∀A	tachments	•			
FolderTests.java FolderTests.java FolderTests.java FolderTests.java FolderTests.java FolderTests.java FolderTests.java		88 ()				F	ename	Desc	ription		
FolderTests.java			✓ Process Calendar						La se	100 100	
FolderTests.java			Everation Data		•	P 100	ay Februa	ry 2010	Assignee	Me & Group	×
BOOK 1. XISX		~	Assigned Date	Mon	1	TUE	2	ivea	3	4	rri
		A	Activity Start Date								
Recent Topics		Sia									

Figure 12–4 Process Spaces Process Instance Page

Using and Customizing a Process Workspace in Process Spaces

This chapter describes the various activities end users can do with Oracle Business Process Management Process Spaces, including acting on tasks, filtering tasks, adding members, adding documents, participating in discussions, initiating and canceling a process instance, and how to customize Process Spaces.

Because Process Spaces is tightly integrated with both Process Workspace and WebCenter Portal, using and customizing a work space sometimes involves performing tasks that are unique to Process Spaces. At other times, the tasks are performed by using either Process Workspace or WebCenter Portal.

This chapter contains these topics:

- Section 13.1, "The Process Workspace Home Page in Process Spaces"
- Section 13.2, "Acting on Tasks in Process Spaces"
- Section 13.3, "Filtering Tasks in Process Spaces"
- Section 13.4, "Adding Members to Process Spaces"
- Section 13.5, "Adding Documents to Process Spaces"
- Section 13.6, "Participating in Discussions in Process Spaces"
- Section 13.7, "Viewing Recent Activity in Process Spaces"
- Section 13.8, "Initiating a Process Instance in Process Spaces"
- Section 13.9, "Filtering Process Instances in Process Spaces"
- Section 13.10, "Canceling a Process Instance in Process Spaces"
- Section 13.11, "Adding Documents to Process Instances in Process Spaces"
- Section 13.12, "Using and Customizing Dashboards in Process Spaces"
- Section 13.13, "Creating an On-Demand Group Space for an Instance in Process Spaces"
- Section 13.14, "Customizing Process Workspace in Process Spaces"
- Section 13.15, "Subscribing to a Process Workspace Group Space in Process Spaces"

13.1 The Process Workspace Home Page in Process Spaces

Figure 13–1 shows an example of a Process Spaces Home page.

Personal Space weblogic My Group S	Process Workspace - M	odeling Space Ins	stanceG51	E I		_
Iome 🗸 Process Tracking Settings					Page Actions + 🧭 Tags 🤅	۲
Available Processes	Y Task List					
	My Tasks Initiated Tasks	Administration Ta	asks			
[HelpDeskRequestBPM] Process v1.0	Actions - View - 🚱 Ass	ignee Me & Group 🛩 S	Status Assigned	Search	Advanced	
[VacationRequestBPM2] Process v1.0	Title	Number	Priority Assignees	State Creat	ted Expires	
	Vacation Request BPM	200001	3 VacationRegu	. Assigned Feb	22, 2010	~
	sequence_task	200003	3 weblogic (U)	Assigned Feb	22, 2010	
	sequence_task	200005	3 weblogic (U)	Assigned Feb	22, 2010	=
	E sequence task	200007	3 weblogic (U)	Assigned Feb	22, 2010	
						Y
	• <				>	
Conan Doyle James Cooper John Partopart Burtopart Steinbeck	8	■ [Va Pro ■ [se Pro	acationRequestBPM2] ocess guence_task] ocess			111
period.	<				>	
	Recent Topics	0	Recent Docum	ents	69	m
			EnderTests.java		~	

Figure 13–1 Process Spaces Home Page

For more information about the Process Workspace Home page in Process Spaces, see Section 12.3.1, "About the Process Spaces Process Workspace."

13.2 Acting on Tasks in Process Spaces

In Process Spaces, you act on tasks by using the Process Workspace panel on the Process Spaces Home page.

For instructions on acting on tasks, see Chapter 2, "Working on Tasks in Process Workspace".

13.3 Filtering Tasks in Process Spaces

In Process Spaces, you filter tasks by using the Process Workspace panel on the Process Spaces Home page.

For instructions about filtering tasks, see Section 1.3.2.3.1, "How to Filter Tasks"

13.4 Adding Members to Process Spaces

For information about how to add members to a group space, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

13.5 Adding Documents to Process Spaces

For instructions about adding documents to Process Workspace, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

13.6 Participating in Discussions in Process Spaces

For instructions about participating in a discussion, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

Note: If you have imported a group space and want to create or select a forum, you must do so manually from the services page of the imported group space.

For information about choosing or creating a forum in a group space imported into Oracle WebCenter Portal, see Oracle Fusion Middleware Administrator's Guide for Oracle WebCenter Portal.

13.7 Viewing Recent Activity in Process Spaces

For instructions about viewing recent activity, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

13.8 Initiating a Process Instance in Process Spaces

In Process Spaces, you initiate a process instance by using the Process Workspace portlet on the Process Spaces home page.

For instructions about initiating a process instance, see Section 3.1, "Initiating Process Instances in Process Workspace"

13.9 Filtering Process Instances in Process Spaces

In Process Spaces, you filter process instances by using the Process Workspace portlet on the Process Spaces Home page.

For instructions about filtering process instances, see Section 3.2, "Searching for Process Instances in Process Workspace".

13.10 Canceling a Process Instance in Process Spaces

In Process Spaces, you cancel process instances by using the Process Workspace portlet on the Process Spaces Home page.

For instructions about canceling a process instance, see Section 3.5, "Canceling a Process Instance in Process Workspace".

13.11 Adding Documents to Process Instances in Process Spaces

A process instance group space is configured with the document service that Oracle WebCenter Portal offers. The widget added to this group space enables you to add and remove documents within the group space folder.

Note: Note that these documents are different from the ones attached to the process instance itself. The documents added in Process Spaces are generally collaboration documents and have no significance after the process is completed.

For instructions about adding documents to process instances in Process Spaces, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.*

13.12 Using and Customizing Dashboards in Process Spaces

For instructions about using and customizing dashboards in Process Spaces, see Chapter 6, "Using Dashboards in Process Workspace".

13.13 Creating an On-Demand Group Space for an Instance in Process Spaces

Process instances that typically run for a long time may require a dedicated collaboration space. To create a dedicated collaboration space, in either the task list or the process instance list, at the end of the row for the process instance, click **Launch Process Instance Space**. A page appears with the new process instance group space. Here, you can perform activities specific to that instance.

13.14 Customizing Process Workspace in Process Spaces

To enable collaboration on a specific set of processes, you can create a new group space by using the Process Workspace template. This creates a new group space that is a replica of the default Process Workspace.

The default Process Workspace can also be customized just like any group space.

To create a customized default group space from the Process Workspace template, customize and then save the template. To customize the template, first create a group space from it, then customize, save, and publish it as Process Workspace Template.

For information about customizing group space pages, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

13.15 Subscribing to a Process Workspace Group Space in Process Spaces

When Oracle Business Process Management Process Spaces is installed, users are not granted access to Process Workspace by default. To access the Process Workspace group space, the user logs in to Process Workspace and performs a search for discoverable spaces. From the search results, the user selects a Process Workspace and is asked to choose a role. Available roles are: Viewer, Participant, and Moderator.

The Participant and Viewer roles can be self-subscribed—that is, when the user chooses the role, the user is granted that role automatically. When the user chooses the Moderator role, that request is submitted to the administrator for approval.

Using and Customizing a Process Modeling Space in Process Spaces

This chapter discusses modeling spaces, those group spaces that enable business analysts to collaborate in developing process models. It also explains how to use Process Spaces to customize your own modeling space

This chapter contains these topics:

- Section 14.1, "About Modeling Spaces in Process Spaces"
- Section 14.2, "The Process Modeling Space Page"
- Section 14.3, "Customizing a Process Modeling Space in Process Spaces"
- Section 14.4, "Adding Members to the Modeling Space in Process Spaces"
- Section 14.5, "Adding Documents to the Modeling Space in Process Spaces"
- Section 14.6, "Participating in Modeling Space Discussions in Process Spaces"
- Section 14.7, "Subscribing to a Modeling Space in Process Spaces"

14.1 About Modeling Spaces in Process Spaces

A modeling space enables business analysts to collaborate on process models. This collaboration includes these activities:

 Browsing published projects by using the Process Catalog—The Process Catalog shows all published processes. When you find and double-click a specific process, it opens in Business Process Composer. From there you can collaborate with process developers and designers to edit the processes.

Note: If you browse the Process Catalog, and Single Sign-on is not set up, then you are prompted for the necessary credentials before being taken to Composer.

 Adding tasks by using Oracle WebCenter Portal—To track the progress of projects being developed, you can add tasks.

For information about adding tasks, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*

 Add documents by using Oracle WebCenter Portal—Use this Oracle WebCenter Portal feature when collaborating on process models. For information about adding documents, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.*

14.2 The Process Modeling Space Page

Figure 14–1 shows an example of a Process Spaces Modeling Space page.



Figure 14–1 Process Spaces Modeling Space Page

Table 14–1 Process Spaces Modeling Spaces Panels

Panel	Description
Tasks	Enables you to track the progress of projects being developed
Recent Documents	Lists the most recently created or modified files in the root folder of the active content repository connection to the page
Recent Topics	Lists recent discussion topics, popular topics, watched topics, and watched forums
Process Catalog	Enables you to browse the various projects. When you find and double-click a specific project, the selected project opens in Business Process Composer.

14.3 Customizing a Process Modeling Space in Process Spaces

To enable collaboration on a specific set of processes, you can create a new group space by using the Process Modeling template. This creates a new group space that is a replica of the default modeling space.

The default Process Modeling Space can also be customized just like any group space.

To create a customized default group space from the Process Modeling template, customize and then save the template. To customize the template, first create a group space from it, then customize, save, and publish it as Process Modeling Template.

For information about customizing group space pages, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

14.4 Adding Members to the Modeling Space in Process Spaces

For information about adding members to a modeling space, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.*

14.5 Adding Documents to the Modeling Space in Process Spaces

For information about adding documents to the modeling space, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

14.6 Participating in Modeling Space Discussions in Process Spaces

For information about participating in modeling space discussions, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

14.7 Subscribing to a Modeling Space in Process Spaces

For information about subscribing to a modeling space, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

Using and Customizing a Process Instance Space in Process Spaces

This chapter describes how to create, manage, and customize a process instance space. A process instance space is created for collaborating on long-running instances and displays specific instances that require a high level of collaboration over time.

Because Process Spaces is tightly integrated with both Process Workspace and WebCenter Portal, using and customizing a work space sometimes involves performing tasks that are unique to Process Spaces. At other times, the tasks are performed by using either Process Workspace or WebCenter Portal.

This chapter contains these topics:

- Section 15.1, "The Process Instance Space in Process Spaces"
- Section 15.2, "Adding Comments and Attachments to a Process Instance in Process Spaces"
- Section 15.3, "Tracking a Process Instance in Process Spaces"
- Section 15.4, "Creating a Calendar for a Process Instance in Process Spaces"
- Section 15.5, "Adding a Document to a Process Instance Space in Process Spaces"
- Section 15.6, "Participating in a Discussion in a Process Instance Space in Process Spaces"
- Section 15.7, "Customizing a Process Instance Space in Process Spaces"

15.1 The Process Instance Space in Process Spaces

Figure 15–1 shows an example of a process instance space.

ersonal Space weblogic My Group	Spaces	Process Workspace M	odeling Space Instanc	eGS1 +		×			_
me 🗸 Settings						Ţ	Page Actio	ons 🔹 🧭 Tags	@ Lir
v Stake holders	_ Ir	✓ Audit Trial stance ≠1 of Process ✓ Details							
		Priority normal Status Running	Process Number	[VacationF Process 1	RequestBPM2]	Creation to	me Feb 22 (2	3 hour(s) ago)	
		Title				Dringiby		Ctate	
		EnterVacationRequest				=		Running	-
Recent Documents		> Audit Trail ♥ Comments		▽.	Attachments	3			
FolderTests.java FolderTests.java FolderTests.java FolderTests.java FolderTests.java					riename	Description			
FolderTests.java FolderTests.java	1	Process Calendar Jue Date		D. T	oday Febru	ary 2010 Assigned	Me & Gro		-
Polder Lests. Java	_	Expiration Date	Mon	Tue	TCOTO	Wed	Thu	Fri	-
maldare and a second	~ 1	🗴 🚽 🗹 Assigned Date	1		2	3		4	Ę
Recent Topics	2	Activity Start Date							
	OEX 1	Activity bid bate							

Figure 15–1 Process Spaces Process Instance Page

Table 15–1 describes the elements in the process instance page shown in Figure 15–1.

 Table 15–1
 Panels in the Process Instance Space

Panel	Description
Stakeholders	Lists any user who has acted on the process, including the initiator
Recent Documents	Lists the most recently created or modified files in the entire root folder of the active content repository connection to the page
Recent Topics	Lists recent discussion topics, popular topics, watched topics, and watched forums
Audit Trail	Shows the details for this instance including any open instances. Enables you to add comments and attachments.
Process Calendar	Shows different dates associated with a process—for example: activity start date, activity end date, and human task dates such as assigned date, expiration date, and due date.

15.2 Adding Comments and Attachments to a Process Instance in Process Spaces

In Process Spaces, you add comments and attachments to a process instance by using the Process Workspace portlet on the Process Spaces home page.

For instructions on adding comments and attachments to a process instance, see Section 3.3.2, "Associating Comments and Attachments with a Process Instance in Process Workspace".

15.3 Tracking a Process Instance in Process Spaces

In Process Spaces, you a track a process instance by using the Process Workspace portlet on the Process Spaces Home page.

For instructions about tracking a process instance, see Section 3.3, "Viewing the Details of a Process Instance in Process Workspace".

15.4 Creating a Calendar for a Process Instance in Process Spaces

If you are an administrator or a user with permissions to edit the group space, you can add a calendar taskflow. This taskflow shows different activities—for example, assigned date, expiration date, due date for a task, activity start date, and activity end date for the process instance.

Note: Do not confuse this calendar with the one in Process Workspace. You cannot create a calendar for an organization unit in Process Spaces.

15.5 Adding a Document to a Process Instance Space in Process Spaces

For instructions about adding documents to a process instance space, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

15.6 Participating in a Discussion in a Process Instance Space in Process Spaces

For instructions about participating in discussion, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

Note: If you have imported a group space and want to create or select a forum, you must do so manually from the services page of the imported group space.

For information about choosing or creating a forum in a group space imported into Oracle WebCenter Portal, see Oracle Fusion Middleware Administrator's Guide for Oracle WebCenter Portal.

15.7 Customizing a Process Instance Space in Process Spaces

Like other templates, the process instance template can be customized. Unlike other templates, the process instance template cannot be used to create other group spaces manually. It is used by the system to create a group space that is connected to a specific process instance.

The created group space can be customized just like any other Oracle WebCenter Portal group space. However, because there is instance-specific data—namely, a process instance identifier link—you should not modify the group space attribute that is added as part of its creation.

Creating Custom Dashboards in Process Spaces

This chapter describes how to create and edit custom dashboards from within Oracle Business Process Management Process Spaces.

This chapter contains these topics:

- Section 16.1, "Understanding Custom Dashboards in Process Spaces"
- Section 16.2, "Creating a Custom Dashboard in Process Spaces"

16.1 Understanding Custom Dashboards in Process Spaces

Dashboards provide analytical information to monitor process execution as well as user workloads. They provide summary information.

From within Process Spaces, you can create and edit custom dashboards.

For information about custom dashboards and how they are created in Process Workspace, see Section 6.2, "Creating Custom Dashboards."

16.2 Creating a Custom Dashboard in Process Spaces

To develop a custom dashboard in Process Spaces, you use the Process Dashboard taskflow found in the WebCenter Portal Resource Catalog.

16.2.1 How to Create a Custom Dashboard in Process Spaces

You can create one or more custom dashboards in either a new or an existing custom page.

To create a custom dashboard in Process Spaces

- 1. In the Process Workspace home page of Process Spaces, from the Actions list, select either Edit page or Create a new page.
- 2. Click Add Content. The Add Content dialog appears.
- **3.** Open the Process Workspace Resources folder and add Process Dashboard. The Process Dashboard region now appears in Process Workspace.
- **4.** In the Process Dashboard region, click **Edit**. The Component Properties dialog box appears.
- **5.** In the Component Properties dialog box, next to the **Widget ID** field, select the spyglass icon. The Pick a Value dialog box appears.

- **6.** From the Pick a Value dialog box, you can either create a new data source or select an existing data source.
 - To create a new data source, in the Widget ID panel, select the **Add** icon. A Widget ID page appears in the right pane.
 - To select an existing data source, select from the list in the Widget ID panel. The corresponding page appears in the right pane.

For information about specifying the data source, see Section 6.2.1, "Specifying Graph Content."

After you have selected and specified the data source, click **Save and Close**. This returns you to the Component Properties dialog box. The **Widget Id** field is now filled in with the value for the data source you just selected.

7. Click **OK**. This returns you to the Process Dashboard region, which now contains the data source you just specified.

Note: If the space containing the custom dashboard is shared between different users, then the custom dashboard is visible only to users with permission to see the data corresponding to that dashboard. You grant permissions for the data source from Process Workspace.

Performing Administration in Process Spaces

This chapter discusses what you need to know as an administrator of Oracle Business Process Management Process Spaces, including roles and permissions, task flows, how to edit pages, and how to configure groups.

This chapter contains these topics:

- Section 17.1, "Understanding Roles and Permissions in Process Spaces"
- Section 17.2, "Understanding Task Flows in Process Spaces"
- Section 17.3, "Editing Pages in Process Spaces"
- Section 17.4, "Configuring a Group Space in Process Spaces"

17.1 Understanding Roles and Permissions in Process Spaces

Both Oracle WebCenter Portal and Oracle Business Process Management Process Spaces use an identity store, each with a different set of application roles. Process Spaces group spaces come with default WebCenter Portal roles—for example: moderator, participant viewer, and public user. However, an administrator can create more roles and provide appropriate permissions to them.

For more information about creating roles and providing permissions, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

17.2 Understanding Task Flows in Process Spaces

Task flows in Process Spaces display the same information available in Process Workspace. These are reusable components that have been consumed in Oracle WebCenter Portal so they can be used in Process Spaces.

For more information about the various task flows and their parameters, see Appendix A, "Creating Custom ADF Applications with Oracle Business Process Management Workspace Task Flows."

17.3 Editing Pages in Process Spaces

To use Process Spaces specific group spaces and templates, use the BPM Process Spaces Catalog. The BPM Process Spaces Catalog includes all the resources from the default group space catalog.

If you have upgraded from an earlier release and you want to edit the existing Process Spaces group space, you can either change the settings to use BPM Process Spaces Catalog for the particular group space or change the settings globally so that all group spaces in the application use the BPM Process Spaces Catalog. For information about how to use a resource catalog, see *Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces*.

BPM Process Spaces Catalog includes the following folders:

- Process Work Space Resources
 - Available processes: All available processes that a logged-in user can instantiate
 - Charts: Charts that show various statistics
 - Process instance list: All process instances a user can access
 - Task list: List of tasks that user has to act upon
- Process Instance Space Resources
 - Audit trail: Audit trail of a given instance in tabular and graphical forms
 - Process calendar: Calendar that shows various important dates for a given process instance. Also capable of showing dates for all process instances of a user in context.
 - Stake holders: List of stake holders for a given instance.
- Process Modeling Space Resources
 - Process catalog: Process catalog viewer that shows all published processes.

While there are no restrictions about using these task flows, they are grouped based on the group spaces to which they belong. However when a custom group space is created, you can mix and match the task flow.

For task flows to function correctly, you must pass the appropriate parameters to them. For configuring task flow parameters, see the Oracle WebCenter Portal documentation

For information about each task flow, see Appendix A, "Creating Custom ADF Applications with Oracle Business Process Management Workspace Task Flows."

You edit a page just as you would any other group space page. For more information, see Oracle Fusion Middleware User's Guide for Oracle WebCenter Portal: Spaces.

Each group space has a settings page for administering the group space. Process Workspace settings are configured to enable self-subscription for users. You can change these settings as needed. The rest of the default group space settings are similar to any other group space.

17.4 Configuring a Group Space in Process Spaces

For information about configuring a group space, see Section 17.3, "Editing Pages in Process Spaces".
Part IV Appendices

This part contains the following appendixes:

- Appendix A, "Creating Custom ADF Applications with Oracle Business Process Management Workspace Task Flows"
- Appendix B, "Customizing a Resource Bundle"
- Appendix C, "Customizing Applications in Process Workspace and the Worklist Application"
- Appendix D, "Deploying Images and JAR files as Part of a Shared Library JAR file"

A

Creating Custom ADF Applications with Oracle Business Process Management Workspace Task Flows

Different features available in Process Workspace are exposed as standalone reusable components, called task flows. This appendix provides basic information about the features in each task flow.

You can embed task flows in any Oracle Application Development Framework (ADF) application. These standalone task flows provide many parameters that enable you to build customized applications.

All the task flows are bundled in an ADF library that you can include in the application in which you are embedding.

Later, in Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows," a sample task flow using TaskListTaskFlow shows detailed steps for embedding the task flow in the application as well as customizing, deploying, and running it.

For descriptions of the various task flows, see Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management.

This appendix covers the following topics:

- Section A.1, "User-Related Task Flows"
- Section A.2, "Process Task Flows"
- Section A.3, "Process Analytics Task Flows"
- Section A.4, ".JAR files for the Task Flows"
- Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows"
- Section A.6, "Sample: Wf_client_config.xml File"
- Section A.7, "Sample: Bean with the Method Returning the WorkflowServicesClientConfigurationType Object"

Note: This appendix assumes that you are familiar with both Oracle Application Development Framework and security.

A.1 User-Related Task Flows

This section contains these topics:

- Section A.1.1, "Task List Task Flow"
- Section A.1.2, "Certificates Task Flow"
- Section A.1.3, "Reports Task Flow"
- Section A.1.4, "Application Preferences Task Flow"
- Section A.1.5, "Flex Fields Task Flow"
- Section A.1.6, "Rules Task Flow"

A.1.1 Task List Task Flow

The task flow takes in the parameters to control how the embedded region displays. This section discusses the following parameters:

- Section A.1.1.1, "federatedMode"
- Section A.1.1.2, "federatedServers"
- Section A.1.1.3, "showServerColumn"
- Section A.1.1.4, "wfCtxID"

For the full list of parameters, see *Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management*.

Figure A–1 Process Workspace Tasks Page

	Tasks Process Tracking Standard Dashboards Hore -	関 Home	Administration Preferences Help Logout weblogic
My Tasks(0) • G G Actions • 🤮 Assignee Me & My Group 🖤	State Assigned 💌 Search	Advanced	le constante la const
Bi Tibe		Process Name	Creator Phonty Assigned
	Please select a task to see the details		
and a second sec			

A.1.1.1 federatedMode

If you set this parameter to true, then the task list is shown in the federated mode. For the task flow to run in federated mode, you must pass the list of federated servers to the task flow. For instructions about how to pass a list of federated servers to the task flow, see Section A.1.1.2, "federatedServers".

A.1.1.2 federatedServers

To pass the federated servers list to the task flow, do one of the following:

Provide the client configuration file, wf_client_config.xml, in the class path The class path is either APP-INF\classes\wf_client_config.xml at the ear level or the

WEB-INF\classes of the web application. The client configuration file contains all federated server details.

For a sample .wf_client_config.xml file, see Section A.6, "Sample: Wf_client_ config.xml File".

 Construct a JAXB object that contains the federated servers list. This JAXB object can be passed to the task flow through the federatedServers parameter. The object should be returned by a backing bean method.

For a sample bean with the method returning the WorkflowServicesClientConfigurationType object, see Section A.7, "Sample: Bean with the Method Returning the WorkflowServicesClientConfigurationType Object"

If you provide both the client configuration file wf_client_config.xml and the JAXB object to the task flow, the JAXB object takes precedence.

A.1.1.3 showServerColumn

If the task flow is run in federated mode, the server column in the task list is not shown by default. To show the server column in the task list, this parameter must be passed as "true".

A.1.1.4 wfCtxID

This is a workflow context token string. It is used to create workflow context inside the task flow.

Note: If the application is SSO-enabled or is secured using ADF security, this parameter is not required. Otherwise, this parameter is required.

To get the workflow context identifier:

Run the following:

```
IWorkflowContext wfCtx =
wfSvcClient.getTaskQueryService().authenticate(username,
```

password,

realm,

null);

wfCtxID = wfCtx.getToken();

A.1.2 Certificates Task Flow

A digital certificate contains the digital signature of the certificate authority so that anyone can verify that the certificate is real. A digital certificate establishes the participant's credentials. It is issued by a certification authority. It contains your name, a serial number, expiration dates, a copy of the certificate holder's public key (used for encrypting messages and digital signatures), and the digital signature of the certificate-issuing authority so that a recipient can verify that the certificate is real

As Figure A–2 shows, when you want to sign a decision, you can upload your digital certificate. When signing a task outcome by using your certificate, you must upload not only the one certificate issued to you by the certification authority but the entire chain of certificates. Upload the chain as a .P7B (PKCS7 format) file.

Preferences	Upload Certificate		
User Preferences Certificates Rules Notification Preferences Accessibility	Browse for Certificate Certificate file type PKCS7 PKCS12 Select Certificate File Certificate Password Type or Paste Certificate Contents	Browse	k
	Upload Reset		

Figure A–2 Uploading a Certificate

To create an application and embed the Certificates task flow in the application, follow all the steps of Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows" except Step 9. This is because certificates have no parameters.

A.1.3 Reports Task Flow

For information about the Reports task flow, see Section 7.6, "Using Reports to Manage Workflows in Process Workspace."

To create an application and embed the Reports task flow in the application, follow all the steps of Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows" except Step 9. This is because Reports have no parameters.

A.1.4 Application Preferences Task Flow

For information about the Application Preferences task flow, see Section 9.2, "Setting the Display of the Work Items Panel in Process Workspace".

To create an application and embed the Application Preferences task flow in it, follow all the steps of Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows" except Step 9. This is because application preferences have no parameters.

A.1.5 Flex Fields Task Flow

For information about the Flex Fields task flow, see Section 9.4, "Managing Mapped Attributes (Flex Fields) in Process Workspace".

To create an application and embed the Flex Fields task flow in it, follow all the steps of Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows" except Step 9. This is because flex fields have no parameters.

A.1.6 Rules Task Flow

For information about the Rules task flow, see Section 2.6.1, "How to Create User Rules in Process Workspace".

The **showOtherUsersRules** parameter takes a Boolean value. When it is passed as True, other users rules are displayed; when it is passed as False, other user rules are not shown. In addition, this user must have the required permission to view other user rules.

To create an application and embed the Rules task flow in it, follow all the steps of Section A.5, "Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows" except Step 9. This is because rules fields have no parameters.

A.2 Process Task Flows

This section contains these topics:

- Section A.2.1, "Process Application Task Flow"
- Section A.2.2, "Process Instances Task Flow"
- Section A.2.3, "Process Instance Details Task Flow"

A.2.1 Process Application Task Flow

The applications available in Process Workspace depend on the process design and the roles assigned to the user currently logged in. The list of applications available also depends on the BPM processes that are deployed and running. The applications panel will be empty if no processes are running, or if the user is not able to execute an application in any of the processes currently running. Figure A–3 shows the location of the Applications panel in Process Workspace.

Figure A–3 Applications Panel

ORACLE Business Pre	ocess Workspace					Logged in as weblogic	Home Administration	Preferences Help	p Logout O
Tasks Process Tracking	Standard Dashboards							» Hide	Tabs 🤮 🗙
🗹 Applications 🛛 🔞	Actions - Views - 🙌	Search		🖉 🕑 Advanced	12		R		
[Project1] HelloWorld v1.0 ∠ Views (List filters)	Title Number Instance #1 of Gate# 1 Instance #2 of HelloV 2 Instance #3 of HelloV 3 Instance #4 of HelloV 4	Initiator weblogic weblogic weblogic	Process Name [MyBPMTest] Gate [Project1] HelloWe [Project1] HelloWe [Project1] HelloWe	Initiated ewa Jun 14, 2010 12:48 i oridi Jun 14, 2010 1:47 Pi oridi Jun 14, 2010 2:14 Pi oridi Jun 15, 2010 3:17 Pi	Status PG NG				
Inbox inbox	Instance #4 of HelloWork	d 		Process [Proj Number 4	ect1] HelloWorld	Creat	ion time 6/15/2010		<u> </u>
	♥ Open Activities Title UserTask								=
	✓ Audit Trail ③ Tabular ○ Graphical								
	Activity		Loop count	Event	R	tesponsible Date		Thread ID	_
	Start	C	0	Activity completed		Jun 15, 201	.0 3:17:47 PM	0	
	Start	C	0	Instance created		Jun 15, 201	0 3:17:47 PM	0	
	Start	G	0	Instance left the activ	vity	Jun 15, 201	0 3:17:47 PM	0	
	▽ UserTask	2	0	Activity processing		Jun 15, 201	0 3:17:47 PM	0	
	UserTask	(P)	0	Instance entered the	activity	Jun 15, 201	0 3:17:47 PM	0	
	4								>

Table A–1 Process Applications Task Flow

Field	Description
Task Flow definition file:	processApplicationsTaskflow.xml
Description:	Renders the list of applications available to the user, that is, the list of tasks that, when executed, will create instances in the different deployed processes.
Arguments:	None

A.2.2 Process Instances Task Flow

Field	Description			
Task Flow definition file:	processInstancesTaskflow.xmll			
Description:	Renders the list of instances corresponding to a certain view (by default, Inbox view) in a table format. The columns that this table will show are defined by the presentation associated to the selected view, which can be edited from here. It also provides search capabilities.			
Arguments:	viewId: specifies the view that will be rendered initially. By default, Inbox is assumed.			

A.2.3 Process Instance Details Task Flow

Field	Description				
Task Flow definition file:	processInstanceDetailTaskflow.xmll				
Description:	Renders the instance details, including open activities, audit trail (both tabular and graphical), comments and attachments.				
Arguments:	instanceId: Task Flow will render the details associated to this instance ID.				

Table A–3 Process Instances Task Flow

Figure A–4 Work Items Details Panel



A.3 Process Analytics Task Flows

This section contains these topics:

- Section A.3.1, "Dashboard Widget Task Flow"
- Section A.3.2, "Application Roles Task Flow"
- Section A.3.3, "Organization Units Task Flow"
- Section A.3.4, "Parametric Roles Task Flow"
- Section A.3.5, "Business Calendar and Holiday Lists Task Flow"
- Section A.3.6, "Extended User Properties Task Flow"

A.3.1 Dashboard Widget Task Flow

Field	Description	
Task Flow definition file:	widgetTaskflow.xml	
Description:	Renders a widget, that is, a graph displaying information stored in Cubes. This widget could be one of the out of the box ones (like Workload per Participant, Performance per Process, and so forth) or a custom one.	
Arguments:	widgetId: The ID of the widget that will be rendered.	
	showSelector: A Boolean value (true/false) stating whether the task flow will render a drop down listing all available widgets, so that the user can select and change from one widget to other.	

 Table A–4
 Dashboard Widget Task Flow





A.3.2 Application Roles Task Flow

Application Roles task flow displays the list of application roles defined in the system for application context OracleBPMProcessRolesApp. The user can search for application roles, select a role and add members (users, groups or other roles) to it or even define business calendar-organization unit mappings for the role.

Members Administrators Administrators UsinesPra Dontracts Dontract	.	Description BPM application admin role, ha Calendars	as full privilege for performing any op	Context OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest OracleBPMProcessRolest	Арр Арр арр арр арр арр арр арр
Administrators UninesPra U		BBM application admin role, ha	as full privilege for performing any op	perations includ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/ OracleBIMProcessRoles/	App App
alesRep	× 2.	Glendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	крр крр крр крр крр крр крр крр крр крр
luisineaPra porrovera Contractas Luomatich Luionatich Luionatich LuionaePra LuionaePra LuionaePra Luionaetich TrocessRofera PMProcessRolesApp s Trype Tators Group	× •	Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	крр крр крр крр крр крр крр крр крр крр
oproven Contracts Uutomaticki BaleReg. I: webbgic Uutomaticki Uuto	× 2.	Glendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App App App App App App App
Indrateds Unsmallch Unsmallch Unsmallch Unsmallch Unsmallch Unsmallch TrocessOve		Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App App App App App Apply Re
utomaticH usinesPra process processAdmin PMProcessRolesApp s Type rators Group	<u></u>	Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App App App App App Apply Re
s starses Type starses	₽ ≈	Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App App App App Apply Re
IusinesPra Joprovers Contracts, J Iudomaticki Homaticki PMProcessRolesApp s Type rators Group	1	Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App App App Apply Re
pprovers_ contracts_1 uutomaticH krocessOw PMProcessRolesApp s 	× 1	Calendars		OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles OracleBPMProcessRoles	App App App App Apply R
Indirates_1 vicensesOw rocessAdmin PMProcessRolesApp s Type rators Group	S 2	Calendars		OracleBPMProcessRoles/ OracleBPMProcessRoles/ OracleBPMProcessRoles/	App App App Apply R
vutomaticH roccessOw PMProcessRolesApp s Type rators Group	S 8	Calendars	_	OracleBPMProcessRoles	App App Apply R
rocessOw rocessAdmin PMProcessRolesApp s Type rators Group	S 20	Calendars			App Apply R
rocessAdmin PMProcessRolesApp s Type rators Group	× •	Calendars		<u>ما</u>	pply R
rocessAdmin PMProcessRolesApp s Type rators Group	3 %	Calendars	_	<u>_</u>	oply R
rocessAdmin PMProcessRolesApp s Type rators Group	₩	Calendars	_		Apply R
PMProcessRolesApp s Type rators Group	≧ %	Calendars	_		
s Type rators Group	× 🕤	Calendars			
Type rators Group			S 😫 💥		
rators Group		Calendar	Organization Unit	1	
ators Group				-	

Figure A–6 Role Details Panel

A.3.3 Organization Units Task Flow

This task flow enables users to create new organization units or edit existing ones.

For information about the Organization Units task flow, see Chapter 10, "Managing Your Organization in Process Workspace".

Figure A–7 shows an Organization Units details panel.

Figure A–7 Organization Units Details Panel

ecercity (<u> </u>	Details :	SOA			
rganization Units	<u>9</u> - X	* Name	SOA			
FMW SOA WEBCENTER	D	escription	ENTW: SQA group			
		Calendar	(F)	62		
			Add Users/Groups/App Role	+×	Add Organization Role	+ 🗙
			jausten	1. II. 1		
			jcooper Jacober			
		Members	istein			
			jverne			
				+×		
		Managers				

A.3.4 Parametric Roles Task Flow

For information about the Parametric Roles task flow, see Section 10.6, "Managing Parametric Roles in Process Workspace".

A.3.5 Business Calendar and Holiday Lists Task Flow

For information about the Business Calendar and Holiday List task flow, see Section 10.3, "Managing Calendar Rules in Process Workspace" and Section 10.2, "Managing Holiday Rules in Process Workspace".

A.3.6 Extended User Properties Task Flow

For information about the Extended User Properties task flow, see Section 10.7, "Managing Extended User Properties in Process Workspace".

A.4 .JAR files for the Task Flows

JAR File	Location
adflibTaskListTaskFlow.jar	<pre>\$FMW_HOME/Oracle_WC1/soa/modules/oracle.bpm.spaces_11.1.1</pre>
oracle.bpm.workspace-adflib.jar	This .jar file is packaged only with a Process Spaces installation. It includes the following libraries that you must specify in your custom ADF application:
	 adf.oracle.domain
	• oracle.bpm.runtime
	 oracle.bpm.client
	 oracle.soa.workflow.wc
	 oracle.bpm.webapp.common
	 oracle.bpm.workspace
	 oracle.bpm.projectlib
adflibWorkspaceTaskFlows.jar	<pre>\$FMW_HOME/AS11gR1SOA/soa/modules/oracle.soa.worklist_11.1.1</pre>

Table A–5 JAR Files for Task Flows

A.5 Samples: Customizing ADF Applications with Oracle Business Process Management Workspace Task Flows

This section contains these topics:

- Section A.5.1, "Sample 1: Creating an Application with an Embedded Tasklist Task Flow"
- Section A.5.2, "Sample 2: Creating an ADF Page with Task Flows"

A.5.1 Sample 1: Creating an Application with an Embedded Tasklist Task Flow

The sample explains how to create an application and embed the TaskList task flow in it.

The Tasklist ADF library jar file adflibTaskListTaskFlow.jar must be in the class path. This jar is available in Oracle JDeveloper in the BPM Worklist Components library.

What follows are the steps to embed the task flow in your application. You can use these same steps to create a new sample.

To embed the Tasklist task flow in your application:

1. Create a new Fusion Web Application in Oracle JDeveloper and name it TaskListTaskFlowSample.



Figure A–8 Creating New Oracle Fusion Middleware Application

2. Add the task flow jar adflibTaskListTaskFlow.jar in the project's class path. Include the BPM Worklist Components library from the Libraries and Classpath section of the JDeveloper and add the library. This library also contains the adflibWorklistComponents.jar that must also be in the classpath.



Figure A–9 Adding BPM Worklist Component Library

3. Add the BPM Services library in the class path.

Figure A–10 Adding BPM Services Library

Oracle Developer 11g Developer	wait.Build	Tread in Tails Tree Sample)	we street mittoffer (pr		5.2
Ele Edt Yew Application Refact	or Search	h Savigate Build Bun V	ereigening Lools Window Help		(A · Lewith
Becketon Becketon	INTA	eno moi la ventione applicatione	inn voi 15 Sanoietean Inva 175 Taditat Tad Ploy Sanoie O	werverw BitestSande box	T Il Consecut Palette
TaditatTadificectangle + 12 -	84 - 2	tow * Full Screen Size *		B/UEENNE.	achibTasktastTaskflow.tw
- Projects 🔂 🌚 🖓 - 🖅 -		Contraction of the second s			
a (2) Model Controller Controller Controller Controller Controller		Project Properties - C.S	nyWork/Taild.toTaildFlow/Simple/ViewControllerWiew	wController.gpr) Task Hows Regions
B S view		60 30 Mt	Ubraries and Classpath		Lig tasklat-task-now-definition
itestSamplePageDef.aml		Project Source Paths ACF Model ACF View	Use Quston Settings Use Project Settings		
🗟 🔄 Web Content		iii Act	Java SE Version:		
8 WEB-DVF		ai Business Components	[1.6.0_13 (Defa.R)	Chagge	
testSample sipx	1	iii Complex	Classpath Entries:		
		Depkoyment	Export Description	Add Ugrary	
		E38 Module	Commons Collections 3.1	6dd MR/Directory	
		Extension III Javadoc	9 10 15F 1.2 9 10 15TL 1.2	Bemove	
		- Java EE Application - SP Tan Libraries	ADE DVT Faces Runtime	Vige	
		- 35P Visual Editor	If 15P Runtime	y are to	
	1	Ubraries and Gasspath	ACF Faces Dynamic Components	Month	
		ai Maven Resource Runde	Oracle 201 Oracle 201 Oracle 201 Oracle 201	Prive Sg.	
		- Run(Debug/Frofile - Technology Scope	Bresource Bundle Support Bresource Bundle Variable Resolver Bresource Bundle Variable Resolver Bresource Bundle Variable Resolver Bresolver Bresource Bundle Variable Resolver Bresource Bundle	6	Chrosely Inpector
The second s		Help		OK Carcel	3
* Application Resources	Design 1				
Descriptors	Messa	ges - Log			
	These 1 (file Beso Reso Trin W2D2) These 1 (file file	<pre>tmms were added to ViewC (/C)/Oracla/MLdilewase/) (/C)/Oracla/MLdilewase/) (/C)/Oracla/MLdilewase/) unce Fundle Support from unce Fundle Surphylic Viriable Bus (ded Fundime from/Task teams were added to ViewC (/C)/Oracle/MLdilewase/) (/C)/Oracle/MLdilewase/)</pre>	enter til at - jog devel ögen / inte gerat i om / ikb/adflikber klist Gragons devel ögen / inte gerat i om / ikb/adflikber fakbist /Fakbist i at Tauht Joy/Yadfont eller //Wadfont el //Takkist i at Tauht Vorgen / inte jog / Takkist i at Tauht Vorgen / inte jog Sist Tauht Ziow/YawGont eller //Wadfont eller // inte / orgen devel ogen / inte gerat i om / ikb/adflikber klist Gragons devel ogen / inte gerat i om / ikb/adflikber klist Gragons	ente.jar uv.jar Lier.jpr YeeCongroller.jpr *054.jar Mo54.jar	
Data Controls 🛛 🖓 🏹	ľ				
Recently Opened Hiles	Messages	Extensions Feedback		141	
landers and a state of the stat	- Constanting	and the second se		- Benny 200	CALL OF A DESCRIPTION O

4. Add the WSRP Container library in the classpath.



Figure A–11 Adding the WSRP Container Library

- **5.** In case you are running your application on non-soa server, you need to perform two additional steps that are documented in the *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.
 - a. Install the oracle.soa.workflow shared library. If your server already has oracle.soa.workflow.wc installed, then you do not need to install oracle.soa.workflow.

For information about installing the oracle.soa.workflow shared library, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

b. Configure foreign JNDI on the non-SOA server.

For information about configuring a foreign JNDI on the non-soa server, see *Oracle Fusion Middleware Programming JNDI for Oracle WebLogic Server*.

Note: If you are running the Task List task flow in federated mode, then you do not need to do this step. See the parameter "federatedMode" on how to use task flow in federated mode.

6. Create a JSPX file testSample.jspx.



Figure A–12 Creating a a JSPX File

7. Choose adflibTaskListTaskFlow.jar from the component palette. It contains the list of all the Task Flows.



Figure A–13 Choosing adflibTaskListTaskFlow.jar

8. Drag and drop one of the Task Flow onto the JSPX page as a region. Figure A–14 shows, by way of example, the TaskListTaskFlow being dropped onto the JSPX page as a region.



Figure A–14 Oracle JDeveloper with TaskList Task Flow on the JSPX Test Page

When you drag a task flow onto a JSPX page, a popup shows a list of the available parameters. For a description of these parameters, see Oracle Fusion Middleware Modeling and Implementation Guide for Oracle Business Process Management.

The Edit Task Flow Binding dialog box appears.

9. In the Edit Task Flow Binding dialog box, pass the task flow parameters.

	(号・)会話通知・(▶・●・)山	(+ Search
Application Application	ns spx 👘 websge-application.uni 🕼 SampleBean java 🔛 Taskist TaskPowSample Overview 🔄 tast Sample.jspx 🖉 💷 🖉 Com	gonent Palette
TaskListTaskFlowGample • 🔂 • 😡 • Show	w*fulScenn Sex* 2	eskt.istTaskPlow.jar
Projects Q Q Q P - F P Model Q NewController Q Application Sources Q Wew	8 - ≣ ≣ ≣ ≣	Plans ons KLat-tasi-Row-definition
DataBindings.cpx	Edit Task Flow Binding	
RestSamplePageDef. and	Task Flow (WEI-39/hask.ist-task-flow-definition.uni#task.ist-task-flow-definition	
afn mi	Input Parameters Rep:	
Web Content	Input Parameters	
CT CT FILLING		
affecting and	Nane Value	
adic-config.mi faces-config.stel	Name Value Roderstroffstore Roderstroffstorers	
Carde-configural faces-configural brandad.configural web.int Page Rows	None Valar PoderateRide Dran RederateRiderarus ExclorenceSuture Dran	
aft-configurit face-configurit face-configurit web.ati balance and configurit aft-configurit aft-config	None Valar Holderstaffrede trus Holderstaffrede trus Anderstaffrede trus Anderstaffrede trus Anderstaffrede trus Anderstaffrede trus Holderstaffrede trus Holderstaffred	

Figure A–15 Passing Task Flow Parameters

🖞 TaskListTaskFlowSample 🛛 🔹 强 •	- Shun - Full Screen Sea * 回 · · · · · · · · · · · · · · · · · ·	offb/TaskLotTaskPlow.jar
Anasta Hodd Hodd Medican Sources Accloant Sources Accloant Sources Medicant So	Form	Task Finns Regions Task Finns Task Finns Task Finns Task Finns
	Edit Task Flow Binding	
	Tash Poor / with PMP haskist-tash-Row-definitionsniftsalcast-tash-Row-definition Input Parameters New / Parameters	
	Verice Verice Moderated/arrows fore Moderated/arrows	Chapter basets Chapter basets C ≠ 2 ≠ 2 ≠ (→ 100 → 2)
And other Designment		
Appacation Resources	Design Source (Bridings Prevew History 1)	
	There items was added to YimoContriller.jpr (nie ////Washindidianas/jereidger/integration/ib/addinbhukinsComponents.jpr his/s///Washindidianas/jereidger/integration/ib/addinbhukinsComponents.jpr herecere bundle Bugers from/TadakinsTadaYimo/TweeGeneriller/JuseGeneriller.jpr herecere bundle Tadakin Energier Fran/TadakinsTadaYimo/TweeGeneriller/Juse franka Juniza i Itras/TadakinsTadaYimo/TweeGeneriller/Juse franka Juniza i Itras/TadakinsTadaYimo/TweeGeneriller/Juse franka/Juse franka/Juse/Juse/Juse franka/Juse/Juse franka/Juse/Juse franka/Juse/Juse franka/Juse/Juse franka/Ju	
Data Controls 👌 🏹) Recently Opened Files	Messages Estansors Fredback	

The following entry is created in the testSamplePagedef.xml.

```
<taskFlow id="taskListtaskflowdefinition1"
taskFlowId="/WEB-INF/taskList-task-flow-definition.xml#taskList-task-flow-defin
ition"
xmlns="http://xmlns.oracle.com/adf/controller/binding">
<parameters>
<parameter id="federatedMode" value="true"</pre>
xmlns="http://xmlns.oracle.com
/adfm/uimodel"/>
<parameter id="showServerColumn" value="true"</pre>
xmlns="http://xmlns.oracle.com/adfm/uimodel"/>
</parameters>
</taskFlow>
```

10. Add the following shared libraries in weblogic-application.xml. If you have oracle.soa.workflow.wc installed on your server, add that library by specifying the following.

```
<library-ref>
library-name>oracle.soa.workflow</library-name>
</library-ref>
```

- **11.** Open and edit the WAR deployment profile for editing. To do this:
 - a. Right Click ViewController.
 - **b.** Select Deployment.

- **c.** Select WAR File.
- d. Click Edit.
- e. Check adflibTaskListTaskFlow.jar, adflibWorklistComponents.jar and wsrp-container.jar.

file tilt yew application Refactor Search Savigate Build Bus Ver ng Iools W **m**tTaskFlowSample.3vs 1 • B • A• M 42 ianple 007.5. Crast version+'1.0 Edit WAR Deployment Profile Pro -Filters De ¥ 12.0.8 nuces_1.6.3ar ng_1.0.4.3ar share.ta Expand All Nodes Collapse All Nodes OK Cancel ttelp

Figure A–16 Editing the WAR Deployment Profile

12. Secure the application using ADF Security by navigating **Secure** then **Configure ADF Security...** from the application-level menu as Figure A–17 shows.



Figure A–17 Navigating to Secure the Application

The Enable ADF Security dialog box appears.

13. Select ADF Authentication as shown in Figure A–18.



Figure A–18 Selecting ADF Authentication

Click Next. The Select Authentication Type dialog box appears.

Note: If you choose ADF Authentication and Authorization, do the following:

- 1. In the Application Resources Descriptors META-INFO folder, open jazn-data.xml.
- 2. Select Resource Grants.
- 3. Select Task Flow type.
- 4. Select Show task flows imported from ADF libraries.

The taskflows are listed in the Resources listing.

- 5. Select the appropriate taskflow that you have consumed in the application—for example, taskList-task-flow-definition (For the task drill functionality to work, taskList-task-flow-definition also requires permission for /taskDetail-task-flow-definition).
- 6. Add Granted to selected role for the "view" permission.
- **14.** Select HTTP Basic Authentication as shown in Figure A–19.



Figure A–19 Selecting HTTP Basic Authentication

For information about enabling ADF security in a Fusion web application see Oracle Fusion Middleware Fusion Developer's Guide for Oracle Application Development Framework.

Click Finish.

Note: If you are using the task flow in federated mode, do the following:

- Pass the list of federated servers to the task flow. For more information, see Section A.1.1.2, "federatedServers."
- Enable global trust between the federated servers. This is done so that the already ted user token is passed to all the federated servers. Do the following:
 - **1.** Login to the Oracle WebLogic Server console.
 - 2. Under Domain Structures, select the domain name base_domain. The domain name may be different if a non-soa server is used. You can rename the domain name during configuration.
 - **3.** Select the Security tab.
 - **4.** Click the Advanced link (near the bottom **Save** button.)
 - 5. Give some password in the "Credential:" field. (The same password must be given for all the federated servers).
 - 6. Click Save.
 - **7.** Restart the server.
- **15.** Create an EAR deployment profile, build the EAR, and deploy it. To deploy the EAR, do the following:
 - **a.** From the application-level menu, navigate to **Deploy** then **TaskListTaskFlowSample_Appl...**.

	······································	(de-Search
Application (JEPM Project Navi. 1(6	Application1 Overview	Component Palette
Projects MendControling Application Sources Property Outsindings cox Property Outsindings cox	Hew Brujett. Black Application Copen Project. Clock Application Reing Application Reing Application. Store Cyrtise Chapter State Chapter State Chap	Task Flows Task Rives Apprefs-task-flow-definition Apprefs-task-flow-definition proys-task-flow-definition program-task-flow-definition program-task-flow-definition taskissi-reports-task-flow-definition taskissi-reports-task-flow-definition
Application Resources	Replace With	
Descriptors	Application Properties.	
🗃 🦾 META-INF	jsp.root •> af.document#d1 •> af:form#f1 •>	~
ips-config xm1	Design source Bindings Preview History -	
ADF META-INE soft-config.xml	These are in the project as a result of your AOF ja krsp-container; jar (file:/scratch/nisabhar/jdevboe, Bpu-services.jar (file:/scratch/nisabhar/jdevboe, AdfilbUcetstdattorC.jar (Jar not found:,./, AdfilbUcetstdattorC.jar (Jar not found:,./, AdfilbUcetstdattorC.jar (Jar not found:/,./, AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar (IntervicetstattorC.jar) AdfilbUcetstattorC.jar) AdfilbUcetstattorC.jar)	r taperts: //develope: Ref/service //buckets: harativeCon Tabric-runi r/devhose, r/devhose, r/devhose, Rich cl: Rich cl:
APRIL AND A APRIL APRIL		

Figure A–20 Deploying the Application

The Deployment Action dialog box appears.

b. Select **Deploy to Application Server** as shown in Figure A–21.

Figure A–21 Deploying to Application Server



Click Next. The Select Server dialog box appears.

 Select the server. Deselect Deploy to all Server instances in the domain. Click Next.

The Server Instances dialog box appears.

- 17. Select the server instance and click Finish.
- 18. Access the following link, specifying the proper values for server and port. http://server:port/TaskListTaskFlowSample-ViewController-context-root/f aces/testSample.jspx.

A login popup appears. Log in as any user. The task list appears for the logged-in user as shown in Figure A–22.

If the task flow is used in federated mode, then the list of servers appear in the Show list. Selecting any combination of servers from the list refreshes the task list to show all the tasks belonging to the selected servers. If you passed the parameter showServerColumn as true, then the server column appears in the task list and indicates the server to which the task belongs.

Figure A–22 Process Workspace Task List

	Nozilla Firefox									0.62
<u>Die</u>	Edit Yew History Bookes	arks Yahoot Tools !	940							
6	- C × 👁	http://sta00048.us	orade.com/7101/TestFe	der al e Cier-a	vsAttribute-1	/ewCoreiske	context-rook/Faces/	test. ppv?_ahLoop=15852961	14336227356_WW 🖄 + 📰+ Commence	ş
ile e	lost Wated 📄 Hy Yahoor 📑 Vi	ahool Bookmarks 🛄 Yake	of Downloads 🛄 Vaho	er Phat 🗋 Vi	shoet 🗋 w	verlagt Bugs - a		CS 📄 HY FRED BUGS 🗋 R	Federated Worklast	
0	RACLE			Search =	å Aria Sea		ugld + 🗣 Problem	ndd + F Farm sob(Job80)	· B Options	
Stor	4	· Actions +	Autoree Me 6 0	map with 3	Lettel Assig	ried	· Seath	Advanced		
Title	1 All		Server	Number	Priority	Assignment	State	Created	Expres	
-	P Pranan Helouros	274	Halidon Resource	200005	1	(stein (U))	Assigned	3J 27, 2009 8:40 AM		
Ť	Compensation Revenue for XXXX	111	Human Resource	2000007	1	istee 0.5	Autored	34 27, 2009 8:42 AM		
E	Required Action : Expense Report	Them JODOPER , USD 425	Perancials	200008	1.	jatmin (U)	Assigned	34 27, 2009 B:38 AM		
0	M1 : Expense Report From 30EC	HD was rejected	Pinanciale	200009	1	jataan (U)	Assigned	3.4 27, 2009 8:39 AM		
Dane										
2	1111 J J B & "	Distant for viter in	The Worklet com	0.00	store Media		Widnes Course	Orade Evention	P.2.534, Televier	02 00 - B 14 PM
			LTH	1 cm		Lente			0	Monday

Click on any of the tasks title links to see the task details in a new browser window.

A.5.2 Sample 2: Creating an ADF Page with Task Flows

This sample shows how to create an ADF page and add three task flows to the page.

The page will be divided into two parts. If the user selects TaskFlow from left side, the corresponding task flow appears on the right side.

To create the ADF page with task flows:

- 1. Follow Steps 1 to 6 of Section A.5.1, "Sample 1: Creating an Application with an Embedded Tasklist Task Flow.".
- **2.** Configure the JSPX page as follows:
 - a. In Component Palette, select ADF Faces.
 - **b.** Drag and drop the Panel Stretch Layout on to the JSPX page.
 - **c.** Delete Top, Start, Bottom, and End from the JSPX page. The JSPX page will look like Figure A–23.



Figure A–23 Configuring the JSPX Page

3. Drag and drop the Panel Splitter from the Component Palette into the JSPX page. The JSPX page now appears as shown in Figure A–24.

Figure A–24 Split Panel in the JSPX Page

file fait View Application Ref.	actor Search Sa	wigate Build Bun Versioning	Jooh Window H	lo .	10 A			
	0.0-1	- 411	-	7 .0			• Search	_
Application X CaPH Project Havin	10 0	Test into administra			•	Component	Palette	
Application2	- 19	- Show - Sull Screen Size -	Note .	daut - Dinne - B. A	RIUEE	ADF Faces		
Applications Appl		👔 * Sheer Full Screen Sizer		fuit Vitron V	VB/VEE^	ADF Faces Common Co Co Common Co Co Common Co C	imponents ibs for for clan bard Layout	o
Application Resources						Contractions	p Layout - vertical -	n_ (
# ADF META-INF		Design Source Bindings Preview	W History	ingenerepsi •/itain •/a	transide out-introducted of C	Halipp	colefaultic (start)	
		Destatement - Log Plansatement - Log Plansatement Collector Participation Collector Participation Collector Participation Collector Participation Collector Participation	0	Valign • Layout	<default> (middle) vertical</default>	•		
	[G11417 Ad] Write Wie April Cartin Hundis to <u>Accretchinistaburileristaturines</u> [G11417] Adj Write Wie April Cartin Hundis to <u>Accretchinistaburileristaturines</u> [G11417] Adj Harbirgh Advintaturine [G11417] Adj Harbirgh Advintaturine [G11414] [G1167] Adj Harbirgh Advintaturines [G11414] [G1167] [G1167] [G1167] [G1167] [G1167] [G1167] [G1167] [G11414] [G1167] [G1167] [G1167] [G1167] [G1167] [G1167] [G1167] [G11414] [G1167] [G		loctr/www.rk/Applicat r/ldcveloptr/www.rk/A lication2_application lication2_application	# Appearance # Style and TI # Debaying	sense			
Data Controls	87	[01:24:54 AH] The following [01:24:54 AH] http://10.229 [01:24:54 AH] Elapsed time [01:24:54 AH] = Deploym	UPL context root(.151.12217001/App1 for deployment1 3 wht finished,	s) were defined and can t ication2-ViewController- 8 seconds	be used as a starting	Landmark: PartialTrigg	<default> inon</default>	•
Recently Opened Files	-	4			2	Customizat	lon	
	11.5	Messages Extensions Dep	loyment		(4) 7.10			
control for all har game laper / nework /A	pariention2/VewCo	nireler/public_nin()est_spi	and the second se			-		Web Line
🐨 🛛 😹 Shell - Konsole	IVN VN	C config	· nisabhar	Oracle J	Developer 11g Release 1 -	1	101	

4. In the navigator, right-click the View Controller and create a new Java Class. Name it SampleController.java.

Add private transient UIComponent rightPanel = null; with getters and setters as shown in Figure A-25.



Figure A–25 Creating SampleController.java

5. In the navigator, select the Faces-config.xml file. Specify the Managed Bean, naming it as sampleControllerBean as shown in Figure A-26.

Figure A–26 Specifying the Managed Bean



- 6. Drag and Drop Panel Stretch Layout on to the Second Facet. Delete Top, Start, Bottom, End from it. And Specify a binding for PanelStrechLayout from the Property Inspector as #{sampleControllerBean.rightPanel}.
- 7. Drag and drop a link from the Component Palette into the JSPX page. In the Property Inspector specify the text as Rules. In SampleController Java file, add the following code:

private String selectedLink = "RULES"; with getter and setters
public void displayRules(ActionEvent event){

```
setSelectedLink("RULES");
if(getRightPanel() != null){
RequestContext.getCurrentInstance().addPartialTarget(getRightPanel());
}
}
```

Add the Action Listener property for the rules as follows:

#{sampleControllerBean.displayRules}

8. Drag and drop a link from the Component Palette into the JSPX page. In the Property Inspector, specify the text as Certificates. In the SampleController Java file, add following code:

```
public void displayCertificates(ActionEvent event) {
```

```
setSelectedLink("CERTIFICATES");
```

```
if(getRightPanel() != null){
```

RequestContext.getCurrentInstance().addPartialTarget(getRightPanel());

} }

Add the Action Listener property for the certificates as follows:

#{sampleControllerBean. displayCertificates }

9. Drag and drop a link from the Component Palette into the JSPX page. In the Property Inspector, specify the text as Reports. In the SampleController Java file, add following code:

```
public void displayReports(ActionEvent event) {
```

setSelectedLink("REPORTS");

```
if(getRightPanel() != null){
```

RequestContext.getCurrentInstance().addPartialTarget(getRightPanel());

} }

Add the Action Listener property for the reports as follows:

#{sampleControllerBean. displayReports}

10. Drag and drop a switcher from the Component Palette into the center facet of the second facet.

In the Property Inspector, specify:

- the facet name as #{sessionScope.sampleControllerBean.selectedLink}"
- the default faces as RULES

Application Phillip Project Maria	And and an other state of the s				
Anthermore and a list	SampleContro	ollerjaca Madminjopic 🕕	Component Palette		6
Application2	• 🔁 • 🖓 • Shaw* Full Screen Size *	None • Default	ADF Faces		
Constant Constant	Ruter Experts Experts	Santoer -	Common Components Components Series Common Entron Lerout Panel Dathboard Panel Form Lucout Coperations	ndar .	(
adfc-config.xml			13 × 11 /	C. Fred	9.830
in trivitad-corfig xml web xml web xml in Page Flows in trijpy Application Resources	9		+ Id Rendered @Facettiame: * Defaultfacet: Eindnos	11 (cdefault> drug) (#jsessenScope sampleControllerdkan selectedLink) (#LES	•
Corrections Desconcer MITA-BH MITA-BH MITA-BH Jps-config.xml mi webbogr-application.xml mi MITA-KH	.cet ▼) af panelstretchlunusf#p Design [Source (Binding) Presi	122 •) ffacer, •) afa witcher #5 ev History (<)	AnnbuteChangeListen	e [~
Des Centrós B Recentr Opines Fires	Determent Image: Constraint of the second seco	profiles profiles profiles pricationication Solute to plication BUS2[peration deploy on BUS4[peration deploy ballydeperation deploy ballydeperation deploy ballydeperation pallydeperations pallydeperations pallydeperations pallydeperations pallydeperations pallydeperations the deployment the deployment pallydeperations the deployment pallydeperations			
and the second se	Messages Extensions De	glbyment (4)	101		1000

Figure A–27 Adding and Specifying a Switcher

- Inside af:switcher define a facet and name it f:facet name="RULES". Inside this facet, drag and drop Rules-Task-Flow-Definition from the Component Palette (inside adflibTaskListTaskFlow.jar).
- 12. Inside af:switcher define another facet and name it f:facet name="CERTIFICATES". Inside this facet, drag and drop Certificates-Task-Flow-Definition from the Component Palette (inside adflibTaskListTaskFlow.jar)
- 13. Inside af:switcher define another facet and name it f:facet name="REPORTS". Inside this facet drag and drop Reports-Task-Flow-Definition from the Component Palette (inside adflibTaskListTaskFlow.jar)
- **14.** Save your work.
- **15.** Follow Steps 10 to 18 of Section A.5.1, "Sample 1: Creating an Application with an Embedded Tasklist Task Flow."

A.6 Sample: Wf_client_config.xml File

This section provides a sample of the client configuration file, Wf_client_config.xml.

If you don't want to include the server in the federated servers list, put excludeFromFederatedList="true" in the <server> element.

The client type must be specified explicitly on the <server> element in the XML.

- If you are using a remote client, enter clientType="REMOTE".
- If you are using a SOAP client, enter clientType="SOAP".

In Example A–1, REMOTE client type is used.

Example A–1 Wf_client_config.xml

```
<?xml version="1.0" encoding="UTF-8" standalone="yes"?>
<workflowServicesClientConfiguration
xmlns="http://xmlns.oracle.com/bpel/services/client">
<server name="default" default="true" excludeFromFederatedList="true"
```

```
clientType="REMOTE">
      <localClient>
<participateInClientTransaction>false</participateInClientTransaction>
      </localClient>
      <remoteClient>
         <serverURL>t3://sta00048.ex.example.com:7001</serverURL>
<initialContextFactory>weblogic.jndi.WLInitialContextFactory</initialContextFactor</pre>
v>
<participateInClientTransaction>false</participateInClientTransaction>
      </remoteClient>
      <soapClient>
         <rootEndPointURL>http://sta00048.ex.example.com:7001</rootEndPointURL>
         <identityPropagation mode="dynamic" type="saml">
            <policy-references>
               <policy-reference enabled="true" category="security"</pre>
                                 uri="oracle/wss10_saml_token_client_policy"/>
            </policy-references>
         </identityPropagation>
      </soapClient>
   </server>
   <server name="ERP" clientType="REMOTE">
      <soapClient>
         <rootEndPointURL>http://sta00147.ex.example.com:7001</rootEndPointURL>
         <identityPropagation mode="dynamic" type="saml">
            <policy-references>
               <policy-reference enabled="true" category="security"
                                 uri="oracle/wss10_saml_token_client_policy"/>
            </policy-references>
         </identityPropagation>
      </soapClient>
      <remoteClient>
         <serverURL>t3://sta00147.ex.example.com:7001</serverURL>
<initialContextFactory>weblogic.jndi.WLInitialContextFactory</initialContextFactor</pre>
у>
         <participateInClientTransaction>false</participateInClientTransaction>
      </remoteClient>
   </server>
   <server name="CRM" clientType="REMOTE">
      <soapClient>
         <rootEndPointURL>http://sta00048.ex.example.com:7001</rootEndPointURL>
         <identityPropagation mode="dynamic" type="saml">
            <policy-references>
               <policy-reference enabled="true" category="security"
                                  uri="oracle/wss10_saml_token_client_policy"/>
            </policy-references>
         </identityPropagation>
      </soapClient>
      <remoteClient>
         <serverURL>t3://sta00048.ex.example.com:7001</serverURL>
<initialContextFactory>weblogic.jndi.WLInitialContextFactory</initialContextFactor</pre>
v>
         <participateInClientTransaction>false</participateInClientTransaction>
      </remoteClient>
   </server>
</workflowServicesClientConfiguration>
```

A.7 Sample: Bean with the Method Returning the WorkflowServicesClientConfigurationType Object

This section provides a sample: bean with the method returning the WorkflowServicesClientConfigurationType object.

Note: You must set the client type explicitly while building the JAXB object. For example:

```
ServerType server1 = new ServerType();
```

server1.setClientType("REMOTE"); // If REMOTE client is to be used.

Or:

server1.setClientType("SOAP"); // If SOAP client is to be used.

If you do not specify anything for the client type, the default is SOAP.

package view;

```
import java.util.List;
```

```
import oracle.bpel.services.workflow.client.config.IdentityPropagationType;
import oracle.bpel.services.workflow.client.config.PolicyReferenceType;
import oracle.bpel.services.workflow.client.config.PolicyReferencesType;
import oracle.bpel.services.workflow.client.config.RemoteClientType;
import oracle.bpel.services.workflow.client.config.ServerType;
import oracle.bpel.services.workflow.client.config.SoapClientType;
import
oracle.bpel.services.workflow.client.config.WorkflowServicesClientConfigurationTyp
e;
import weblogic.nodemanager.common.ServerType;
public class WorklistBackingBean {
 WorkflowServicesClientConfigurationType wscct = null;
 public WorklistBackingBean() {
   super();
  }
 public WorkflowServicesClientConfigurationType getWSCC() {
   if (wscct == null) {
     wscct = new WorkflowServicesClientConfigurationType();
     List<ServerType> servers = wscct.getServer();
      /**** Setting default server in the list ****/
      ServerType defaultServer = new ServerType();
     defaultServer.setClientType("SOAP");
      servers.add(defaultServer);
      defaultServer.setDefault(true);
      //If this is true, the server is excluded from federated server list
      //By defalut it is false
      //defalutServer.setExcludeFromFederatedList(true);
      defaultServer.setName("default");
     RemoteClientType rct = new RemoteClientType();
      rct.setServerURL("t3://defaultHost:port");
      rct.setInitialContextFactory("weblogic.jndi.WLInitialContextFactory");
      rct.setParticipateInClientTransaction(false);
```

```
defaultServer.setRemoteClient(rct);
  SoapClientType sct = new SoapClientType();
  PolicyReferencesType prts = new PolicyReferencesType();
  PolicyReferenceType prt = new PolicyReferenceType();
  prt.setEnabled(true);
  prt.setCategory("security");
  prt.setUri("oracle/wss10_saml_token_client_policy");
  prts.getPolicyReference().add(prt);
  IdentityPropagationType ipt = new IdentityPropagationType();
  ipt.setMode("dynamic");
  ipt.setType("saml");
  ipt.setPolicyReferences(prts);
  sct.setRootEndPointURL("http://defaultHost:port");
  sct.setIdentityPropagation(ipt);
  defaultServer.setSoapClient(sct);
  /****** Setting Federated Server 1 to the list ****/
  ServerType server1 = new ServerType();
  server1.setClientType("SOAP");
  servers.add(server1);
  server1.setName("Human Resource");
  RemoteClientType rct1 = new RemoteClientType();
  rct1.setServerURL("t3://HumanResourcesHost:port");
  rct1.setInitialContextFactory("weblogic.jndi.WLInitialContextFactory");
  rct1.setParticipateInClientTransaction(false);
  server1.setRemoteClient(rct1);
  SoapClientType sct1 = new SoapClientType();
  PolicyReferencesType prts1 = new PolicyReferencesType();
  PolicyReferenceType prt1 = new PolicyReferenceType();
  prt1.setEnabled(true);
  prt1.setCategory("security");
  prt1.setUri("oracle/wss10_saml_token_client_policy");
  prts1.getPolicyReference().add(prt1);
  IdentityPropagationType ipt1 = new IdentityPropagationType();
  ipt1.setMode("dynamic");
  ipt1.setType("saml");
  ipt1.setPolicyReferences(prts1);
  sct1.setRootEndPointURL("http://HumanResourcesHost:port");
  sct1.setIdentityPropagation(ipt1);
  server1.setSoapClient(sct1);
}
return wscct;
```

}

Customizing a Resource Bundle

This appendix explains how an administrator of either Oracle Business Process Management Workspace or Oracle BPM Worklist can add or modify strings shown in the application by creating a custom resource bundle.

Note: Create a temporary directory at any location, for example, /scratch/jstein/sharedLib. This will contain all the individual JAR files as well as the shared library JAR file, custom.client.applib.jar. Do this step before doing any of the following steps.

B.1 How to Customize a Resource Bundle

This section tells you how to customize a resource bundle.

To customize a resource bundle:

 Create an Oracle Application Development Framework Project in Oracle JDeveloper. Figure B–1 shows the Create Fusion Web Application (ADF) dialog box.



Figure B–1 Creating an ADF Project in JDeveloper

2. Create a Java class and name it CustomResourceBundle. Figure B-2 shows the Create Java Class dialog box.

💙 Create J	ava Class		×
Enter the	details of your new class.		
<u>N</u> ame:	CustomResourceBundle		
<u>P</u> ackage:	view		Q
<u>E</u> xtends:	java.lang.Object		Q
Optiona	al Attributes		
İmplei	ments:		⊹ ×
-Acce	ss Modifiers	-Other Modifiers	
⊙ p	<u>u</u> blic		
⊂ p	ac <u>k</u> age protected) <u>a</u> bstract () <u>f</u> inal	
ro⊃ 🔽	n <u>s</u> tructors from Superclass		
imp ⊡ <u>M</u> ai	ilement A <u>b</u> stract Methods n Method		
Hel	p	ОК	Cancel

Figure B–2 Creating a Java Class

3. Include Worklist Application JAR files. Figure B–3 shows the BPM Worklist Components library being added.

To include the Worklist Application JAR files:

- **a.** From the Project Properties panel, select Libraries and Classpath.
- **b.** Select Add Library.
- c. From the Libraries list, select BPM Worklist Components.
- d. Click OK.



Figure B–3 Including Worklist Application in the Library

4. Modify the CustomResourceBundle class. The class should look as in Figure B-4:

Figure B–4 Modifying the CustomResourceBundle Class

👸 Cu	stomRe	sourceBundle_en.java 🗴 👶 WorkflowResourceBundle_ar.java 🗴 🕌 WorkflowResourceBundle.java 🗴 👶 AppPreference:
60-	Find	- 小合) 🖄 🄌 🦧 🖓 🖓 🖓 🖓 🐺 🕽 🐘 🐂 🐂 🗎 🗎
6		<pre>public CustomResourceBundle_en() {</pre>
7		super();
8 0		}
10		/**
11		* Since this method is being overwritten, it will make the super.getContents() invisible
12		* to the caller. Hence we need to copy all the resources (contents) here and concatenate
13		* with the newly added resources and return the resultant 2D Object array.
14		*
15		* @return
10		7 while Object D D get(enterts() (
18		public object] gettorcents() {
19		Object parentKeys = super.getContents();
20		<pre>int superLength = parentKeys.length;</pre>
21		<pre>int length = contents.length;</pre>
22		Object [] result = new Object[superLength + length][2];
23		$\operatorname{Int} 1 = 0;$
24 25		for $(i = 0; i < superlength; i +) $
26		result[i] = parentKevs[i]:
27		}
28		
29		<pre>for (int j = 0; i < superLength + length; i++, j++) {</pre>
30		result[1] = contents[];
31		3
33		return result:
34		}
35		
36		<pre>static final Object [] contents =</pre>
37		{ "LABEL_WURKLISI_CUSIUM_IIILE", "EN_LUSTOMIZEd WORKLIST IIIE" }, { "LABEL_WURKERACE (USTOM IIIE", "EN_LUSTOMIZEd Workloppe Tile" }.
38 30	ι	{ LABEL_WORKSFACE_COSTOM_ITILE , EN_COSCOMIZED WORKSPACE (TTTE }, };
	л Г	

Make sure that this class extends the correct WorklistResourceBundle and that the getContents() method is implemented the same way in all the supported language bundles.

5. Create various custom resource bundles for other language support.

You can create various custom resource bundles with different names under the same package. When you do so, you must append the locale to the file name. For example, for English US, your Custom Resource Bundle name will be CustomResourceBundle_en.java. Make sure you implement the getContents() method the same way as described in Step 4.

Note: By default, you need to create both the file CustomResourceBundle.java and the file CustomResourceBundle_ en.java. This is because we use the English (_en) resource bundle by default during the application login. Otherwise, login would fail.

6. Create a new JAR deployment profile as shown in Figure B–5.

Figure B–5 Creating a Deployment Profile for a JAR File

Project Properties - /home/cnkira	n/mywork/Customization/ViewController/ViewController.jpr	O X
(Search	Deployment	
⊞ Project Source Paths	○ Use <u>C</u> ustom Settings	Customize Settings
⊞ ADF Model		
ADF View	Deployment Profiles:	
Create Deployment Pr	file	<u>E</u>dit
Co Click OK to create you	new deployment profile and immediately open it to see its configu	<u>N</u> ew
De <u>Archive Type</u>		Delete
EJB JAR File		•
Ext Name:		
ResourceBundle		
ISP Description:		
JSP Lib Re: Ru Te	eploying a simple archive to the file system. The JAR file consists of	
Help	OK	
Help	OI	Cancel

7. Make sure your JAR deployment profile looks like what is shown in Figure B–6.
| Edit JAR Deployment Profile Proper Search JAR Options File Groups Project Output Contributors Filters | Ites Filters Files Patterns This file group includes the project output directory as a contributor. You may need to compile the project to see all files coming from the output directory. Building Contents of This File Group's Contributors |
|---|---|
| Profile Dependencies | Image: Data Image: Data |
| New Dele <u>t</u> e | Expand All Nodes |
| Help | OK Cancel |

Figure B–6 Verifying JAR Deployment Profile

8. Edit the JAR file location to point to the newly created directory as shown in Figure B–7.

Figure B–7 Editing the JAR Deployment Profile

🗎 Search	JAR Options	
JAR Options	JAR File:	
File Groups	/scratch/cnkiran/sharedLib/ResourceBundle.jar	owse
Contributors	✓ Compress Archive	
····· Filters Library Dependencies	Compression Level Default	
Profile Dependencies	Include Manifest File (META-INF/MANIFEST.MF)	
	Main Class:	owse
	Additional Manifest Files to Merge into MANIFEST.MF	
		<u>x</u> dd
)elete
New Delete		

9. Deploy the resource bundle as a JAR file as shown in Figure B–8.



Figure B–8 Deploying the Resource Bundle

The file is then deployed to the pre-defined location, namely, the shared library directory.

For more information about resources bundles and how to customize them, see *Oracle Fusion Middleware Developer's Guide for Oracle SOA Suite*.

С

Customizing Applications in Process Workspace and the Worklist Application

This appendix explains how, for Process Workspace, you can create customized external applications and links that become available in the Favorites list of the Views, Applications, and Favorites pane. It also explains how, for both Process Workspace and the Worklist Application, you can specify the columns that appear in the inbox.

Note: Create a temporary directory at any location, for example, /scratch/jstein/sharedLib. This will contain all the individual JAR files as well as the shared library JAR file, custom.client.applib.jar. Do this step before doing any of the following steps.

C.1 How to Customize an Application

To customize and application:

 Create an Oracle Application Development Framework Project in Oracle JDeveloper. Figure C–1 shows the Create Fusion Web Application (ADF) dialog box.



Figure C–1 Creating an ADF Project in JDeveloper

- 2. Create a Java class and name it WorkspaceCustomisationImpl.
- **3.** Include Worklist Application JAR files. Figure C–2 shows the BPM Worklist Components library being added.



Figure C–2 Including Worklist Application in the Library

 Create a customization class. This sample class modifies the inbox columns and customize external application and links. For example: WorkspaceCustomisationImpl and its contents as follows:

```
package view.customisationimpl;
import java.util.ArrayList;
import java.util.HashMap;
import java.util.List;
import java.util.Map;
import oracle.bpel.services.workflow.client.IWorkflowServiceClient;
import oracle.bpel.services.workflow.runtimeconfig.IRuntimeConfigService;
import oracle.bpel.services.workflow.runtimeconfig.model.AttributeLabelType;
import
oracle.bpel.services.workflow.runtimeconfig.model.AttributeLabelUsageList;
import oracle.bpel.services.workflow.runtimeconfig.model.AttributeLabelUsages;
import oracle.bpel.services.workflow.verification.IWorkflowContext;
import oracle.bpm.ui.customization.CustomLink;
import oracle.bpm.ui.customization.IBPMUICustomizations;
public class WorkspaceCustomisationImpl implements IBPMUICustomizations {
    private static Map displayNameMap = new HashMap();
    public WorkspaceCustomisationImpl() {
        displayNameMap.put("instanceId", "Instance Id");
        displayNameMap.put("protectedTextAttribute1", "Business Status");
    }
    public List<CustomLink> getCustomGlobalLinks() {
        CustomLink globalLink1 =
            new CustomLink("Oracle Home Page", "www.oracle.com", null);
        CustomLink globalLink2 =
           new CustomLink("Self Services Application",
"http://global-ebusiness.example.com/",
                           null);
```

```
CustomLink globalLink3 =
               new CustomLink("BUG DB", "https://bug.example.com/", null);
            List<CustomLink> globalLinks = new ArrayList<CustomLink>();
            globalLinks.add(globalLink1);
            globalLinks.add(globalLink2);
            globalLinks.add(globalLink3);
            return globalLinks;
        }
        public String getColumnNames() {
            return "title,taskNumber,instanceId,creator,protectedTextAttribute1";
        }
        private static void initDisplayMap(IWorkflowServiceClient client,
            IWorkflowContext context) {
            // you can use service to load all label names for text attributes
            if (displayNameMap == null) {
                synchronized (String.class) {
                    if (displayNameMap == null) {
                        displayNameMap = new HashMap();
                        try {
                            IRuntimeConfigService service =
                                client.getRuntimeConfigService();
                            AttributeLabelUsageList list =
                                service.getAttributeLabelUsages(context, "Text");
                            List<AttributeLabelUsages> list1 =
                               list.getAttributeLabelUsages();
                            for (AttributeLabelUsages usage : list1) {
                                AttributeLabelType type = usage.getLabel();
                                displayNameMap.put(type.getTaskAttribute(),
                                                   type.getLabelName());
                            }
                        } catch (Exception exc) {
                        }
                    }
               }
            }
        }
        public String getColumnDisplayName(IWorkflowServiceClient client,
                                           IWorkflowContext context,
                                           java.lang.String colName) {
            initDisplayMap(client, context);
            return (String)displayNameMap.get(colName);
        }
5. Create a JAR deployment profile. Create a JAR file and name it
```

```
customer-impl.jar.
```

6. Deploy it as a part of the shared library, custom.client.applib.jar. To do this, update the MANIFEST.MF file and update the shared library as explained in Appendix D, "Deploying Images and JAR files as Part of a Shared Library JAR file".

Depending on your customization, you can now see its effects.

If your customization is for Process Workspace and involves either creating an external application or specifying inbox columns in the Task Details pane or both, you will see the following:

• Your custom application listed in the Links panel of the Process Workspace Home page as shown in Figure C–3.

Figure C–3 Links Panel in Process Workspace



• The columns of the Task Details inbox adjusted according to your specifications as shown in Figure C-4.

Figure C–4 Customized Columns in Task Details Pane

ORACLE Business Pro	Tasks Process Tracking Standard Dashboards More -	
Views Applications Links	My Tasks(2) 🗢 🕂 🥒 💥 🛛 🖾 Actions 🗸 🗍	🔍 🗸 🕴 Me & My Group 🗾 Assigned 🗾 🛃 👻
Orrada Marra Barra	Title	Number Instance Id Creator Business Status Process
Oracle Home Page	ToDo Subtask	200028 weblogic
Self Services Application	💼 💀 ToDo Task	200027 weblogic
BUG DB		

Deploying Images and JAR files as Part of a Shared Library JAR file

If you have customized an application, a skin, or a resource bundle, you need to deploy it as a JAR file within a shared library JAR file. This appendix explains how to do this.

D.1 How to Deploy Images and JAR Files as Part of a Shared Library JAR File

In this section, we use an example in which you include in a shared library JAR file the files you previously created:

- customSkin.jar as described in Section 9.2.7, "How to Choose a Skin in Process Workspace"
- customer-impl.jar as described in Appendix C, "Customizing Applications in Process Workspace and the Worklist Application"
- ResourceBundle.jar as described in Appendix B, "Customizing a Resource Bundle".

To deploy images and JAR files as part of a shared library JAR file:

- 1. Create a temp directory, mkdir temp.
- 2. Unzip all the jar files into the temp folder.

```
unzip customSkin.jar -d temp/
unzip customer-impl.jar -d temp/
unzip ResourceBundle.jar -d temp/
```

After you do this, you can see a META-INF folder included in the temp folder.

Note: You can create an images directory inside the temp directory and copy your customized logo file into it. Then from the application preferences, you can refer to it as /images/customizedLogo.jpg.

3. Create a MANIFEST.MF file under META-INF directory. This file should contain the following:

```
Manifest-Version: 1.0
Ant-Version: Apache Ant 1.7.1
Created-By: 19.1-b02 (Sun Microsystems Inc.)
```

Implementation-Vendor: Oracle
Implementation-Title: Custom Client Applib
Implementation-Version: 1.0
Product-Name: Custom Client Applib
Specification-Version: 1.0
Extension-Name: custom.client.applib

4. To ensure that all required class files and other skin files are available in the shared library, package everything into the file custom.client.applib.jar. To do this, navigate to the temp directory and run the following command:

jar -cvfm custom.client.applib.jar META-INF/MANIFEST.MF *

You have now created the shared library under the temp directory. It looks like this: temp/custom.client.applib.jar.

- **5.** Deploy custom.client.applib.jar on the Oracle WebLogic Server as a shared library. To do this: From the Oracle WebLogic Server Console, click **Deployments and Install**, navigate to the JAR file location, and deploy.
- **6.** Restart the Worklist Application or Process Workspace from the Oracle WebLogic Server. Figure D–1 shows the interface for restarting the Worklist Application.

Figure D–1 Stopping Worklist Application

		✓ ⊕ _{Co} workist-wis			Active	🖋 ОК
Image: Constraint of the second se					Active	🖋 ОК
					Active	🖋 ОК
	In	stall Update Delete Start ~	Stop ~			
			When work completes			
			Force Stop Now			
rights reserved. Stop, but continue servicing			Stop, but continue servicing administration requests			

Figure D–2 shows the Oracle WebLogic Server interface for restarting the Worklist Application.

Figure D–2 Restarting Worklist Application

	Toworklist-wls			stop Running	🖋 ок
	🗈 📑 wsil-wls			Active	🖋 ОК
	E B wsm-pm			Active	🖋 ОК
Ir	Install Update Delete Start Stop Servicing all requests				

7. Log in to either Worklist Application or Process Workspace, navigate to the Application Preferences page, and specify your customized items. Figure D–3 highlights the fields to specify.

Login page realm label LABEL_LOGIN_REALM				
Resource bundle	view.CustomResourceBundle			
Use language settings of O Browser Identity Provider				
Branding And Skinning				
* Branding Logo	/faces/logo.png			
* Branding Title	LABEL_WORKLIST_CUSTOM_TITLE			
Choose a Skin	sporty 💌			
Application customization class name	simple minimal blafplus-rich blafplus fusion-base fusion-ti.t.1.3.0 fusionFx fusionFx fusionFx fusionFx workspace			

Figure D–3 Setting Application Preferences

Note: To be sure that your branding logo is retrieved, in the **Branding Logo** field, be sure to precede the file name of your logo with the path name /faces/.